Evaluation of a Liquorice Extract as a Novel Antimicrobial

ΒY

Madiha El Awamie

Thesis submitted to the University of Nottingham for the degree of Doctor of Philosophy

2018

PH.D. RESEARCH STUDENT DIVISION OF FOOD SCIENCE SCHOOL OF BIOSCIENCES



UNITED KINGDOM · CHINA · MALAYSIA

SUPERVISORS: DR. CATHERINE E. D. REES

ABSTRACT

The foodborne disease listeriosis is rare but has a high mortality rate therefore it is a major concern for the food industry. It is caused by the Grampositive bacterium *Listeria monocytogenes* which can survive mild food preservation conditions, (e.g., high salt and low pH), and its psychrotrophic nature means that it can grow at refrigeration temperatures. Hence listeriosis is associated with the consumption of a broad range of ready-to-eat (RTE) foods, including sliced deli meats. Liquorice extract is common a food flavouring that is known to have antimicrobial activity and this project was designed to investigate the antimicrobial activity of a waste product produced during the production of liquorice flavourings for the food industry.

The antimicrobial efficacy of the liquorice extract was used to challenge different types of bacteria to determine whether this food by-product had antimicrobial activity. A range of both Gram-positive and Gram-negative bacteria were tested and it was found that the extract was only effective against Grampositive bacteria suggesting that the extract had a similar mode of action in each case with a concentration of 50 µg ml⁻¹ resulting in total growth inhibition, therefore this was defined as the MIC. The use of bacterial strains containing plasmids expressing bioluminescence revealed that at sub-MIC concentrations (12.5 μ g ml⁻¹) the metabolic state of the cells was reduced, indicated that membrane integrity was affected. The mechanism of action of the liquorice extract was further investigated using a fluorescent LIVE/DEAD[®] stain. Exposure to the MIC for this extract (50 µg ml⁻¹) resulted in total growth inhibition and the presence of non-viable cells was detected using fluorescence microscopy. In addition, fluorescent microscopy revealed the formation of long filaments consisting of cells that had not completed the cell division process, which indicates that cell wall synthesis was also affected. In contrast no antimicrobial effect or

i.

toxic effect was detected when the extract was used to treat Gram-negative bacteria or the yeast *Saccharomyces cerevisiae*.

When applied to the surface of sliced deli meat, inhibition of *Listeria* growth was seen for up to 10 days, hence the extract has the potential as a natural food preservative and could be of use to improve the safety of this type of RTE product. To confirm that the extract was safe to use as a food preservative, the mutagenic potential of the extract was assessed using the Ames test for detecting carcinogens and mutagens by using different strains of *Salmonella* Typhimurium including TA100, TA102 and TA1535, no adverse results were gained. In addition *in vitro* cytotoxicity was measured using Caco-2 cells and the MTT cell viability assay. At the MIC concentration, some cytotoxicity was detected, and this requires further investigation but these results could indicate the potential for use of liquorice extract has a specific effect on Gram-positive bacteria and further work could lead to its use as an antimicrobial agent to control the growth of these bacteria either in food systems or in other settings.

ACKNOWLEDGMENTS

Firstly I would like to thank Allah, Lord of the worlds for this chance who give me to complete my study. I would like to express my appreciation to the University of Nottingham, for providing me with all the facilities and support to finish this research.

I would also like to thank all the people who helped me and gave me advice during my PhD. Special thanks to my PhD supervisor Dr. Cath Rees for her continuous guidance, supervision and encouragement me throughout my PhD, without her support and guidance this study could not be completed. I would like to thank my sponsor, the Libyan Government for giving me a scholarship to enable me to complete my PhD programme. I would also like to thank Richard Argent for his help in my cell culture work which was the first experience for me to work in this area.

In addition, I thank Tania Perehinec, Ben Swift and Lorraine Gillett for their assistance given to me through my research in the laboratory. Finally, I would like to thank all my family and friends for the love and for all support and encouragement.

This thesis is dedicated to my parents for all their love and motivation and encouragement me to reach this stage.

iii

TABLE OF CONTENTS

ABSTRACT	I
ACKNOWLEDGMENTS	III
TABLE OF CONTENTS	IV
LIST OF FIGURES	IX
LIST OF TABLES	XII
LIST OF ABBREVIATIONS	XIII
LIST OF ORIGINAL PUBLICATIONS	XV
CHAPTER 1: REVIEW OF THE LITERATURE	1
1.1 Introduction	2
1.2 Listeria spp. and Listeria monocytogenes	7
1.2.1 Incidence of listeriosis in humans	9
1.2.2 Listeria monocytogenes as a food pathogen	12
1.2.3 Contamination by <i>L. monocytogenes</i> in the food industry	13
1.3 The use of plant extracts as antimicrobials	14
1.3.1 Use of natural antimicrobials for food preservation	16
1.4 Effect of natural compounds on bacteria cells	17
1.4.1 Measuring the susceptibility of bacteria by using different me	thods21
1.5 The liquorice plant	23
1.5.1 The antimicrobial activity of liquorice extracts	27
1.5.2 Toxicity of liquorice plant	33
1.6 Aims and objectives	35

2.1 Chemicals and Media	.38
2.2 Bacterial strains, plasmids, and growth conditions	40
2.3 Preparation of liquorice extract	.40
2.4 Preparation of bacterial cultures	41

2.4.1 <i>Listeria monocytogenes</i> strains41
2.4.2 <i>Salmonella</i> strains/mutants41
2.4.3 Pseudomonas fluourescens chromosomally tagged lux41
2.4.4 <i>Staphylococcus aureus</i> chromosomally tagged <i>lux</i> 45
2.5 Preparation of yeast cultures45
2.6 Monitoring growth of bacterial cultures45
2.7 Frozen stock cultures of bacteria45
2.8 Measurement the optical density of bacteria46
2.9 Determination of the growth rate of Gram-positive bacterial cultures46
2.10 Tecan microplate reader use for measuring bacterial growth47
2.11 Minimum inhibitory concentration determination47
2.12 Miles and Misra technique48
2.13 Gram-staining method
2.14 Recovery of plasmid containing strains from frozen culture
2.14.1 Confirmation of light emission (Luminometry)50
2.14.2 Confirmation of fluorescence phenotype (GFP)50
2.15 Live/Dead stain51
2.16 Evaluating the activity of liquorice extract on bologna-style sausage51
2.17 Mutagenicity of liquorice extract (Ames test)52
2.18 Cell culture conditions
2.18.1 Maintenance of cell culture53
2.18.2 Counting cells with a Haemocytometer54
2.18.3 Determination of cell viability (MTT assay)55
2.19 Cell freezing procedure56
2.20 Evaluation of toxicity of liquorice extracts on the yeast
2.21 STATISTICAL ANALYSIS

СНАР	TER	3:	EVALUATIO	N OF	LIQUORICE	EXTRACT	AS	AN
ANTI	васті	ERIAL						58
Abstra	act							59
3.1	Introd	uctior	۱				•••••	60
3.2	Result	S						67
3	8.2.1 T	est so	lubility of liqu	orice in e	thanol solvent			67
3	8.2.2 E	stabli	shing the grow	th condi	tions required fo	or different Gra	am-	
F	ositive	e and	Gram-negativ	e bacteria	a			68
3	8.2.3 B	acteri	al challenge ex	kperimen	ts			72
3	3.2.4 D	emon	strating the sp	pectrum	of activity of liqu	orice extract	at activ	/e
С	concent	tratior	าร					78
3	8.2.5 E	ffect o	of liquorice ext	ract on b	acterial viability	·		81
3	3.2.6 A	ssess	ment of cell m	orpholog	y using Gram-st	aining		84
3	8.2.7 C	onfirn	nation of the f	luorescer	nt phenotype of	<i>Listeria</i> strain	s carry	ing
S	synthet	ic <i>gfp</i>	: <i>luxABCDE</i> op	eron				88
	3.2.7	7.1 Co	onfirmation of	light emi	ssion (Luminom	etry)		89
	3.2.7	7.2 Co	onfirmation of	fluoresce	nce phenotype	(GFP)		92
3	3.2.8 U	se of	bioluminescen	ce to inv	estigate the effe	ect of liquorice	extrac	t
С	on <i>Liste</i>	e <i>ria</i> st	rains					96
3	3.2.9 C	onfirn	nation of the p	resence	of plasmid after	treatment wit	h and	
v	vithout	: liquo	rice extract					.101
3	8.2.10	Micros	scopic assay o	f membra	ane integrity aft	er exposure to	o liquor	ice
e	extract							109
3	.2.11 l	Jse of	<i>gpf</i> ⁺ marked s	strains to	confirm <i>Listeria</i>	identification		.115
3	.2.12 M	1orph	ology changes	and cell o	elongation			.118
3.3	Discus	sion						.121
3.4	Conclu	ision						.132

C	CHAPTER 4: INVESTIGATION OF MECHANISMS OF ACTION OF LIQUORICE		
E	EXTRACT AS A FOOD PRESERVATIVE	.134	
	Abstract	135	
	4.1 Introduction	.135	
	4.2 Results	.139	
	4.2.1 Evaluating liquorice extract activity on bacteria inoculated on to		
	sausage	.139	
	4.3 Discussion	.145	
	4.4 Conclusion	.148	

CHAPTER 5: EVALUATION OF THE SAFETY AND TOXICITY OF LIQUORICE

EXTRACT	149
Abstract	150
5.1 Introduction	151
5.2 Results	153
5.2.1 Evaluating the mutagenicity and toxicity of liquorice extract	153
5.2.1.1 Confirmation of the presence of plasmids in Salmonella	
strains	155
5.2.1.2 Ames test	156
5.2.2 Evaluating cytotoxicity using a cell viability with MTT assay	161
5.2.3 Evaluation of toxicity of liquorice extracts on yeast	167
5.3 Discussion	169
5.4 Conclusion	173

CHAPTER 6: GENERAL DISCUSSION AND FUTURE WORK......175

6.1 General discussion	
6.2 Inhibition of Gram-positive bacteria by liquoric	ce extract177
6.3 Evaluation of safety of liquorice extract as a fo	ood preservatives182

6.4 Conclusion	
6.5 Future study	

REFERENCES	.189
APPENDICES	.215
APPENDIX 1: Certificate of Analysis of the liquorice extract	.216
APPENDIX 2: Preparation of media	.219
APPENDIX 3: Preparation of chemical solutions	.222

PUBLICATIONS22	28
----------------	----

LIST OF FIGURES

Figure 1.1. Human cases of <i>L. monocytogenes</i> in England and Wales 2008-
201712
Figure 1.2. The mechanism of action of antibacterial agents on the structure of
bacteria at different locations21
Figure 1.3. Images of <i>G. uralensis</i> plants and roots25
Figure 3.1. Bacterial bioluminescence62
Figure 3.2. Growth curves for different types of Gram- negative and Gram- positive
bacteria71
Figure 3.3. Layout of samples in microtitre plate73
Figure 3.4. The effect of different concentrations of liquorice extract on growth of
Gram-positive bacteria75
Figure 3.5. The effect of liquorice extract on the growth of Gram-negative
bacteria77
Figure 3.6. The effect of liquorice extract on the growth of Gram-negative bacteria
(a) & (b) and Gram-positive bacteria (c)-(e)-(f)
Figure 3.7. The effect of liquorice extract on the growth (a) and the viable count
(b) of <i>Listeria innocua</i> 83
Figure 3.8. Effect of liquorice extract on cell morphology of Gram-positive
bacteria
Figure 3.9. Detection of Green fluorescence in individual cells of <i>Listeria</i> 96
Figure 3.10. Growth curve of (a) LM 10357, (b) LM EGD, (c) LM 10403S, (d) LM
7973 and (e) LM 23074 containing plasmid pSB3008 and (f) L.
<i>innocua</i> (pSB3007)98
Figure 3.11. Effect of liquorice extract on (luminescence) light emitted by (a) LM
10357, (b) LM EGD, (c) LM 10403S, (d) LM 7973 and (e) LM 23074 containing
plasmid pSB3008 and (f) L. innocua(pSB3007)99

Figure 3.12. Effect of liquorice extract on (a) LM 10357, (b) LM EGD, (c) LM 10403S, (d) LM 7973 and (e) LM 23074 containing plasmid pSB3008 and (f) L. innocua(pSB3007) reported as RLU per cell mass unit......100 Figure 3.13. Viable count for Listeria monocytogenes 10403S(pSB3008) on different agar plates......103 Figure 3.14. Effect of liquorice extract on growth and metabolism of (a) S. aureus and (b) *P. fluorescence* chromosomally tagged *lux*.....106 Figure 3.15. The effect of liquorice extract on the growth of Mycobacterium Figure 3.16. The effect of liquorice extract on membrane integrity after applying LIVE/DEAD[®] BacLight[™] kit......111 Figure 3.17. Comparison of effects of liquorice on growth and viability staining......114 Figure 3.18. Detection of fluorescence in *L. monocytogenes* 23074(pSB3008) Figure 3.19. Effect confirmed on the elongation of Listeria monocytogenes 23074(pSB3008) after 24 h treatment with 50 µg ml⁻¹.....119 Figure 3.20. Filament formation in *L. monocytogenes* cells grown in BHI broth with and without liquorice extract......120 Figure 4.1. Method used to inoculate L. monocytogenes onto slices of bologna-Figure 4.2. Bologna-style sausage inoculated with high numbers of Listeria Figure 4.3. Growth of L. monocytogenes on bologna-style sausage at 6 °C......144 Figure 5.1. Method for preparing agar plates for Ames test158 Figure 5.2. Effect of treatment of liquorice extract, sodium azide and Mitomycin C treatment on revertant of histidine auxotrophy of the Ames strain S. Typhimurium TA100, TA102 and TA1535......160

Figure 5.4. Caco-2 cell line with different concentration of liquorice extract	and
positive and negative control	164
Figure 5.5. Histogram showing results of MTT assays	.166
Figure 5.6. The growth of yeast in the presence of liquorice extract	.168

LIST OF TABLES

Table 2.1. Chemical substances and media used during this study
Table 2.2. Bacterial strains and yeast used in this study42
Table 2.3. Bacteria with plasmids used in this study43
Table 2.4. Plasmid used in this thesis44
Table 2.5. Human tissues used in this research44
Table 2.6. Chemicals used as a positive control in the Ames test
Table 3.1. Growth of bacteria used in this experiment70
Table 3.2. The growth rate of Gram-positive bacteria
Table 3.3. Observed and calculated mean cell dimensions (μ m) of <i>L. innocua</i> and
B. subtilis
Table 3.4. Effect of growth phase on light emission by Listeria
Table 3.5. Measurements of green fluorescent protein (Gfp) for all Listeria
strains
Table 3.6. GFP for Listeria strains
Table 3.7. Plasmid stability in LM 10403S without selection104
Table 3.8. Plasmid stability in LM 10403S with selection (ERM)104
Table 4.1. The number of bacteria emerging on the slices of sausage inoculated
with lower numbers of Listeria monocytogenes, counted after 24 h incubation, and
converted to log ₁₀ cfu ml ⁻¹ 143
Table 5.1. Genotype of the most commonly used <i>Salmonella</i> test strains*155
Table 5.2. The genotype of the <i>Salmonella</i> Typhimurium TA100 &TA102156
Table 5.3. The cell viability MTT Assay165

LIST OF ABBREVIATIONS

Amp	Ampicillin
ATP	adenosine triphosphate
BHI	Brain Heart Infusion
BPW	Buffered Peptone Water
Caco-2	Human colonic adenocarcinoma cells
٥C	Degrees Celsius
CFU	Colony Forming Unit
DMEM	Dulbecco's Modified Eagle Medium
DMSO	Dimethyl Sulfoxide
FMNH	flavin mononucleotide
Ery	Erythromycin
F.W	Formula weight
FBS	Foetal Bovine Serum
g	gram
GFP	Green Fluorescent Protein
GM	Glucose Minimal Agar
h	hours
KDa	kilodalton
I	Litre
LM	Listeria monocytogenes
log	Logarithm
Lux	Bacteria Luciferase Gene
<i>lux</i> AB	Bacterial Luciferase
μ	Specific Growth Rate
MBC	Minimum bactericidal Concentration
MDR	Multidrug Resistant
μg	microgram
mg	milligram

MIC	Minimum Inhibitory Concentration
min	minute
mm	micromiter
μΙ	microliter
ml	mililiters
MRD	Maximum Recovery Diluent
MTT	3-(4,5-dimethylthiazol-2-yl)-2,5-diphenyltetrazolium bromide
NaCl	Sodium Chloride
OD600	Optical Density at a wavelength of 600 nanometers
PBS	Phosphate Buffered Saline
pSB	Plasmid Sutton Bonington
% (v/v)	percentage volume per volume
% (w/v)	percentage weight per volume
PI	Propidium Iodide
RLU	Relative Light Units
RO water	Reverse Osmosis
rpm	Revolutions Per Minute
TSB	Tryptic Soy Broth
RTE	Ready-to-Eat
SDS	Sodium Dodecyl Sulphate
Tet	Tetracycline
TMTC	Too much to count
x <i>g</i>	Relative Centrifugal Force
Ø	Diameter
MSC	Biological Safety Cabinet
wt	Wild type

LIST OF ORIGINAL PUBLICATIONS

1-EL AWAMIE, M. & REES, C. 2016. Identification of the Antimicrobial Effect of Liquorice Extracts on Gram-Positive Bacteria: Determination of Minimum Inhibitory Concentration and Mechanism of Action Using a *luxABCDE* Reporter Strain. *World Academy of Science, Engineering and Technology, Int. J. Med., Health, Biomed. Bioeng. Pharmaceut. Eng.*, 10, 294-302.

CHAPTER 1

INTRODUCTION AND LITERATURE REVIEW

1.1 Introduction

According to The World Health Organization (2011) approximately 55 million people died worldwide, and one-third of these deaths were due to infectious diseases. In the developing world, infectious diseases remain the main cause of morbidity and mortality (Nabavi et al., 2015). The definition of foodborne illnesses by The World Health Organization (WHO), these are diseases caused by agents that enter the body through the ingestion of food which are either infectious or toxic in nature (WHO, 2015). Serious foodborne disease and poisoning are a widespread and growing public health problem and considered a major contributor to human illnesses, hospitalizations, and deaths each year. According to data from the Centers for Disease Control and Prevention (CDC) it is estimated that about 48 million cases of foodborne diseases occur each year in the United States (about 1 in 6 Americans) resulting in 128,000 hospitalizations and 3,000 deaths annually (Baraketi et al., 2018; CDC 2011). Losses due to medical expenditure and loss of productivity associated with foodborne illness are estimated to be approximately \$152 billion in the United States per year. Although different countries have made significant progress in decreasing the incidence of foodborne disease. Hence for society and the food industry, outbreaks of human disease have a considerable effect on the economic situation (Ivanek et al., 2005) and therefore the problem of food borne disease has been given more attention by both Governments and food industry (Le Loir et al., 2003; Choffnes et al., 2012).

Over the last few decades, several new microbiological problems in food safety have arisen due to the changes in life style, such as an increase in the consumption of ready-to-cook and ready-to-eat products (Kotzekidou, 2013), meaning that foodborne pathogens have become an increasing concern due to contamination of such processed products. This phenomenon is not restricted to one product type, and an increase in infections associated with meat, fresh fruits and vegetables and dairy products has been seen, caused by bacteria such as

Listeria monocytogenes, Campylobacter jejuni and Escherichia coli (Harris et al., 2013). This is not a localised problem and health hazards posed by microbial pathogens in food are of major concern world-wide because circulation of contaminated food between countries increases the chance of outbreaks of foodborne disease caused by imported products as well as outbreaks associated with more locally produced foods. According to International trade statistics produced by the World Trade Organization (WTO, 2015), food represents 42 % of total world exports, and levels have increased 6 % per year since 2010, and so food is a substantial and increasing area of trade and therefore represents and ever increasing risk to human health. Although generally in developed countries the safety of food has dramatically improved overall, foodborne outbreaks from microbial contamination are still prevalent in many countries, and failure to discover the presence of pathogenic microorganisms especially in the food industry may lead to serious effects on human health (McLinden et al., 2014). In addition to the risk of infection, microbes that cause food spoilage can secrete toxins which also impacts on the safety of food. Therefore antimicrobials that can eliminate or delay microbial growth are important to maintain the quality and safety of food products (Davidson et al., 2013).

Foodborne diseases have an impact on economic as well as public health. For instance approximately 1.8 million deaths every year caused by diarrhoeal disease mostly in developing countries, and a large proportion of these cases can be attributed to contamination of food and drinking water (WHO, 2004). However, levels of foodborne disease are generally under-reported as most are self-limiting and not reported to the health care system, therefore it is difficult to measure the impact of this accurately (Van de Venter, 2000). In addition to the impact on the consumers, the food industry also faces large problems when bacterial contamination causes outbreaks of food borne infection since these can lead to loss of consumer confidence, and loss of income. In addition there is also an increasing chance of litigation if an outbreak of disease occurs.

To address these problems, there is an increased emphasis in the food industry to produce safe food through the application of food preservation techniques. Regulatory agencies and consumers are generally more concerned about the microbial safety of foods and the control of foodborne pathogens, but there is also a drive to reduce food spoilage. Some bacteria may bring about desirable changes in food, such as during the production of fermented foods, but there is also a relationship between spoilage of food and microbial activity in terms of the production of metabolic compounds that lead to changes in the taste and colour that are not acceptable for the consumer, resulting in an in economic loss to the industry. This is also a world economic issue since the Food and Agriculture Organization (FAO) of the United Nations reports that one-third of food produced for human consumption is lost or wasted globally, amounting to approximately 1.3 billion tons of food lost per year. This represents a waste of the resources used for the production of the food, including land, water, energy and inputs, and increasing greenhouse emissions without producing any benefit (FAO, 2011).

For the control of microbial spoilage many of methods have been used to reduce contamination levels in food products, such as fermentation, pasteurization or the addition of antimicrobials. The use of heat, chemical compounds, and UV radiation to minimize microbial load may expose food to changes in a taste which are important to consumer acceptability (Lau *et al.*, 2014). However the problem of food spoilage is still a challenge leading to high levels of food waste and a solution has not yet been found (Raybaudi-Massilia *et al.*, 2009). To make this task more difficult, consumers are demanding products which are less processed and more natural or fresh (Roman *et al.*, 2017; da Silva Dannenberg *et al.*, 2016) and so the food industry is looking for alternative ways to produce safe food. Hence the use lower amounts of additives in foods has been investigated by the food

industries but it is important that these changes do not affect the quality of food, since preservatives such as salt, sugar and organic acids also contribute to the taste perception of the food, and this is considered to be the most important concern of consumer.

Natural preservatives have been used as alternatives to traditional chemical preservatives, however a limited number have been commercially developed and many remain to be investigated as sources of safer and effective antimicrobials. Therefore, alternative sources of safe, effective and acceptable natural preservatives need to be explored (Lau et al., 2014; Negi, 2012). Many plant-derived antimicrobial compounds have been found to have activity against foodborne bacteria and thus, they could be used as natural preservatives in foods (Cho *et al.*, 2008). The ability of some Gram-positive bacteria to produce spores poses a particular problem, as their ability to resist tolerate extreme environmental conditions, such as heat and chemical treatments, make them more difficult to eliminate than vegetative cells of the same organisms. Therefore, in addition to targeting the vegetative cells present in a food, agents to prevent the germination the spores into vegetative cells under appropriate condition is an area of concern since outgrowth of these spores can lead to food spoilage and foodborne diseases (Fernández-No et al., 2011). Therefore, many natural products such as plant extracts, essential oils and organic acids to ensure food safety and control of bacterial spores have been investigated as alternative sources of antimicrobials (Burt, 2004; Chouhan et al., 2017).

In different commercial sectors, including medicinal, the use the plant products growing dramatically due to the variety in phytochemical compounds that have been identified in different species of plants (Al-Terehi *et al.*, 2015a). It is estimated the plant species present on Earth number 250,000 to 500,000 and only a small proportion have been explored as sources of useful natural products until now (Negi, 2012; Srujana *et al.*, 2012). Plant products have been traditionally

used in folk medicine as a medicinal plant in different countries of the world to treat many diseases including bacterial infections. Geetha and Roy (2012) indicates that the same plants were used by people in different regions of the world to treatment of many ailments for many centuries. As a result of the development of resistance to antibiotics by some types of bacteria caused by the misuse of antibiotics (Li and Webster, 2018), this has also led researchers to try and discover new antimicrobials from novel sources, such as plants. Consequently, researchers have established that herbal remedies such as clove, oregano, thyme, cinnamon, and cumin are effective against a wide range of microorganisms and can be used to treat infectious diseases or protect food because they were shown to possess antimicrobial activities against pathogenic and spoilage fungi and bacteria (Chouhan et al., 2017). However it is important to know the chemical constituents of the plants and to screen them in vitro for any unwanted toxicity before they are used as therapeutics (Geetha and Roy, 2012) since, in addition to many antimicrobial substances, plants are also known to produce many of the most toxic compounds known to man (Patel *et al.*, 2013).

What makes these plant extracts particularly good candidates as sources of food preservatives is that they are commonly used as a flavourings in food products and also as natural food preservatives in many countries and, therefore, the use of plant extracts with known antimicrobial properties can be of important significance in food preservation. The impact of plant extracts on microorganisms is known to be due to the presence of a wide range of chemical ingredients including polyphenols, flavonoids, tannins, alkaloids, terpenoids, lectins, polypeptides which produce inhibitory effects on the growth a range of bacteria, moulds and yeast (Edeoga *et al.*, 2005; Cowan, 1999). Hence many studies have been carried out on plant products in different countries worldwide to examine the performance and efficacy these products against many types of microorganisms (Negi, 2012).

1.2 Listeria spp. and Listeria monocytogenes

Until recently the genus of Listeria included 17 species (Orsi and Wiedmann, 2016). However, recently a new species (Listeria costaricensis; Núñez-Montero et al., 2018) was identified and therefore the genus now contains 18 species. Within the different species of Listeria, L. monocytogenes and L. ivanovii are the only ones considered to be pathogenic. L. monocytogenes is an important human foodborne pathogen and is considered to be the third major cause of foodborne deaths in the USA due to contamination of food with this microbe (Scallan et al., 2011). Although all strains of *L. monocytogenes* have the ability to cause serious diseases in humans and a broad variety of animals (Drevets and Bronze, 2008). Doumith et al. (2004) and Zhang and Knabel (2005) reported that L. monocytogenes serotypes 1/2a, 1/2b, and 4b are responsible of the vast majority of human listeriosis. In contrast, *L. ivanovii* can rarely cause infections in humans and is primarily regarded as an animal pathogen, mainly causing disease in sheep and cattle, but their pathogenic potential means that these two species are the best studied members of the group (Graves et al., 2010; Leclercq et al., 2010). According to European laws regarding Microbiological Criteria for food products (Commission Regulation EC 2073 / 2005), L. monocytogenes is the pathogenic bacterium that particular attention should be paid to because many challenges exist for achieving control of *L. monocytogenes* in the food chain such as able to survive and grow in harsh conditions including low temperatures, dry environments and over a wide range of pH values (Bertrand et al., 2016).

All members of the genus are small rod-shaped, Gram-positive bacteria that are prevalent everywhere in the environment including soil, decaying plants, sewage, dust, and water. Due to this extensive distribution, *L. monocytogenes* and other *Listeria* spp. regularly contaminate food processing plants, with high risk areas being recognised including receiving, handling, and processing areas (Ryser and Marth, 2007; Weller *et al.*, 2015). Although low temperature is normally used

to prevent the growth of bacteria in such environments, Listeria spp. are psychrotrophs and therefore have the unusual ability to adapt to low-temperature conditions. It can grow over a wide temperature range between 3 and 45 °C which also means it can grow to dangerous levels in refrigerated foods. The physical state of the lipids in the membrane of the bacterial cells in the form of crystalline liquid is considered important for the preservation of the membranes structure and their function to all enzymes activity and transport of solutes across the membrane (Beales, 2004). To allow growth to continue when the temperature is reduced below 7 °C changes need to occur in the to maintain the liquidity of the membrane. In L. monocytogenes one of the main changes seen is an increase in the proportion of C15:0 (Pentadecanoic is a saturated fatty acid) lipid at the expense of C17:0 (Margaric is a saturated fatty acid), as well an increase in the degree of unsaturated fatty acids, which helps enhance the liquidity of the membrane and allow growth to continue at low temperatures (Hsu et al., 2011). Another adaptation seen is that when grown at temperatures between 20-25 °C, Listeria spp. are actively motile by peritrichous flagella and the production of flagella help the cells attach to food surfaces (Vatanyoopaisarn *et al.*, 2000). When grown at temperatures between 30-37 °C, *Listeria* strains are non-motile because the organism does not have the ability to synthesize flagella due to lack the expression of the flagellin proteins at this temperature (Way *et al.*, 2004). This is another example of the adaptive processes that *Listeria* spp. undergo when they transition between being an environmental organism and a bacterium adapted to growth in animal hosts.

As a facultative anaerobe, *Listeria* has the ability to multiply rapidly in aerobic or microaerophilic conditions and it can also tolerate and grow over a wide range of pH levels, with growth reported to occur from pH 4.1 to 9.6 (Lungu *et al.*, 2009). It is also desiccation-resistant and can tolerate low moisture levels and can form biofilms making it resistant to disinfection. All these features

demonstrate that it has a great adaptability to different environments, which explains why the bacterium can colonize food production equipment and environments. Consequently, these bacteria have been shown to persist for months to years in food-processing plants, acting as a possible source of cross-contamination of product. Indeed, contamination of food products with *Listeria* has been identified at many different stages of food production including slicing, packaging and chill-storage. Thus, control and elimination of this type of bacteria from food products is the best strategy to ensure food safety (Vázquez-Boland *et al.*, 2001).

1.2.1 Incidence of listeriosis in humans

Amongst foodborne bacterial pathogens, infections caused by L. monocytogenes have the highest hospitalization and mortality rates (Scallan et al., 2011). Infection by this bacterium results in the disease listeriosis that can have a number of different serious consequences including meningitis, meningoencephalitis and septicaemia and is distinguished by a high mortality rate of up to 30 % and represents a significant public health threat (Weller et al., 2015; Painter and Slutsker, 2007). Infection is generally seen in people with weakened immune systems, including pregnant women, infants, and the elderly. With rising consumption of RTE foods or frozen foods which require minimal heat treatment before consumption, the risk of listeriosis has increased due to food processing plants often using conditions that can be tolerated by Listeria, such as refrigeration, low pH, or high salt concentration to preserve foods. Moreover, L. monocytogenes is known to persist by adhering to food contact surfaces, leading to the post-process contamination of final food products (Ferreira et al., 2014). The severity of the diseases caused by this bacterium means that there is a high economic cost associated with infection and therefore it is a major concern for the food industry (Raghu, 2013; Olszewska et al., 2015). In the United States L.

monocytogenes is considered to be of one of the most important causes of diseases transmitted through foods, where it listed among the top five most important pathogens (Gialamas *et al.*, 2010; Theinsathid *et al.*, 2012), and the health costs associated with listeriosis was estimated at \$2.6 billion per year (Hoffmann *et al.*, 2012). Similarly in Canada the costs associated with the cases (including medical and nonmedical costs and productivity losses) were estimated to be nearly \$242 million Canadian dollars (Thomas *et al.*, 2015).

The incidence of listeriosis has increased in both Europe and North America over the last few decades with at least 13 illness outbreaks associated with RTE meat products and dairy products between 1987 and 2008 where L. monocytogenes was identified as the leading cause of infection (Lara-Lledó et al., 2012). For instance, during an outbreak in Canada in 2008 related to the consumption of contaminated delicatessen meat, 57 cases of listeriosis were reported, resulting in 24 deaths, and was one of the biggest outbreaks of listeriosis in Canadian food history (Thomas *et al.*, 2015). Despite the lessons learned from this case, the difficulty faced by food companies ensuring this bacterium does not contaminate products continues to be an issue. For instance L. monocytogenes (LM P93) was found to be responsible for 38 infections, resulting in the death of 5 people, in a food poisoning listeriosis outbreak in cheese made from pasteurized milk in Canada in 2012 (Gaulin et al., 2012). Other recent notable Listeria outbreaks have been associated with sliced melons in Texas in 2010 (146 cases and 31 deaths; Gaul et al., 2012) and in Australia another outbreak associated with melons resulted in 20 cases and 7 deaths (WHO, 2018). Recently in January 2017, South Africa has been affected by the largest outbreak of *L. monocytogenes* infections recorded, which caused a death of at least 200 people and further confirmed infections of 1024 people since the outbreak was discovered (WHO, 2018). Hence Listeria is regarded by the food industry as a major cause for concern

(Montañez-Izquierdo *et al.*, 2012), and therefore it was chosen in this study as a key organism for testing the efficacy of a novel antimicrobial.

According to the data published by the Public Health England in 2017, 135 cases of listeriosis were reported in England and Wales, representing a 17.7 % decline compared to the previous 6 year period (2010- 2016; Figure 1.1; PHE, 2018). An outbreak of listeriosis has been identified across Europe between August 2016 and June 2018 with cases identified in a number of different countries, including Austria, Denmark, Finland, Sweden, and the United Kingdom. This outbreak was linked with the consumption of frozen corn and other frozen mixed vegetables that had not been cooked before being consumed. As of 15 June 2018 it was reported that there were a total of 47 cases of listeriosis associated with this outbreak which caused nine deaths, 23 cases were revealed in Finland, 11 in United Kingdom, 7 in Sweden, 14 in Denmark and 2 in Austria (Authority *et al.*, 2018).

Figure 1.1. Human cases of *L. monocytogenes* in England and Wales 2008-





Data from the Public Health England showing human listeriosis caused by *L. monocytogenes* in England and Wales from 2008 to 2017. Data shows that a decrease occurred in 2017 compared to that recorded in the previous 6 years between 2010-2016 (adapted from PHE, 2018).

1.2.2 Listeria monocytogenes as a food pathogen

L. monocytogenes is a foodborne pathogen that is almost exclusively transmitted to humans through consumption of contaminated food, although zoonotic and person-to-person (mother-to-child) infections also occur (Rees *et al.* 2017). As it is a non-spore forming bacterium, the utilization of high temperature treatment of products will eliminate the pathogen, therefore the major concern is contamination of the final product with *Listeria* before packaging, and the psychrotrophic nature of the bacterium means that levels of contamination can increase to high levels over weeks or months of refrigerated storage (Nørrung *et al.*, 2009; Sofos and Geornaras, 2010). Therefore many different techniques are

employed to inhibit the spread and survival of *L. monocytogenes* in food products. Its unusual biology – being able to grow equally well in soil or in the GI tract of animals or within food production facilities - means that it is associated with a broad range of ready-to-eat (RTE) foods. L. monocytogenes have been discovered in a broad variety of food types including raw and processed foods. For instance in open markets in China it has been reported that *L. monocytogenes* was routinely isolated from raw pork (Luo et al., 2017). Fresh produce including pepper, radishes, cabbage, seafood and fish products, milk and dairy products, meats and meat products such as beef and sausages, have also all been associated with Listeria contamination (Lianou and Sofos, 2007). As mentioned above, L. monocytogenes has the ability to survive in the food-processing and producepacking environment and equipment. Hence, cross-contamination of product by Listeria is a significant concern for the food industry (Montañez-Izquierdo et al., 2012), especially because of their ability to survive and grow in a wide range of environmental conditions (Zhu et al., 2017), as reviewed above, which amplifies the problem.

1.2.3 Contamination by L. monocytogenes in the food industry

As illustrated by the two outbreaks described above, ready-to-eat meat products pose the most serious threat to the public and deli meats are one of the RTE foods which represent a high rate of risk of infection and death caused by *L. monocytogenes* (Gallagher *et al.*, 2003). This is due to the high risk of contamination after the heat treatment due to the high number of processing steps after cooking (slicing and packing) and also the fact that these products are not cooked before being consumed which would inactivate any *Listeria* contamination (Barmpalia *et al.*, 2005). Although thermal treatments are used in the production of meat products such as bologna-style sausage implicated in the South African outbreak mentioned earlier, contamination of these products by *L. monocytogenes*

can occur from bacteria that have become resident in the food processing environment during slicing and packaging (Swaminathan and Gerner-Smidt, 2007; Espitia *et al.*, 2013) as *L. monocytogenes* has the ability to attach to several food-contact surfaces including stainless steel, polystyrene and glass (Di Bonaventura *et al.*, 2008). As *L. monocytogenes* can survive well in the food environment, it can also colonise slicing machines which can then lead to large scale contamination of product by spreading this organism on the food contact surface (Beresford *et al.*, 2001; Frye *et al.*, 2002; Gormley *et al.*, 2010).

Since contamination of RTE meat products is very difficult to control, to prevent the growth of organisms in order to ensure the safety and extend the shelf-life of these products, several techniques have been used such as packaging in a controlled atmosphere, packaging using activated films, non-thermal pasteurisation treatments and irradiation are employed (Davidson *et al.*, 2013). However, although all of these treatments could help to protect food from the growth of pathogenic *Listeria*, on the other hand they will also have an impact on the product in terms of quality by causing changes in the organoleptic characteristics of foods, such as loss of colour or flavour and which reduces consumer acceptability especially as consumers are now demanding products that are less processed. Therefore chemical food preservatives are required, but even these when used at levels acceptable to the consumer cannot entirely eliminate food pathogens such as *L. monocytogenes* or entirely delay microbial spoilage (Gutierrez *et al.*, 2009).

1.3 The use of plant extracts as antimicrobials

Herbal remedies have been used for many centuries for the treatment of a range of illnesses, particularly in developing countries but has also been favoured in some developed countries such as Germany, France, Italy and the United States in order to help maintain health (Murray and Pizzorno Jr, 2000). The increase in medicines derived from medicinal plants is estimated at about 25 % over the past decade. Today herbal drugs are used by 75-80 % of the world's population, especially in developing countries, where there is significant use of medicinal plants in the healing of ailments, and there is also still a dependency on traditional medicines and folk treatments for their healthcare needs (Kamboj, 2000). Medicinal plants grow in the wild and are wide spread all over the world, and since they are easily obtainable and inexpensive they are considered an important source of income for native and rural populations (Robinson *et al.*, 2011).

While scientific research plays a major role in the validation of conventional treatments, there has been an emphasis on identifying the biological activity of plant extracts and the active ingredients used in traditional medicines (Tanaka *et al.*, 2009). Many plants with biological and anti-microbiological properties have been studied since there has been a relevant increase in the incidence of antibiotic overuse and misuse and the emergence of new resistant strains of bacteria (Saxena, 2005). Thus there has been a resurgence of interest in the search for compounds which are derived from natural sources and used against bacteria (Sedighinia and Afshar, 2012; Hatano *et al.*, 2000; Irani *et al.*, 2010). Moreover, about 60 % of the drugs available on the market, such as anti-tumour and anti-microbial drugs are derived mainly from natural products for medicinal plants (Bhatnagar and Kim, 2010; Lv *et al.*, 2011).

Primary metabolites produced by plants are those known to have an essential role in plant metabolism, whereas secondary metabolites are those where no role has yet been found in growth, photosynthesis, reproduction, or other "primary" functions. Although they may have essential functions for their producers that have not yet been identified, they could also just be waste products of primary metabolism. Plants produce a high diversity of bioactive secondary metabolites, a great number of which serves to protect themselves against microbial, insect and animal attacks (Wink, 2018; Tiwari *et al.*, 2009), and include

antioxidants and antimicrobial agents which enable the plants to survive different sorts of biological challenges. The bioactive ingredients which are considered to have the most significant role in these medicinal plants include alkaloids, tannins, flavonoids and phenolic compounds (Nirmala and Selvaraj, 2011). Thus, food poisoning microorganisms and spoilage microorganisms whose metabolic end products or enzymes cause off-odours and off-flavours, as well as texture problems and discoloration, are considered the main targets for antimicrobials derived from plants that can be employed as food preservatives, such as essential oils extracted from different plants including garlic, onion, thyme, marjoram, rosemary and oregano (Swamy *et al.*, 2016).

1.3.1 Use of natural antimicrobials for food preservation

The global issues of foodborne disease and food spoilage mean that it is necessary to find a variety of antimicrobial agents to control the growth of food pathogens (Mastromatteo *et al.*, 2010; Hereu *et al.*, 2012; Holck *et al.*, 2011). Although many chemicals are used as preservatives to control foodborne pathogens as well as to extend the shelf-life by inhibiting spoilage organisms, there is increasing concern about adverse effects of such compounds on human health and so there has been a growing trend towards replacing synthetic preservatives by natural ones. As a result of the potential health risks associated with synthetic additives, and increase awareness of consumers, this has led to increased interest in use natural alternatives (Ekor, 2014).

Plants that have a relatively high level of antimicrobial action may be sources of compounds that inhibit the growth of foodborne pathogens (Ibrahim *et al.*, 2006). For instance it is known that essential oils and plant extracts have been used for thousands of years for many purposes such as food preservatives as well as being used as medical and pharmaceutical products to prevent diseases. Zhang *et al.* (2009) studied the antibacterial properties of essential oils from different

plants, including liquorice, and tested them against four common meat spoilage and pathogenic bacteria (L. monocytogenes, Escherichia coli, Pseudomonas fluorescens and Lactobacillus sake) using modified atmosphere packaged, fresh pork and vacuum packaged ham slices. Their results showed that individual extracts of various plants, including liquorice, contained strong antibacterial activity, but the mixture of rosemary and liquorice extracts (1:1, v/v) was found to be the best inhibitor against all four types of microbes when compared with other spices. Since these herb extracts are widely used in the food industry and are generally regarded as safe, consequently, they may be considered as natural preservatives tolerable by the food industry. Thus, the use of natural antimicrobials such as essential oils and plant extracts could be an excellent alternative to guarantee food safety (Appendini and Hotchkiss, 2002; Lucera et al., 2012; Del Nobile et al., 2012). Accordingly there has been an increase in research focussing on natural antimicrobials to apply to food products to extend shelf-life by inhibiting microbial growth (Lanciotti et al., 2004; Fattouch et al., 2007), either focussing on inhibiting/controlling the growth of pathogenic organisms (food safety) or on eliminating the growth of spoilage microorganisms (food preservation) (Tajkarimi *et al.*, 2010).

1.4 Effect of natural compounds on bacterial cells

Many natural compounds are not yet utilized commercially and are still under investigation. The antimicrobial compounds in plant materials can be located in different parts of plant, such as essential oil commonly found in leaves (rosemary, sage, basil, oregano, thyme, and marjoram), seeds (caraway, fennel and parsley), flowers or buds (clove), bulbs (garlic and onion), rhizomes (liquorice) or other parts of plants (Gutierrez *et al.*, 2008; Burt and Reinders, 2003; Douglas *et al.*, 2004).

Some antimicrobial compounds, such as phenols, flavones, and flavonoids, tannins, alkaloids, aldehydes, isoflavonoids and terpenoids which have been shown to possess potent antibacterial activity are known to delay or inhibit the growth of bacteria, yeast, and molds, through affecting the structure of the bacterial cell, which is considered a target for action by these components of natural products. Although, the action of natural antimicrobials against both Grampositive and Gram-negative bacteria may signal the existence of broad-spectrum antibiotic components or metabolic toxins, generally plant extracts are found to have a much more significant inhibition effect against Gram-positive than Gramnegative bacteria (Srinivasan et al., 2001; Tegos et al., 2002). For instance (Fiamegos et al., 2011) discovered that plant derived compounds had a better effect against Gram-positive bacteria such as Staphylococcus aureus, Enterococcus faecalis and Bacillus cereus than it did on Gram-negative bacteria, and it has been suggested that Gram-negative bacterial may have natural multidrug resistance to many antimicrobial compounds due to the presence of efflux pumps which confer antibiotic resistance on these types of cells (Piddock, 2006).

Consequently, the mechanisms of action of bacterial agents depends on the type of microorganisms which may be linked to their cell wall structure as well as the presence of an outer membrane. It is known that the cell membrane acts as a selective barrier which allows some small molecules to pass through and excludes others, although the bacterial cell can also control the access of these molecules to the cell. Therefore increased membrane permeability is a critical factor in determining the antimicrobial action of many compounds, although antimicrobial agents may not have to cross the membrane to reach their target site, but rather cause membrane damage, resulting in loss of membrane integrity and eventually leading to cell death (Wu *et al.*, 2016).

Thus, the difference in cell structure and membrane arrangement between Gram-positive and Gram-negative bacteria may account for differences in sensitivity to different antimicrobials (Bechinger and Gorr, 2017; Wu et al., 2016). In general, due to the lipopolysaccharide layer of the outer membrane, Gramnegative bacteria are less sensitive to the antimicrobial agents due to the ability of the outer membrane to limit the diffusion of the hydrophobic compounds. Hence the action of bactericidal agents that interfere with biosynthetic or metabolic functions of cells will be affected by the physical and chemical properties of the chemicals and whether or not they can cross the membrane to reach their target sites to either kill or inhibit the cells. However, it does not mean that Gram-positive bacteria are always more susceptible (Burt, 2004). For instance it has been found that essential oils were more effective against Gram-positive than Gram-negative bacteria (Ceylan and Fung, 2004). The same is true also for many spices and herbs (Shan et al., 2007). Another study using the disc-diffusion method confirmed that compounds with phenolic groups, such as oils of clove, oregano, rosemary, thyme, sage, and peppermint, are effective antimicrobials, and were more inhibitory against Gram-positive than Gram-negative bacteria (Kozłowska et al., 2015).

Plant antimicrobial compounds have been found to cause disruption of a number of different target sites including the cytoplasmic membrane, by disruption of the phospholipid bilayer. Phytochemical constituents of several extracts have been shown to have antimicrobial activity, which is probably due to the combined effects of adsorption of antibacterial agents with bacterial membranes which cause membrane disruption and dissipation of the proton motive force (PMF) leading to a non-energised membrane which subsequently leads to leakage of cellular contents (Farha *et al.*, 2013). Another target are the enzymes of the electron transport system and active transport systems, again resulting in cells with reduced energy levels. Cota *et al.* (2013) reported that the various biologically active compounds found in plants diffuse across the cytoplasmic membrane and

compete for the active sites of particular enzymes inside the cell, thereby prevent cell growth. Inhibition of the ATPase syntheses which lead directly to induce the alteration in the energy state of the bacterial cells and leads to cell death, is caused by plant-derived natural product such as phenolics, flavonoids, quinines, tannins, coumarins, alkaloids, terpenoids and polypeptides (Gill and Holley, 2006a; Gill and Holley, 2006b; Upadhyay *et al.*, 2014).

Alternatively they may cause denaturation of cellular components resulting in coagulation of cytoplasmic contents (Figure 1.2; Burt, 2004) or they may compromise the genetic material of bacteria leading to mutations and cell death (Burt *et al.*, 2007; Lanciotti *et al.*, 2004; Arqués *et al.*, 2008; Proestos *et al.*, 2008). Cell wall synthesis, metabolic pathways, protein synthesis, nucleic acid synthesis, can all be inhibited by the mechanisms of action of antimicrobial agents such as essential oils (Kherallah, 2002). However some bacteria have mechanisms which help to pump out harmful substances to protect themselves against these antimicrobial compounds. Bacteria can also gain plasmids which encode genes that confer resistance to certain antibacterial compounds or encode additional efflux pumps. Interestingly many active constituents derived from plants act as efflux pump inhibitors (EPIs) and specifically block this resistance mechanism particularly for Gram-positive bacteria (Chérigo *et al.*, 2008; Kumar *et al.*, 2008; Prasch and Bucar, 2015).
Figure 1.2. The mechanism of action of antibacterial agents on the structure of bacteria at different locations.



Mechanisms and sites of action of antibacterial agents to destroy the microorganisms and disrupt or damage to this structure which lead to leakage of important solutes (adapted from Nazzaro *et al.*, 2013). Schematic represents a Gram-positive cell type with a thick layer of peptidoglycan which contains teichoic acids; in Gram-negative bacteria an outer membrane and thin layer of peptidoglycan replaces the thicker layer of peptidoglycan forming the cell wall.

1.4.1 Measuring the susceptibility of bacteria by using different methods

Different laboratory techniques can be applied to examine or screen the antimicrobial activity of an extract or a pure compound. When carrying out the evaluation of antimicrobial agents, researchers routinely use the agar dilution, broth microdilution or disk diffusion assays to measure the *in vitro* susceptibility of bacteria. These are the standard methods recommended by the Clinical and Laboratories Standards Institute (Baraketi *et al.*, 2018).

The broth microdilution and agar dilution methods are used to gain quantitative results such as minimum inhibitory concentration and provide qualitative estimations of the sensitivity of different cells types which result in classifications of susceptible, intermediate resistance or resistant. However, the disk diffusion method is not convenient for testing non-polar samples or samples that do not easily diffuse into agar but it is often used as a first screen despite the fact it is unable to accurately determine minimal inhibitory concentrations (MIC; defined as the lowest concentrations of the test compounds that prevented visible bacterial growth) and is not quantitative, so, it is impossible to quantify the amount of the antimicrobial agent diffused into the agar medium. The broth microdilution method can be routinely used due to it being easier to perform and is less time-consuming, whereas the agar dilution method is more time consuming, and is unsuitable to be carried out in most of clinical microbiology laboratories (Kobayashi et al., 2004; Cos et al., 2006). There are different ways to define the end-point in broth dilution studies, but the most utilized methods are optical density (OD) (turbidity) measurements or the viable count. The first method has the advantage of being automatable using instruments such as plate readers (Stevenson et al., 2016) whereas viable counts are more accurate but are more labour intensive (Biesta-Peters et al., 2010).

Additional in depth studies about the effect of an antimicrobial on the growth of pathogenic microorganisms can be achieved through the use of a number of different techniques including the time-kill test and flow cytofluorometric methods which give real information on the nature of the inhibitory effect whether bactericidal or bacteriostatic as well as determining whether its action is dependent on the time or concentration and also shows if there is any damage inflicted on the microorganism during treatment with the antimicrobial (Balouiri *et al.*, 2016).

1.5 The liquorice plant

Liquorice is derived from the dried roots and rhizomes of *Glycyrrhiza* species (Leguminosae family). The Leguminosae family, also known as Fabaceae, is one of the most widely utilized families for medicinal purposes and an important natural sweetening agent; the Fabaceae family represents a patrimony of high biodiversity known for their abundance of secondary metabolites, which are of especial importance for human health (Wink, 2013). This family includes about 727 genera and 19327 species and contains many genus, including *Glycyrrhiza* distributed over on the tropical and moderate areas which have been shown to be effective against different strains of bacteria (Lewis, 2005). The plant is found to be distributed in the subtropical and hot districts in the world such as Asia, the Mediterranean and regions of southern Europe (Saxena, 2005).

In medical treatments liquorice plant is known as a one of the major plants that has a long history of use in both the medical and culinary spheres due to the fact that it contains flavonoid and phenolic compounds as well as coumarins, volatile oil, saponine and minerals. Extracts are used as spices and flavours for various food products as well as effective drugs for many applications in folk medicine, especially in Africa and Asia (Yamamura *et al.*, 1992; Al-Terehi *et al.*, 2015b). Several diseases ranging from simple infections to serious diseases such as skin diseases, asthma, malaria and other infections use several plant species including liquorice plant as treatments (Roshan *et al.*, 2012; Aggarwal *et al.*, 2015). The first documented medicinal use of liquorice can be traced back 4000 years to ancient Assyrian, Egyptian, Chinese and Indian cultures, so it is considered one of the most significant traditional medicinal plants for its therapeutic qualities (Fukai *et al.*, 2002a). The medicinal use of liquorice plant was also documented in the written form by the ancient Greeks, and has been well used in Europe since these times (Fiore *et al.*, 2005).

Not only has the liquorice plant been used in folk medicine as a medicinal plant in different countries of the world to treat many diseases including bacterial infections, it is also commonly used as a flavouring in food products and also as a natural food preservative in many countries (Škrinjar and Nemet, 2009). Different species of liquorice include G. glabra, G. uralensis (Figure 1.3), G. inflata, G. eurycarpa, G. aspera, and G. korshinskyi (Zhang and Ye, 2009; Yang et al., 2014) are all rich in chemical constituents such as flavonoids which are used as drugs for treatment of different kinds of disease such as gastritis, gastric ulcer, stomachic, bronchial diseases, gastrointestinal diseases, cough, liver and bile diseases, duodenal ulcers and gastrointestinal in the Western countries. In Japan it is used as an over-the-counter medicine and may be purchased without a doctor's prescription (Fukai et al., 2002b). The extract of different species of liquorice plants consists of many known medically active chemical ingredients, such the flavonoids glabridin, glabrene, licochalcone A, licoricidin and licoisoflavone which have been shown to have inhibitory activity against the growth of Helicobacter pylori, which is known to be a major cause of gastric ulcers, in vitro (Fukai et al., 2002a). A study conducted by Asha et al. (2013) reported that Glabridin is the major flavonoid present in GutGard® which exhibited good anti-Helicobacter pylori activity in both agar dilution and microbroth dilution methods.

Figure 1.3. Images of *G. uralensis* plants and roots



G. uralensis plant

Glycyrrhiza uralensis is described as a short shrub with oval leaflets, flower clusters with a white and purplish colour shown in panel A and the roots of liquorice shown in panel B (Adapted from Zhang and Ye, 2009; Yang *et al.*, 2014).

Recent chemical analyses have found a wide variety of bioactive compounds present in liquorice. So far, more than 400 compounds have been isolated from *Glycyrrhiza* species. The active constituents such as triterpene saponins and flavonoids in liquorice contribute to the bioactivity of liquorice and are considered to be responsible for these beneficial effects seen in the treatment of diseases (Zhang and Ye, 2009). The content of these bioactive in *Glycyrrhiza* vary considerably with the different plant species, due to plant age, seasonal differences, geographical changes and time of harvest and process collect plant, all of these diversity affect their biological activity and thus affect the therapeutic effects of liquorice (Fiore *et al.*, 2008). Statti *et al.* (2004) collected *G. glabra* L. from different regions of Calabria, Italy, and confirmed an inhibitory effect on bacteria and fungi, and found that the biological activity varied due to differences in the chemical composition of this plant obtained from different sites. In addition the chemical constituents are often influenced by genetic, environmental and processing factors (Isbrucker and Burdock, 2006).

G. uralensis root

As described by He et al. (2006) bioactive ingredients such as essential oils, flavonoids, triterpenoids, alkaloids, phenols, polyamines, and polysaccharides produced by liquorice have all been evaluated by many research groups for potential use as new drugs. The presence of flavonoids including liquiritin, isoliquiritin (a chalcone) in the liquorice plant gives them a yellow colour (Nazari et al., 2017). Most of the pharmacological studies of liquorice saponins indicate that glycyrrhizin and its constituents appear to possess activity against a variety of cancers. In addition to the antioxidant activity due to the presence of isoflavones glabridin and hispaglabridins A and B (Vibha et al., 2009). The activity of glycyrrhizin is due to the presence of aglycone and 18β -glycyrrhetinic acid, and these two compounds exhibit extensive biological activities and a large number of biological activities have been reported for liquorice plants such as antiinflammatory activity (Kwon et al., 2013), antiviral, antimicrobial, antioxidative and anticancer activities (Shibata, 2000), and immunomodulatory and cardioprotective effects which are believed to explain the extended spectrum of activities associated with the use of these plants. However, many of these are preliminary studies and further research is needed to enhance the application of these potential impacts (Asl and Hosseinzadeh, 2008).

In addition to the biological activity of Glycyrrhizin, it is also responsible for its sweet taste (Saxena, 2005) and is fifty times sweeter than sugar which is a saponin glycoside (Nazari *et al.*, 2017). Saponins are also used as flavouring agents, which are generally considered to be bitter, but, can be sweet (Heng, 2005). Thus, the presence of glycyrrhizin in liquorice extracts had led to its uses in pharmaceutical and sweetmeats industries.

1.5.1 The antimicrobial activity of liquorice extracts

The antimicrobial activities of liquorice extract have been demonstrated against both foodborne Gram-positive and Gram-negative bacteria by many researchers and the extract exhibits potent anti-bacterial activity. For instance Shirazi et al. (2007) investigated the antibacterial activity of G. glabra extract in vitro against Gram-negative bacteria, including Salmonella Typhimurium, Salmonella paratyphi B, Shigella sonnei, Shigella flexneri and Enterotoxigenic Escherichia coli (ETEC) using both agar-well and disc diffusion methods with different concentrations of liquorice extract (5, 7.5, 10 and 15 %). All of the bacterial strains examined showed sensitivity to high concentrations of liquorice, however, Salmonella paratyphi B showed no sensitivity to concentrations less than 7.5 % (Shirazi et al., 2007) which is far higher than the allowed levels for food. Compounds derived from the root of *Glycyrrhiza uralensis* have also been evaluated for activity against oral pathogenic bacteria that cause dental caries namely, *Streptococcus mutans* (He et al., 2006). Biological activity of the ethanolic Glycyrrhiza glabra roots extract was identified by Meghashri (2009) and shown to be antimicrobial against six bacteria and two fungal strains using both the well diffusion method and microdilution method. A significant antimicrobial activity was exhibited by the extract against all tested microorganisms, with MICs ranging from 0.8 to 200 mg ml⁻¹ for the fungal strains and MICs ranging from 0.2 to 1.2 mg ml⁻ ¹ for bacterial strains including S. aureus, L. monocytogenes and E. coli. An ethanolic extract of glycyrrhizol B and gancaonin G was also shown to be effective against Gram-positive bacteria with MICs of 1 and 2 μ g ml⁻¹, respectively, and it was suggested that it was the flavonoids present in the extract that led to inhibition of growth. A similar study conducted by Ajagannanavar et al. (2014) further confirmed the inhibitory effects of liquorice on the growth of pathogenic bacteria by using aqueous and alcoholic liquorice root extract (0.2 %) against Grampositive bacteria (Streptococcus mutans and Lactobacillus acidophilus), with results being compared to the activity of the antimicrobial chemical Chlorhexidine.

In this study it found that the alcoholic liquorice root extract showed a higher inhibition of *S. mutans* and *L. acidophilus* compared with both the aqueous form and Chlorhexidine. Rodino *et al.* (2015) tested ethanolic extracts of liquorice (*Glycyrrhiza glabra*) against Gram-positive bacteria (*Staphylococcus aureus, Bacillus cereus* and *Enterococcus faecalis*) and Gram-negative bacteria (*Escherichia coli, Pseudomonas fluorescens*). The inhibition of growth was observed against all bacteria tested with maximum inhibition seen for *E. coli, E. faecalis* and *B. cereus*, while *P. fluorescens* showed the lowest level of inhibition.

Chabuck et al. (2018) investigated the activity of some herbal extracts including Glycyrrhiza glabra in vitro against Gram-positive and Gram-negative bacteria using well-diffusion method, by measuring the diameters of inhibition zones and the results showed that *Glycyrrhiza glabra* exhibited good activity against all bacteria tested with inhibition zones ranging from 18-25 mm, the largest inhibition was against E.coli (25 mm) followed by P. aeruginosa and S. saprophyticus (24 and 23 mm respectively), whereas the lowest inhibition was 8 mm against both k. pneumoniae and E. aerogenes. Another study showed that flavonoids isolated from *Glycyrrhiza glabra* had a strong inhibitory effect against both S. aureus and E. faecalis whereas they showed less inhibitory effects against E. coli and Pseudomonas aeruginosa (Rashid et al., 2013). Hence, depending on the extract used, different patterns of sensitivity were seen with both Grampositive and Gram-negative bacteria being affected. Similar results were reported by Soulef et al. (2014) who tested the in vitro effect of the glycoside extracted from the liquorice plant G. glabra L. against the Gram-positive bacterium S. aureus and the Gram-negative bacteria E. coli and P. aeruginosa and showed that the extract has an inhibitory effect on S. aureus and P. aeruginosa, but no effect on E. coli suggesting that cell structure alone does not account for the pattern of results seen.

When specific components of liquorice were examined, a more consistent pattern was seen. Fukai et al. (2000b) who investigated the antimicrobial activity of nineteen flavonoids commonly found in three species of liquorice which included glabridin and glabrene isolated from Glycyrrhiza glabra, licochalcones A and B isolated from G. inflata and liquiritigenin, liquiritin, licoisoflavone B Formononetin, licoricidin, glycyrol, isoglycyrol, 3-O-methylglycyrol, (3R)-vestitol, licoricone, glycyrin, isolicoflavonol, gancaonol B, glyasperin D and gancaonin I isolated from G. uralensis. They tested these flavonoids against three Gram-positive bacteria (Bacillus subtilis, Staphylococcus aureus and Micrococcus luteus) and three Gramnegative bacteria (Escherichia coli, Klebsiella pneumoniae and Pseudomonas aeruginosa) and found that these compounds showed significant antibacterial activity against S. aureus, B. subtilis and M. luteus but did not have an affect against the Gram-negative bacteria. This study concluded that the methanol extract showed activity against Gram-positive rather than Gram-negative bacteria suggesting that a common difference in cell structure was responsible for the difference seen, and may reflect the fact that the compound could not penetrate the outer membrane of the Gram-negative bacteria since the lipopolysaccharide layer in the outer membrane of Gram-negative bacteria which is known to serve as an active permeability barrier to many environmental substances. Irani et al. (2010) also reported that extracts of G. glabra L. using both paper disc agar diffusion and serial dilution methods and detected a potent activity against Grampositive bacteria S. aureus, B. subtilis and E. faecalis. Irani et al. (2010) and Badr et al. (2011) also found that liquorice extract had a potent bactericidal effect against E. faecalis. Hatano et al. (2000) reported that the main antimicrobial activities of liquorice extract against both methicillin-sensitive and methicillinresistant of S. aureus were due to the phenolic compounds and reported MICs of between 3 and 16 µg ml⁻¹ depending on the strain tested. However, again none of the tested compounds showed antibacterial effects on *E. coli* and *P. aeruginosa*. Another group that investigated the antimicrobial effect of β -Glycyrrhetinic acid isolate from *G. glabra* on both bacteria and fungi found a strong antimicrobial effect against *B. subtilis* and *S. epidermidis* with an MIC of 7.6 and 12.5 μ g ml⁻¹ respectively, while there was no affect against *E. coli, P. vulgaris* and various fungi tested (Kim *et al.*, 2002). These studies are considered as evidence that the effect of liquorice extract is mostly due to flavonoid compounds present in plants, and also that generally the extracts affect Gram-positive bacteria more significantly than Gram-negative bacteria.

Set against these findings is that liquorice and glycyrrhizin have been used for a long time in the treatment of peptic ulcers caused by the Gram-negative bacterium *Helicobacter pylori*. A study conducted by Krausse *et al.* (2004) reported that a total of 29 *H. pylori* strains tested (ca. 79.3 %) were inhibited by low concentrations (\leq 50 mg l⁻¹) including clarithromycin-resistant strains, and therefore liquorice could be considered as the basis for an alternative therapeutic agent against these pathogens. This observation corresponds to another study by Fukai *et al.* (2002a, 2002b) investigated the *in vitro* antibacterial activity of chemical constituents, glabridin, glabrene and licochalcone A isolated from liquorice plants (*G. glabra*, *G. inflata* and *G. uralensis*) against *H. pylori*. Similar results were reported by Malek Jafarian and Ghazvini (2010) who showed that all clinical strains of *H. pylori* tested were sensitive to treatment with *Glycyrrhiza* extract.

Mycobacterium tuberculosis a facultative intracellular bacillus that causes human tuberculosis (TB) and it is considered the number one cause of deaths due to infectious diseases in the world. Friis-Møller *et al.* (2002) reported that several species of *Mycobacterium*, including *M. tuberculosis*, *M. bovis*, *M. kansasii*, *M. xenophii* and *M. marinum* were inhibited by licochalcone A with an MIC \leq 20 mg ml⁻¹ whereas all non-*M. tuberculosis* complex mycobacteria were resistant suggesting that there is a target site specific to this type of cell structure for this compound. Chandran (2015) studied the antibacterial activity of different solvent extracts of liquorice roots against the acid-fast bacterium Mycobacterium smegmatis using the disc diffusion method and found that chloroform extract recorded the maximum anti-mycobacterial activity with an inhibitory zone of 22 mm in hot extract process and 24 mm in cold extract. Another study demonstrated the anti-tubercular activity of an acetone extract of G. glabra against M. tuberculosis H37Rv using Resazurin Microtiter Plate Assay (REMA) and the extract showed significant activity against Mycobacterium tuberculosis H37Rv with an MIC of 0.97–1.95 µg ml⁻¹ (Nair *et al.*, 2015). Similarly Gupta *et al.* (2008) demonstrated antimicrobial activity of extracts of G. glabra against of M. tuberculosis at a concentration of 500 µg ml⁻¹ using BACTEC assay and showed that ethanolic extract from the roots of *Glycyrrhiza glabra* showed antibacterial activity at a concentration of 29.16 μ g ml⁻¹ against two strains of *M. tuberculosis* H37Ra and H37Rv and subsequent phytochemical analysis identified the active component to be glabridin which was more active against Gram-positive bacteria than Gram-negative. This group also found that an ethyl acetate extract containing glabridin at a concentration of $100-250 \ \mu g \ ml^{-1}$ showed good activity against these acid fast bacteria that are notoriously resistant to antimicrobial compounds due to their thick, waxy cell walls that contain high levels of mycolic acid (Gupta et al., 2008).

Geetha and Roy (2012) investigated the antimicrobial activities of both ethanol and aqueous liquorice (*G. glabra*) root extracts against *Staphylococcus aureus* (MRSA) using the agar well diffusion method at a concentration of 500, 1000 and 2000 μ g ml⁻¹. The ethanolic extract showed a zone of inhibition of 24 mm diameter at concentration of 2000 μ g ml⁻¹ against *staph. aureus*. Another study focussing on bacterial pathogens which cause oral infections and dental caries by Sedighinia and Afshar (2012) tested the antibacterial activities extracts of liquorice (*G. glabra*) *in vitro* using the agar dilution method and again showed effective inhibitory activity against the microorganisms tested, with an MIC of 12.5

mg ml⁻¹ for *S. mutans, Actinomyces viscosus* and *E. faecalis*, no effect seen on *E. coli* which exhibited the most resistance to *G. glabra* (Sedighinia and Afshar, 2012).

The most problematic Gram-positive bacteria in public health and healthcare-related infections is *S. aureus*, not only due to the fact it is widespread but also because it has become more resistant to most of antibiotics available (Chambers, 2001). Especially in hospitals and the communities the Methicillinresistant S. aureus (MRSA) has become a principal source of infection (Long et al., 2013). The increase in antibiotic resistance in *S. aureus* strains has led to a search for other therapies to treat it. Consequently, a wide variety of bioactive phenolic constituents have been isolated from liquorice and the antimicrobial activity of liquorice flavonoids have been confirmed against methicillin-resistant S. aureus (Fukai et al., 2002b; Shibata, 2000). The antimicrobial activities of 70 % ethanolic extracts among of 19 plant were tested against 59 methicillin-resistant S. aureus (MRSA) using the agar well diffusion method, and liquorice (*Glycyrrhiza Glabra*) ethanolic extract exhibited the highest antimicrobial effect against all MRSA isolates with MICs of 8 μ g ml⁻¹ (Bassyouni *et al.*, 2012). Hatano *et al.* (2000) also reported that two phenolic compounds, Glicophenone and Glicoisoflavanone present in G. glabra had a strong effect against both methicillin resistant S. aureus (MRSA) and methicillin-sensitive *S. aureus* with an MIC of 16 μ g ml⁻¹.

Licochalcone A extracted from *G. glabra* had potent activity against different types of Gram-positive bacteria and the effective MICs of Licochalcone A was 5 mg ml⁻¹ for *Lactobacillus acidophilus* and *Lactobacillus plantarum*, 6 mg ml⁻¹ for *Enterococcus faecalis* and *Enterococcus faecium*, as well as 8 and 5 mg ml⁻¹, respectively, for *Streptococcus lactis* and *Staphylococcus mutans* (Nowakowska, 2007). Hao *et al.* (2013) also reported that Licochalcone A isolated from the roots and rhizomes of *Glycyrrhiza inflate*, inhibited the biofilm formation of *Streptococcus suis* with MICs of 4 mg ml⁻¹ for *S. suis* serotype 2 strains and 8 mg

ml⁻¹ for *S. suis* serotype 7 strains using the microtiter plate assay. Sultana *et al.* (2010) also found that methanolic extracts of *G. glabra* showed activity against both Gram-positive (*S. aureus, Bacillus megaterium* and *B. subtilis*) and Gramnegative (*E. coli, P. aeruginosa* and *Salmonella paratyphi*) bacteria using the disc diffusion method. Most organisms were found to be sensitive to the extract except *Pseudomonas aeruginosa*. However it was most effective against *S. aureus* which produced the largest zones of inhibition of 22 mm in a disc diffusion assay. Activity against these microorganisms was due to the presence of glabrene, licoisoflavone B, isolicoflavonol, and gancaonin I.

The overall conclusion of this research is that active ingredients are present in liquorice, and that essential oils, triterpenoids, alkaloids, phenols, polyamines and polysaccharides have potent antibacterial activities. Generally more effect is seen against Gram-positive bacteria, presumably reflecting the difference in cell wall structure of these two groups of bacteria. However activity against Gramnegative bacteria has been seen, and Glabridin isolated from liquorice has been documented to shown numerous of pharmacological activities shown to equally affect antimicrobial activity against methicillin-resistant *S. aureus* (Fukai *et al.*, 2002b; Hatano *et al.*, 2000) and *H. pylori* (Fukai *et al.*, 2002a) suggesting that there are multiple mechanisms by which the different bioactive compounds present in liquorice affect bacterial cells.

1.5.2 Toxicity of liquorice plant

The most important factor during the search for new antimicrobial agents from plants is a consideration of the toxicity of plant extracts and plant components toward cells, especially mammalian cells. One way to ensure the function of the microbial agents is safe to use as an antibacterial is through inclusion of a parallel evaluation on host cell lines (cytotoxicity evaluation) and other microbial screens on bacteria, fungi, parasites and viruses (Cos *et al.*, 2006). Although the liquorice plant is in general considered safe, hypertension is one of the most usually recorded side effects associated with liquorice consumption. Accordingly, based on data collected between 2006 and 2010 for the European and Brazilian poison centres, liquorice was reported as being one of the ten food plants most frequently reported to have caused harmful side effects (Lüde *et al.*, 2016). These adverse effects are probably due to the effect that liquorice has on the renin-angiotensin-aldosterone system, causing salt retention and hypokalaemic hypertension associated with overconsumption of liquorice saponins (White, 2001). Thus, the overconsumption of these products is associated with raised risk of stroke or heart disease (Johns, 2009) since excessive consumption can produce the symptoms of hypertension due to increasing the absorption of sodium and water in kidneys, however, the symptoms usually disappeared when the dose is reduced (Miettinen *et al.*, 2010).

According to Renaissance physician Paracelsus (1493-1541) the treatment dose is responsible for the safety of substances, where the right dose distinguishes between a poison and a therapy which means that just the dose determines that if the substance is poisonous or not - otherwise all substances may be considered as poisons (Grandjean, 2016). Therefore, to ensure the efficacy and safety in foods and clinical use, strict control on liquorice is needed (Zhang and Ye, 2009). For instance, liquorice root can be used safely for therapy of duodenal and gastric ulcers, however, deaths from its excessive use have been reported. There is a recommendation that due to reported side effects, an average daily dose should not exceed 5–15 g of dried root, which equates to 200–600 mg glycyrrhizin, for the treatment of gastrointestinal ailments, with treatment course not continuing for more than 4–6 weeks (Isbrucker and Burdock, 2006).

The US FDA, the Council of Europe, and the UK Food Additive and Contaminants Committee recognized liquorice and liquorice extract are safe for use in foods, but it is recommended by the European Scientific Committee on Food that consumption levels do not exceed 3.5 g/day of liquorice extract which corresponds to about 100 mg/day of glycyrrhizin (Scientific Committee on Food, European Commission, 2003). Thus, the FDA assumes that glycyrrhizin levels in foods do not pose a health hazard, provided that these foods are not consumed in excess or by individuals who are sensitive to low levels of glycyrrhizin (Delbò, 2013). According to The World Health Organisation the consumption of 100 mg/day would not be to cause adverse effects, whereas The Dutch Nutrition Information Bureau advised against daily glycyrrhizin consumption in excess of 200 mg/day (Sontia et al., 2008). However, the maximum acceptable amount of liquorice or its extract still unknown. Although severe toxicity of liquorice extract is very rare, and the LD_{50} in mice and rats is greater than 4 g glycyrrhizinate/kg, which is well below the recommended safe use level for foods, however, overconsumption levels of glycyrrhizin may cause pseudoaldosteronism (Isbrucker and Burdock, 2006). Despite this, the major component of liquorice and liquorice derivatives glycyrrhizin or glycyrrhizic acid, has been approved as Generally Recognized as Safe (GRAS) for use in foods by the U.S. FDA.

1.6 Aims and Objectives

The aim of this study was to assess a waste material from the production of commercial liquorice extract for its antimicrobial activity against the Grampositive bacteria including *L. monocytogenes* strains, *L. innocua*, *B. subtilis*, *Staph. aureus*, *E. faecalis* and *P. aeruginosa*.

Hence the particular aims and objectives of the study were:

 To investigate the potential of plant extracts as natural antimicrobial compounds. In particular to determine if the compounds had effectiveness against food borne pathogens including *L. monocytogenes*.

- 2) To determine what the antimicrobial mechanism of action of the extract
- To determine the MIC for the bacteria that were sensitive to this material and therefore too establish an effective working of concentration of extract.
- 4) To investigate the physiological effects of the extract on bacterial cells.
- To evaluate the ability of the extract as a novel food preservative in a model food system.

CHAPTER 2

MATERIALS, METHODS AND STANDARD PROCEDURES

2.1 Chemicals and media

Details for the preparation of solutions and media used in this thesis are presented in Appendix 1 and 2. The suppliers for all chemicals substances and media used in this thesis are listed in Table 2.1.

Table 2.1. Chemical substances and media used during this study

Compound	Supplier
Brain Heart Infusion Broth (CM 1135)	Oxoid, UK Ltd
Agar Bacteriological (Agar No.1) LPOOII	Oxoid, UK Ltd
Nutrient Broth No.2 (IVD) (CM0067)	Oxoid, UK Ltd
Nutrient Broth (CM0001)	Oxoid, UK Ltd
Maximum Recovery Diluent (MRD) (CM0733)	Oxoid, UK Ltd
Palcam agar base (CM087)	Oxoid, UK Ltd
Listeria selective agar Base (CM0856)	Oxoid, UK Ltd
Tryptone soya broth (IVD) (CM0129)	Oxoid, UK Ltd
Tryptone soya agar (IVD) (CM0131)	Oxoid, UK Ltd
phosphate buffered saline	Oxoid, UK Ltd
Listeria Selective Agar Supplement (SR0140)	Oxoid, UK Ltd
Listeria Palcam Selective Supple (VWR), B-3001	Leuven, Belgium
Buffered peptone water (ISO) (F0213W)	VWR,Prolabo chemicals
D-Glucose, anhydrous	Fisher Scientific, UK Ltd
Potassium phosphate, dibasic, anhydrous (K2HP04)	Fisher Scientific, UK Ltd
Sodium ammonium phosphate (NaHNH ₄ PO ₄ .4H ₂ O)	Fisher Scientific, UK Ltd
Sodium chloride (NaCL)	Fisher Scientific, UK Ltd
Magnesium sulphate (MgS0 ₄ .7H ₂ 0)	Fisher Scientific, UK Ltd
Glacial acetic Acid	Fisher Scientific, UK Ltd

Table 2.1. Continued

Dimethylformamide	Fisher Scientific, UK Ltd
Erythromycin	ApoLLo Scienctic Limited
Ampicillin Sodium Salt BioChemica	PanReac AppliChem
Tetracycline	Sigma-Aldrich
D-Biotin (F.W.244.3)	Sigma-Aldrich
Citric acid Anhydrous (C ₆ H ₈ O ₇) (C-0759)	Sigma-Aldrich
Sodium dodecyl sulfate, for molecular biology, approx. 99% (L4390)	Sigma-Aldrich
Mitomycin C from Streptomyces. Vial contains 2 mg mitomycin C and 48 mg Nacl (M-0503)	Sigma-Aldrich
Sodium azide (S-2002)	Sigma-Aldrich
Sodium dodecyl sulfate	Sigma-Aldrich
Trypsin – EDTA Solution (T3924)	Sigma-Aldrich
Mem Non-essential amino acid solution (100x) (M7145)	Sigma-Aldrich
L-Glutamine solution (G7513)	Sigma-Aldrich
foetal bovine serum (FBS) (F9665)	Sigma-Aldrich
Dulbecco's Modified Eagle Medium (DMEM) high glucose (D6571)	Sigma-Aldrich
Trypan Blue solution	Sigma-Aldrich
Dimethyl sulphoxide (DMSO)	Sigma-Aldrich
L-Histidine. HCL. H ₂ 0 (F.W.209.63)	Acros Organics
Isopropanol, 99.5+%, pure	Acros Organics
3-(4,5-dimethylthiazol-2-yl)-2,5- diphenyltetrazolium bromide (MTT)	Acros Organics

2.2 Bacterial strains, plasmids, and growth conditions

Bacterial strains and plasmids used in this study are listed in (Tables 2.2 and 2.3), respectively. These strains were obtained from the laboratory stocks of Food Sciences Division, University of Nottingham, United Kingdom. *Listeria* strains were recovered from Microbank[™] beads held at -80 °C by inoculating into Brain Heart Infusion (BHI) broth and then plating onto BHI agar with erythromycin (Erm; 5 µg ml⁻¹) to select for the presence of the plasmids when required. Plates were incubated for 24 h at 37 °C and stored at 4 °C until needed. *Staphylococcus aureus* and *Pseudomonas fluourescens* strains chromosomally tagged with a *lux* operon (*luxABCDE*) that was modified to function in both of them.

Salmonella strains for the Ames test carried different mutations in various genes in the histidine operon. When needed, the culture medium was supplemented with antibiotics at the following concentrations; 24 µg ml⁻¹ ampicillin for *Salmonella* Typhimurium TA1537, TA100 or 24 µg ml⁻¹ ampicillin and 2 µg ml⁻¹ tetracycline for *Salmonella* Typhimurium TA102.

In general, the cultures were incubated overnight at optimal temperatures dependent on the type of bacterium. An isolated colony was inoculated in the broth supplements with antibiotic as required. With most organisms in this study, 24 h incubation period is adequate, but an additional time was needed for the slow growing microorganisms.

2.3 Preparation of liquorice extract

3 mg of liquorice powder was weighed into a sterile bijou bottle using scales accurate to 0.1 mg then dissolved in 1 ml of 70 % (v/v) ethanol solvent, and mixed thoroughly before incubating at 37 °C for 30 min to ensure the extract dissolved completely then stored at -20 °C prior to use.

2.4 Preparation of bacterial cultures

2.4.1 Listeria monocytogenes strains

Listeria strains from frozen stocks were grown by streaking one bead into brain heart infusion (BHI) agar plates and incubated at 37 °C for 24 h. From these plates one colony of each strain was transferred into 10 ml BHI broth and allowed to grow for 24 h (overnight) at 37 °C in a rotary shaker at 150 rpm. Growth was routinely measured as optical density at 600 nm (OD_{600nm}). In addition, the number of culturable cells was determined by counting colonies grown on BHI agar plates supplemented with erythromycin (5 µg ml⁻¹), cultures were also always plated onto BHI agar without antibiotics to ensure that the cultures were not contaminated and to ensure marker stability. The inocula were diluted to the appropriate cell densities ($OD\sim0.05$) in BHI broth.

2.4.2 Salmonella strains/mutants

Strains stored at -80 °C were streaked into brain heart infusion (BHI) agar to isolate pure isolated colonies and incubated at 37 °C for 18-24 h subsequent one colony was transferred into 5 ml of nutrient broth then incubated at 37 for 18-24 h in shaking incubator with aeration (150 rpm) to reach approximately 1×10^9 cfu ml⁻¹.

2.4.3 Pseudomonas fluourescens chromosomally tagged lux

P. fluorescens chromosomally tagged *lux* (see Table 2.2 for details) was maintained by streaking the strain on nutrient agar plates which were incubated at 30 °C for 24 h. The grown cultures were stored at 4 °C, one colony was grown in nutrient broth at 30 °C for 18-24 h in shaking incubator with 150 rpm.

Bacteria	Strain numbers	Gram reaction
	or source/reference	group
Enterococcus faecalis	NCTC 775	+ve
Staphylococcus aureus	Chicken isolate,	+ve
	Nottingham lab collection	
Bacillus subtilis	var. Niger 168	+ve
Listeria innocua	Factory isolate,	+ve
	Nottingham lab collection	
Listeria innocua	ATCC 11994 (Serotype 6a)	+ve
Listeria monocytogenes	10403S (wild type; 1/2a)	+ve
	(Bishop and Hinrichs, 1987)	
Listeria monocytogenes	EGD (wild type; 1/2a)	+ve
	(Murray <i>et al.</i> , 1926)	
Listeria monocytogenes	ATCC 23074 (wild type; 4b)	+ve
Listeria monocytogenes	NCTC 10357 (Δ <i>prfA</i>)	+ve
Listeria monocytogenes	NCTC 7973 (<i>prfA</i> *)	+ve
Staphylococcus aureus	Xen 29	+ve
	Tn4001 luxABCDE	
	(Xiong <i>et al.,</i> 2005)	
Pseudomonas fluorescens	Field isolate, Nottingham lab.	-ve
	Collection	
	mini-Tn5 (gfp:luxABCDE, Tet ^r)	
	Dr P. Hill, University of	
	Nottingham, (unpublished)	
S. enterica serovar	Nottingham lab. Collection	-ve
Typhimurium TA102	Deletion mutation (Levin <i>et al</i> .,	
	1982)	
S. enterica serovar	Nottingham lab. Collection	-ve
Typhimurium TA100	Deletion mutation (McCann et al.,	
	1975)	
S. enterica serovar	Nottingham lab. Collection	-ve
Typhimurium TAI535	Deletion mutation (Ames et al.,	
	1973b)	

Table 2.2. Bacterial strains and yeast used in this study

Table 2.2. (continued)

Mycobacterium smegmatis	mc ² 155; ATCC 700084	Acid Fast
Escherichia coli	NCTC 86	-ve
Pseudomonas aeruginosa	Nottingham lab collection	-ve
Saccharomyces cerevisiae	NCYC 363/ Brewing Science	Lab isolate strain
Saccharomyces cerevisiae	S288C/ Brewing Science	Wild type

*ATCC: American Type Culture Collection

**NCTC: National Collection of Type Cultures

***NCYC: National Collection of Yeast Cultures

Table 2.3. Bacteria with plasmus used in this study

	Bromotor and		Antibiotic
Bacteria	Promoter and	Vector	selection
	Insert		[µg ml⁻¹]
L. innocua	BS10, luxABCDE, afp	pUNK1	Erm [5 µg ml ⁻¹]
L. monocytogenes	BS10 luxABCDE, gfp	pUNK1	Erm [5 µg ml ⁻¹]
<i>L. monocytogenes</i> NCTC 10357	xylA, luxABCDE, gfp	pUNK1	Erm [5 µg ml ⁻¹]
<i>S. enterica serovar</i> Typhimurium TA100	hisG46	pKM101	Amp [24 μg ml ⁻¹]
<i>S. enterica serovar</i> Typhimurium TA1535	hisG46	No plasmid	None
<i>S. enterica serovar</i> Typhimurium TA102	hisG428	pKM101, pAQ1	Amp [24 μ g ml ⁻¹] and Tet [2 μ g ml ⁻¹]
Pseudomonas fluourescens	mini-Tn5 (gfp:luxABCDE, Tet ^r)	None	Tet [2 µg ml ⁻¹]
<i>Staphylococcus aureus</i> Xen 29	Tn4001 luxABCDE Perkin Elmer		Kan [5 µg ml ⁻¹]

*Amp (Ampicillin), *Tet (Tetracycline) & *Ery (Erythromycin)

Plasmid	Description	Reference
pSB3007	pUNK1,PxyIA:gfp:luxABCDE:rrnBT1T2	Perehinec <i>et al</i> ., 2007
pSB3008	pDEST-pUNK1 containing	R. Gaddipati, 2010, PhD
	P _{BS10} ::gfp3:luxABCDE::rrnBT1T2	Thesis, University of
		Nottingham

Table 2.4. Plasmid used in this thesis

Table 2.5. Human tissues used in this research

Cell line	Description	Source
Caco-2 cells colon	Human Caucasian colon adenocarcinoma	ATCC HTB-37

Table 2.6. Chemicals used as a positive control in the Ames test

Bacterial strain	Chemical
S. Typhimurium TA100	Sodium azide
S. Typhimurium TA102	Mitomycin C
S. Typhimurium TA1535	Sodium azide

The chemical required for every strain is different due to the different mutagenicity in gene operon for all the strains, where sodium azide serves as a positive control for TA100, TA1535 and mitomycin C for TA102.

2.4.4 Staphylococcus aureus chromosomally tagged lux

Staphylococcus aureus chromosomally tagged with *lux* was maintained by streaking the strain on nutrient agar plates which then were incubated at 37 °C for 18-24 h. Subsequently one colony was grown in nutrient broth at 37 °C for 18-24 h in shaking incubator with 150 rpm.

2.5 Preparation of yeast cultures

Two strains of *Saccharomyces cerevisiae* (S288C, NCYC 363) were used in this study. Yeasts were cultured at 30 °C in YPD media, containing (2 % peptone, 1 % yeast extract, 2 % dextrose) for 48 h in shaking incubator with aeration at 230–270 rpm until the OD_{600nm} was in the range of 0.4–0.6.

2.6 Monitoring growth of bacterial cultures

After an adequate period of incubation, culture growth was determined using Optical Density (OD_{600nm}) and the density of each sample was adjusted to a specific level as required. In all tests bacterial cultures were diluted an initial concentration of ~10⁷ cfu ml⁻¹ ($OD_{600nm} \simeq 0.05$) whereas, *Salmonella* strains were adjusted to give a final density of 1X10⁸ cfu ml⁻¹ (approximately $OD_{600nm} = 0.5$). For the diluted cultures, the samples were used within 30 min to initiate the experiments.

2.7 Frozen stock cultures of bacteria

All strains used in this study were stored long term at -80 °C by preparing frozen stocks. Cultures were grown on nutrient or BHI agar plates dependent on the type of bacteria (see Section 2.4). The frozen stocks were prepared by using a sterile inoculation loop to pick off enough colonies from a pure overnight culture into cryovials containing cryopreservative fluids (Microbank[®], Pro-Lab Diagnostics U.K). The inoculated vials were tightly closed and inverted 4-5 times to mix the bacteria to allow them to bind to the beads. By using Gilson pipettes the excess cryopreservative was removed from the vials, then the vials were kept at -80 °C. When a fresh culture was prepared, one bead was picked up from the vial by using a sterile tooth then streaked four-way streak plate onto the appropriate agar plates. For short term storage bacteria cultures were kept on BHI agar plates at 4 °C and maintained by sub-culturing on BHI agar plates every 4 weeks.

2.8 Measurement the optical density of bacteria

The bacterial strains used for this study were *E. faecalis*, *S. aureus*, *B. subtilis*, *L. innocua*, *L. monocytogenes* (Gram-positives) and *E. coli* and *P. aeruginosa* (Gram-negatives), one colony inoculated into 10 ml of broth media followed by incubation overnight at 37 °C in 150 rpm shaking incubator then mixed thoroughly using a vortex mixer, cell growth was monitored with a spectrophotometer (Eppendorf Biophotometer Plus - Thermo Fisher Scientific, Germany) at OD_{600nm}, 1 ml of overnight culture was inoculated into sterile 250 ml Erlenmeyer flasks containing 100 ml of BHI broth then 1 ml of samples was transferred into a plastic cuvette for optical density measurement, the inoculum of the broth culture was adjusted until the bacterial suspension reached an OD_{600nm} 0.05 which was equivalent to an approximately 1×10^7 cfu ml⁻¹.

2.9 Determination of the growth rate of Gram-positive bacterial cultures

To measure the growth rate of bacteria, a single colony was transferred into 10 ml of BHI broth and grown at 37 °C overnight in shaking incubator (150 rpm), then different dilutions were made for overnight cultures (1:10, 1:50 and 1:100) in broth media. Dilution cultures (200 µl) were transferred into wells of a

microtitre plate (at least 4 wells) of sterile 96-well polystyrene microtiter plates. Culture growth was monitored by using optical density at a wavelength of OD_{600nm} with readings taken every 10 min for 24 h, readings were taken at regular periods over the course of the experiment as required. The growth rate of the bacteria was calculated using the equation shown below (Widdel, 2007).

The specific growth rate (μ)

$$\mu = 2.303 (\log_{10} N_x - \log_{10} N_o) (t_x - t_0)$$

Where:

e: N_0 = initial number of cells at t_0 N_x = number of cells at time t_x t_o = time at the beginning of log phase t_x = time at a later time

2.10 Tecan microplate reader use for measuring bacterial growth

To allow multiple growth experiments to be carried out, a microtitre method was developed. This method was used to measure the optical density of cultures as well as luminescence. The sterile polystyrene microtitre plates contained 96-wells (Krystal Microplate, clear bottom or Porvair 96 well black, Porvair Sciences, England), and had a capacity of 200 μ l per well. For bioluminescence or fluorescence measurements black plates were used as they reduce the background signal. For optical density measurements clear plates were used.

2.11 Minimum inhibitory concentration determination

The MIC of liquorice extract was determined by a microdilution broth method performed as described by Motyl *et al.* (2006) with a slight modification, using 96 well sterile plates which have a maximum capacity of 200 μ l. Overnight cultures were prepared and aliquots inoculated into sterile 250 ml Erlenmeyer flasks containing 100 ml of BHI broth. The inoculum of the broth culture was adjusted until the bacterial suspension reached an OD_{600nm} 0.05 which was equivalent to an approximately 1×10^7 cfu ml⁻¹. Three flasks were prepared for each type of bacteria; the first flask (control) contained 1 % (v/v) ethanol, the second flask contained 12.5 µg ml⁻¹ of liquorice extract and the third flask contained 50 µg ml⁻¹ of liquorice extract.

Samples of the cultures (200 μ I) were transferred into the wells of a microtitre plate using at least 12 wells from a sterile 96-well polystyrene microtiter plates for each bacterium. These were tested at two different concentrations (12.5 and 50 μ g ml⁻¹) of liquorice extract. Cell suspensions with 1 % (v/v) ethanol and BHI broth without culture were used as a controls. The microtiter plate was then incubated at 37 °C for 24 h in a Tecan Genesis Pro microplate reader. Cell growth was monitored by using OD_{600nm} with readings taken every 10 min for 24 h, readings were taken at regular periods over the course of the experiment as required.

2.12 Miles and Misra technique

The minimum bactericidal concentrations (MBC) were determined using a modified Miles and Misra method (Miles *et al.*, 1938) to determine viable counts (cfu ml⁻¹) of a bacterial suspension in Brian Heart infusion (BHI) broth treated with liquorice extract (Section 2.11), 10 μ l of treated samples and controls were 10-fold serially diluted into maximum recovery diluent (MRD) with a maximum dilution factor of 10⁻⁵. Petri dishes contain BHI agar media were divided into 3 sectors 10⁻³, 10⁻⁴ and 10⁻⁵, 5 replicate 20 μ l samples per dilution (total sample volume = 100 μ l) were dispensed onto the surface of BHI agar plate, in triplicate, then allowed to dry into the agar plate next to a Bunsen burner. Once dried, the BHI agar plates were incubated at 37 °C for 24 h to determine the number of viable bacteria. The colonies were counted after overnight incubation using a

Coulter counter (Digital colony counter, Stuart SC6 Colony counter - Stuart Equipment, protected by BioCote, UK). For counted populations of *L. monocytogenes* counts were made on drops showing between 3 and 30 colonies, data were converted into a log (cfu ml⁻¹) using the following formula:

No. CFU/ml = No. (In countable range) $\times \frac{1}{dilution \ factor} \times \frac{1}{volume}$

2.13 Gram-staining method

The Gram staining procedure involved placing a loopful of maximum recovery diluent (MRD) on a glass microscope slide. By using a 1 µl inoculation sterile loop, a drop of overnight culture from the test of assessment of cell morphology (Section 3.2.6) was mixed with the MRD and the sample was emulsified. The drop was left to dry for about 30 s and then fixed using a Bunsen flame by passing briefly in the flame 2 to 3 times. The slides were placed on a rack and then immersed in primary stain (methyl violet: purple stain) for 1 min. Excess stain was rinsed off with water, then the slide was placed into mordant (Lugols iodine) for 30 s, and then excess stain was washed off with water and slides were then placed in iodine (mordant) for 30 s and then excess was rinsed off with water. After that the slides were immersed in decolourising agent (alcohol) for 1 min and washed in water. Slides were then placed into secondary stain/counter stain (carbol fuchsin: red stain) for 30 s and excess stain was washed in water. Slides were dried, then cells were imaged using a standard light microscope after Gramstaining using a x100 oil immersion lens. Images were captured using microscope with camera attached to a computer, and also with using electronic microscope (Nikon) connected to a Cannon camera (Canon EOS 700D, Japan).

2.14 Recovery of plasmid containing strains from frozen culture

One colony of *Listeria* strains carrying the *Lux* gene (see Table 2.2 and 2.3 for details) from -80 °C were streaked onto brain heart infusion (BHI) agar to obtain pure isolated colonies and incubated at 37 °C for 18-24 h. All colonies arising on agar plates could be visualized both by the *GFP* phenotype (as fluorescent colonies) microscopic and by the luciferase phenotype (as light-emitting colonies) luminometry. The distinction between fluorescent and non-fluorescent colonies was examined using the Biospace lab photon imager (Biospace Lab, France) for Gfp to ensure the present of plasmid.

2.14.1 Confirmation of light emission (luminometry)

The frozen stored *Listeria* strains were reactivated in BHI agar plates and incubated for 24 h at 37 °C. A colony was transferred into 5 ml of (BHI) broth which contained 5 µg ml⁻¹ erythromycin for plasmid selection, as appropriate for 24 h at 37 °C in shaking incubator (150 rpm), the strains were analysed in a Luminometer (Model TD-20e Luminometer, Turner designs) by transferring 3 ml of overnight cultures into luminometer tubes to measure the light emitted. To prepare cultures for monitoring luminescence during growth, 1 ml of the overnight culture was added to 49 ml of fresh BHI broth (1:50) and it was grown in a shaking incubator (approx. 150 rpm) at 37 °C until mid-exponential phase growth (approx. 3-4 h). In each case strains that did not contain a plasmid were used as a negative control. After 4 h, 3 ml of samples were transferred into tube and again the light was measured in the TD-20e Luminometer.

2.14.2 Confirmation of fluorescence phenotype (GFP)

One colony of *Listeria* strains carrying the *lux* genes from a -80 °C stock was streaked onto a BHI agar plate than incubated for 24 h at 37 °C, then the

plates investigated using the Biospace lab photon imager, the colonies that were bright was used in this study.

2.15 Live/Dead stain

Bacterial cell membrane integrity was determined using the LIVE/DEAD[®] BacLight[™] Bacterial Viability Kit (Molecular Probes; catalog no. L-7012), for microscopy and quantitative assays. The stock solution of the two stains was prepared according to the protocol issued by Molecular Probes (Mol. Probes, 2004) by adding 2 µl of SYTO 9 and 2 µl of propidium iodide (PI) in equal proportions then mixed and diluted 1:100 in 0.85 % NaCl. To carry out a microscopic assessment of membrane integrity, 60 µl of the sample (Section 3.2.10) was mixed with 18 µl of dyes mixture in an Eppendorf tube. Samples were incubated in the dark for 15 min at room temperature, then 5 µl of the stained samples was placed between a glass slide and a cover slip which was examined after a further 10 min incubation. Fluorescent cells were observed at the excitation and emission wavelengths of 480/500 nm for SYTO 9 stain and 490/635 nm for propidium iodide by using a fluorescence microscope (Nikon) equipped with an UV lamp and a 100 X magnification objective, visualized by using a digital Canon camera (Canon EOS 700D, Japan) coupled with fluorescence microscope at 100x magnification.

2.16 Evaluating the activity of liquorice extract on bologna-style sausage

Bologna sausages (Morliny Polish Doktorska, 450 g) were purchased from a retail supermarket (ASDA, UK). The inoculation of bologna with *L. monocytogenes* was done according to the method described by Lara-Lledó *et al.* (2012) with some modifications. Bologna-style sausages were sliced (~3 mm) in the laboratory using a food slicer and inoculated by dipping for 30 s in *L. monocytogenes* with and without 50 µg ml⁻¹ liquorice extract to achieve an initial inoculum level of approximately 4 log cfu ml⁻¹. The dipped slices were then allowed to dry for 30 min to remove excess liquid.

The inoculated slices were stored at 6 °C and samples (n = 3) removed for sampling at day 0 (slicing) and thereafter at day 5, 10 and 15. At sampling, slices were placed in a stomacher bag (177 mm x 305 mm) with 0·1 % (w/v) BPW and the blend was mixed for 60 s at 300 RPM speed in a Stomacher Laboratory blender (Stomacher 400, Seward Limited, UK) to distribute the *Listeria* equally. The mixture was serially diluted (1:10) with MRD, and the level of *Listeria* in the samples was determined by selective enumeration by plating appropriate dilutions (10⁰, 10⁻¹, 10⁻², 10⁻³ and 10⁻⁴) on Palcam Agar to selectively enumerate *Listeria*. This was done by adding 10 µl treatment or control samples to 90 µl maximum recovery diluent (MRD). Then Miles and Misra technique as describe in Section 2.12 was used to determine the viable count.

2.17 Mutagenicity of liquorice extract (Ames test)

The strains used in the experiment were provided from the University of Nottingham Microbiology and Food Saftey culture collection by Dr Jon Hobman. Overnight cultures were grown by transferring a single colony into a Universal tube containing 5 ml of Nutrient broth no. 2 with loose fitting caps. The inoculated tubes were incubated at 37 °C for 16-18 h and cultures were shaken at approximately 200 rpm to insure sufficient aeration and to reach a density of ~1.0 to 2.0 x 10⁹ cfu ml⁻¹ (OD₅₄₀ between 0.1 and 0.2), 2 ml of molten top agar supplemented with limited histidine and biotin at a temperature between 43 °C and 48 °C to avoid killing of the strains was added to 0.1 ml of bacterial suspension with and without 50 µg ml⁻¹ of liquorice extract. The contents of the test tubes were then mixed and poured onto the surface of glucose minimal (GM) agar plates then spread over the surface of the agar. The negative control was conducted by replacing the extracts with 1 % (v/v) ethanol solvent.

The antibiotics required for each strain are described in Appendix 2. These were added directly to the GM media after autoclaving and mixed in the agar, for the positive control 0.1 ml inoculum and 2 ml molten top agar with biotin and a trace of histidine were added to a bijou bottle, then spread over the surface of the GM agar. Under aseptic conditions a sterile filter paper disc (6 mms, Whatman) was placed in the centre of each MG agar plate then 10 µl of the positive control (mutagenic) solution sodium azide or mitomycin C (depending on the strain) was dispensed to the disc (Table 2.6).

When the top agar was set (2-3 min), the plates were inverted and covered with aluminium foil to protect them from the effects of light on photosensitive chemicals and incubated at 37 °C for 48 h.

2.18 Cell culture conditions

2.18.1 Maintenance of cell culture

Human Caucasian colon adenocarcinoma (Caco-2) cells were obtained from the Food Science Department, University of Nottingham, United Kingdom. The stock of Caco-2 cells were kept in liquid nitrogen tank. 1 vial containing 1 ml of the cells was defrosted and maintained in complete medium Dulbecco's modified Eagle's Medium (DMEM) High Glucose (D6571, Sigma), supplemented with 10 % fetal bovine serum (FBS) (F9665, Sigma), 1 % L-Glutamine solution (G7513, Sigma), 2 % non-essential amino acid solution (100x) (M7145, Sigma), cells were grown in T-75 tissue culture flasks (Corning), cells were incubated at 37 °C in humidified atmosphere of 5 % CO₂ (SANYO CO₂ Incubator) and allowed to adhere for 24 h and the cell lines were maintained by changing medium every 2-3 days. Cell culture was passaged when cells reached to required confluence (approximately 80 % confluence) by removing medium and rinsing with 15 ml of Phosphate Buffer Saline (PBS), then the PBS was removed. The adherent cells were removed by treating with 2 ml of trypsin- EDTA solution and incubating for

2 min at 37 °C to detach the cells from the surface. Then the cells were examined under the microscope to see separate cells after treatment with trypsin, 8 ml of fresh pre-warmed media was then added to the flask to inactivate trypsin. Cell clumps were disrupted with gentle pipetting up and down on the plate to help the cells to detach to break the cell clumps, then the contents of the flask were transferred into a 50 ml Falcon tube (Sigma-Aldrich), the cells were centrifuged using centrifuge (Thermo Scientific) at 500 x *g* for 5 min and the supernatant was discarded to obtain a cell pellet and cell pellets were re-suspended in 8 ml of fresh medium, again small amount of suspension cells were transferred into a new flask with 8 ml of medium. Cell density were counted using a haemocytometer as a described in Section 2.18.2.

2.18.2 Counting cells with a Haemocytometer

Viable Cells were assessed by trypan blue solution (Sigma, UK), then cells were counted by using a Haemocytometer (Neubauer's chamber). Before each use, the haemocytometer was washed with distilled water and wiped with 70 % (v/v) ethanol. The sample solution was prepared by mixing 50 µl of cell suspension with 50 µl trypan blue into a micro-centrifuge tube, then 50 µl of cells suspension were transferred by a Gilson pipette into haemocytometer. The cells suspension was dropped from the pipette to fill the upper and lower counting chamber. The haemocytometer was then examined under an inverted microscope (PBA Microscopes). Cells were counted in the square surrounded by triple ruled lines in the centre of the haemocytometer which divided into a grid pattern, consisting 25 large squares. Each squares contained 16 sub squares, and cells which touched lines were only counted if on the top or left line of the square and using the contrast to distinguish the cells and count both live and dead cells, the dead cells will appear blue and live cells will appear clear (white). A total of 25 large squares

in the centre were counted in the upper and lower chamber and the average number from these multiplied by 10^4 giving number of cells ml⁻¹.

2.18.3 Determination of cell viability (MTT assay)

Cell viability was determined using the protocol described by Riss *et al.* (2016) with slight modification. Briefly, Caco-2 cells were seeded at a density of 0.3×10^5 cells per well, 5 ml of DMEM with cells was placed at every well in a 6-well plates, cells were treated with two different concentrations of liquorice extract: 12.5 and 50 µg ml⁻¹, 1 % (v/v) ethanol solvent and 4 % vinegar (acetic acid) were used as a negative control and sodium dodecyl sulfate (SDS) (20 %) was used as a positive control. Untreated cells with no extract added were also tested. Triplicate wells were prepared for each individual treatment. Cells were kept in the incubator for 24 h in a humidified atmosphere of 95 % air and 5 % CO₂ at 37 °C.

After 24 h, culture medium was then aspirated, washed with 5 ml of PBS to remove dead cells and cellular debris. PBS was replaced with fresh DMEM medium containing yellow tetrazolium salt, cells were re-incubated for an additional 3 h at 37 °C. The supernatant was removed, and the residual purple formazan product was solubilized in 2 ml solubilization buffer, then plates were gently shaken and re-incubated for 15-30 min to dissolve the purple formazan dye crystals formed and to stop the reaction. One ml of treated and non-treated cells were transferred into a cuvette and the absorbance was measured by using a spectrophotometer at OD_{570nm} wavelength to measure absorbance of formazan product. The percentage of viable cells was calculated from triplicates using the following formula (1): as a described by (Rezk *et al.*, 2015).

Cell viability (%) =
$$\frac{\text{Absorbance of treated cells}}{\text{Absorbance of control cells}} \times 100$$

The percentage of cell viability was calculated by the absorbance readings from the test sample and divided by the control sample then multiplied by 100.

2.19 Cell freezing procedure

To make frozen stocks, Caco-2 cells were harvested using the standard procedure described in Section 2.18.1, after the supernatant was discarded, 90 % DMEM with 10 % Dimethyl sulfoxide (DMSO) were prepared by adding 9 ml of DMEM with 1 ml of DMSO and mixed. The mixture was filter sterilized using 0.22 µm filters, then 6 ml of the solution was added to Falcon tube that contain pellet of cells. The pellet was resuspended in the solution and 1 ml of cell suspension was transferred into a cryogenic vial, and then placed into a container containing 250 ml of isopropanol (99.5+ %, pure). The tubes were marked with name of cells, number of passages and date. Tubes were kept in -80 °C for one night before permanent storage in the gas phase of a liquid nitrogen tank (at -180 °C) for optimum long-term survival. Frozen cells were recovered by rapid thawing at 37 °C, then washed and maintained in complete medium by mixing directly with 10 to 15 ml of sterile complete medium. Each vial containing 1 ml of frozen cells thawed in a 75 mm tissue culture dish.

2.20 Evaluation of toxicity of liquorice extracts on the yeast

A single colony of yeast strains S288C and NCYC 363 was inoculated in yeast extract -peptone dextrose (YPD) broth medium and incubated at 30 °C for 48 h. Cells were harvested by centrifugation and re-suspended in 20 ml distilled H₂O. This process was repeated twice to ensure all the YPD broth was removed from the culture and then water was added to a final volume of 10 ml. Three dilutions were prepared 10^{-1} , 10^{-2} and 50/50. Two Microcentrifuge Eppendorf tubes were filled with 900 µl water and one was filled with 100 µl methylene blue, 100
µl of culture was added to the first tube mixed thoroughly then 100 µl was transferred into the second Microcentrifuge Eppendorf tube and mixed then 100 µl was transferred into third tube that contain of 100 µl of methylene blue then mixed thoroughly. 10 µl of 50/50 suspension was drawn from the pipette to fill the counting chamber in one side, then cells counted using a Neubaure Haemocytometer. The number of cells was calculated then diluted in YPD broth to a density of 1x10⁶ cfu ml⁻¹. Three 100 ml flasks for each strain were prepared containing a total volume of 20 ml of culture. The first flask (control) contained 1 % (v/v) ethanol, the second flask contained 12.5 µg ml⁻¹ of liquorice extract and the third flask contained 50 µg ml⁻¹ of liquorice extract. 200 µl of the cultures were transferred into a 96 well plate. The plate was then shaken in a microplate reader (TECAN) at 25 °C for 1800 cycles of orbital (30 min) shaking followed by OD measurements for 48 h.

2.21 STATISTICAL ANALYSIS

Data were converted to log (cfu ml⁻¹) for bacterial counts and parametric statistics (means \pm standard errors) were used. All the results were expressed as mean \pm SD. Analysis of variance (ANOVA) were used to analysis the data. Each data was compared to one another to show if there is any differences between the means. Statistical analysis was carried out using GraphPad Prism V7 (GraphPad Software, Inc, CA, USA) and Microsoft[®] Excel 2010/XLSTAT©-Pro (2013.4.03, Addinsoft, Inc., Brooklyn, NY, USA) statistical package at 95 % confidence level. Analysis of variance ANOVA vs Mann Whitney was used. Differences were evaluated by using Tukey's post hoc analysis for multiple comparisons within groups of normally distributed data and the values were considered as statistically significant when p values were less than 0.05 (p < 0.05).

CHAPTER 3

EVALUATION OF LIQUORICE EXTRACT AS AN ANTIBACTERIAL

Abstract

In this chapter the antibacterial activity of liquorice extract was determined by screening for growth inhibition using eight species of bacteria of food relevance. The Gram-positive bacteria included *L. monocytogenes, L. innocua, S. aureus, E. faecalis* and *B. subtilis*. The Gram-negative bacteria include *P. aeruginosa, E. coli* and *S.* Typhimurium but none of this latter group were affected by the extract. For all of the Gram-positive bacteria tested, growth was inhibited as monitored using optical density, but the cells were not killed since the cells were still viable when plated out. The Minimum inhibitory concentration [MIC] and Minimum bactericidal concentration [MBC] of the extract was also determined and a concentration of 50 µg ml⁻¹ was found to have a strong bacteriostatic effect on Gram-positive bacteria. In addition the growth of *M. smegmatis* was also inhibited, suggesting that a wide spectrum of Gram-positive cell types are sensitive to this extract.

In addition the use of a reporter strain of *Listeria* transformed with the bioluminescence genes *luxABCDE* indicated that cell energy levels were reduced when treated with either 12.5 or 50 µg ml⁻¹ of the extract, with the reduction in light output being proportional to the concentration of the extract used and also in this study the mechanism of action of the liquorice extract was further investigated using the LIVE/DEAD[®] viability stain. Exposure to the MIC for this extract (50 µg ml⁻¹) resulted in total inhibition of bacteria growth and samples imaged using the LIVE/DEAD stain showed that the cells were unable to actively remove the fluorescent dye from the cytoplasm, suggesting that membrane integrity was affected. Exposure to a sub-lethal level of the extract (12.5 µg ml⁻¹) resulted in some growth inhibition and a lower percentage of cells showing evidence of membrane damage.

Evidence that cell division is inhibited was also seen since cells exposed to 50 µg ml⁻¹ were generally shorter and did not take up Gram-stain effectively. Together these results indicate that the extract targets a site in the cell wall of

Gram-positive bacteria and it has potential as a natural food preservative either for high risk *Listeria* products or for products where spoilage occurs due to Grampositive bacteria.

3.1 Introduction

Bacteria such as *Campylobacter*, *Salmonella*, *L. monocytogenes* and *E. coli* are commonly reported to be responsible for foodborne disease and outbreaks (Alocilja and Radke, 2003; Chemburu *et al.*, 2005). Generally, all these bacterial pathogens are grouped together, but physiologically they are very different, in particular a major structural difference exists between the different cell types. For instance, *Campylobacter*, *Salmonella* and *E. coli* are all Gram-negative bacteria and have a cell wall structure including two layers of membrane and a thin layer of the structural polysaccharide peptidoglycan (PG). In contrast Gram-positive bacteria, such as *Listeria*, *Staphylococcus*, *Enterococcus* and *Bacillus*, have only one cell membrane encased in a thick layer of PG. Hence when considering the action of new antimicrobial agents, it is important to test them against a range of bacterial cell types.

Generally, when testing antimicrobial agents, cells are challenged with the test chemical and the effect measured by the level of growth inhibition observed. Despite the fact that the definition of viability depends on the capacity of bacteria to multiply and divide, growth of bacteria in a liquid medium, or the formation of colonies on solid agar, is not considered sufficient evidence that bacteria are either alive or dead after treatment since antimicrobial agents may simply be bacteriostatic and prevent growth without causing permanent cell damage. Thus, intermediate situations such as cell injury are hard to identify by plating methods. To address this, other indirect methods have been developed to report on cellular viability, including monitoring enzyme, respiratory activity and measurement of

ATP levels (Riss *et al.*, 2011). These parameters can also report on sub-lethal damage to cells, for instance factors that affect membrane integrity or require energy for repair will reduce ATP levels but not result in cell death (Nocker *et al.*, 2011; Hill and Stewart, 1994). Indeed many studies have shown the significant effect of microbial stress on cell growth due to the depletion of energy levels and it has been pointed out that the rate of cell growth is related to the stress applied (Khan *et al.*, 2010).

Bioluminescence is the process by which visible light is emitted by an organism as a result of a chemical reaction catalysed by a luciferase enzyme and this phenomenon is seen in a variety of organisms, such as insects, fish, squid, shrimp, and jellyfish, but is also seen in bacteria. Bacterial luciferase (Lux) is encoded by the *luxAB* genes which catalyse the reduction of a long-chain fatty aldehyde in the presence of molecular oxygen and flavin mononucleotide (FMNH₂) and results in the emission of light (Figure 3.1 B). Cells encoding the full *luxCDABE* operon (Figure 3.1 A) generate a bioluminescence phenotype without the need to add any additional cofactors or exogenous substrates as the accessory genes luxC, *luxD* and *luxE* produce the aldehyde within the bacterial cell (Hill and Stewart, 1994; Meighen, 1993). Since $FMNH_2$ production is dependent on a functional electron transport chain, only metabolically active bacteria emit light and therefore it can be used to rapidly monitory antimicrobial activity (Hill and Stewart, 1994). Although many bacteria of interest are not naturally bioluminescent, the lux operon can be introduced into a number of different cells on plasmids as long as the genes are engineered to function in that particular cell type (Perehinec et al., 2007) and therefore bioluminescence has been widely used to study antimicrobial agents (Hill and Stewart, 1994; Robinson et al., 2011).

Figure 3.1. Bacterial bioluminescence



Summary diagram showing the reaction catalyzed by bacterial luciferases: Panel A) Organisation of the *lux* operon found in *Vibrio* and *Photorhabdus* species, including the luciferase genes *luxA* and *luxB* and the *luxCDE* genes which encode a fatty acid reductase complex that regenerates the long chain aldehyde substrate (luciferin). Panel B) The chemical reaction giving rise to light production in bioluminescent bacteria in which the aldehyde is converted to the carboxylic acid in the presence of oxygen reduced flavin mononucleotide (FMNH₂); the by-products of this reaction are flavin mononucleotide (FMN) and light in the visible spectrum (bioluminescence) as blue/green light emission (^{λ}MAX ~ 490 nm) (Adapted from Gahan, 2012).

Another commonly used marker gene is the green fluorescent protein (Gfp) produced by the jellyfish *Aequorea victoria* which has become a simple and flexible tool used in many applications in the field of molecular biology, medicine and cell biology. The protein is small (27 kDa) and produces green fluorescence at 509 nm when excited with UV light at 398 nm (Lorang *et al.*, 2001; Zimmer, 2002). As Gfp protein is extremely stable *in vivo*, it has proved to be useful when studying cells exposed to some sort of stress condition, as the fluorescence signal does not decrease allowing cells that produce this protein to be easily visualised, even if the cells are non-viable (Lowder *et al.*, 2000). Again, bacteria do not naturally produce Gfp but this can also be introduced into bacteria either on plasmid vectors or integrated into the chromosome (Qazi *et al.*, 2001). It is also possible to introduce multiple marker genes onto a plasmid, so that Gfp can be introduced in conjunction

with the *lux* genes so that cells containing these plasmids are both bioluminescent and fluorescent and as the spectrum of light produced by each marker is different, there is no interference between the two signals (Qazi *et al.*, 2001). Overall, these reporter proteins allow the evaluation of bacterial viability in an inexpensive, rapid and easy way. Measurements are also easily automated, which makes luciferase and Gfp-based measurements two of the most efficient ways to investigate the viability of different microorganisms.

The main requirement in different fields of microbiology, including public health, biotechnology, food manufacture and the water and pharmaceutical industries, is the evaluation the viability of bacteria that commonly cause contamination such as Escherichia coli, Pseudomonas aeruginosa, Pseudomonas syringae, and Salmonella enterica serovar Typhimurium (Pyle et al., 1999; Nyström, 2001). Because bacterial cell death has long been defined as the failure of a cell to grow to a visible colony on bacteriological media, therefore bacterial death can only be monitored retrospectively using traditional culture methods. However intermediate cases exist, such as cell injury, which are difficult to measure using plating methods (Berney et al., 2007). Due to various environmental stresses, such as nutrient starvation, high or low temperature, high pressure or changes in pH, several bacteria have also been shown to enter into a viable but non-culturable (VBNC) state, where they are alive but lose the ability to recover into active growth until suitable conditions for growth are reestablished. However there is evidence that ingestion of such VBNC bacteria can lead to infection and therefore this state is a major concern when assessing the effectiveness of antimicrobials (Pienaar et al., 2016; Khan et al., 2010).

Despite these limitations, the most common method for enumerating *L. monocytogenes* present in food samples is culture, but it is time consuming and depends on cell growth, so that – as mentioned above - injured and non-culturable

cells have no opportunity to be detected (Olszewska *et al.*, 2015). In addition, for bacteria which have the ability to form clumps or chains, the viable count is not a good method to accurately measure cell numbers. This is also true for bacteria forming biofilms which is a form of bacterial growth that is widespread in factory environments and makes it difficult to release bacteria from surfaces for accurate determination of viable count by culture methods (Bettencourt *et al.*, 2010). Especially in high-throughput experiments, colony counting is a labour-intensive and time-consuming process due to the need to create serial dilutions and plating samples on agar plate, which makes it difficult to use for a large number of samples.

In contrast, traditional direct-count investigations of bacterial viability rely on metabolic characteristics or membrane integrity (Paparella et al., 2012). A large number of investigations and processes have been developed in order to investigate the physiological characterization of bacteria at the single-cell level to understand all the processes that occur inside the cell. Most of these methods require long periods of time to complete, in some cases up to 8 hours, such as fluorescence-based methods which involve direct viable count using a microscope or methods combined with nucleic acid staining as well as measurement of membrane integrity (Joux and Lebaron, 1997; Boulos et al., 1999; Yokomaku et al., 2000). Although bioluminescence measured directly reflects the number of viable cells, all living cells are not necessarily always in the same metabolic state which may limit the effectiveness of bioluminescence as an assay of absolute cell numbers. In contrast, fluorescence can be used to assess bacterial cell number. GFP is known as a stable protein and has the capacity to fluoresce in the absence of the need to deliver a substrate, and unlike bioluminescence, Gfp fluorescence density is reported to be proportional to total cell number. Thus, Lehtinen et al. (2006) reported that the Green fluorescent protein (Gfp) might be considered as a flexible tool to study the impact of antimicrobial agents on the growth of bacteria

as a rapid alternative to monitoring cell number using culture-based methods (Lehtinen *et al.*, 2006).

However the most commonly used alternative approach to traditional methods of assessing cell viability that has been developed is to use LIVE/DEAD staining kits which can be used for viability assessment for estimating total cell count, including both dead and live cells (Kell et al., 1998; Leuko et al., 2004). Hence, in different fields of microbiology, the LIVE/DEAD BacLight kit has seen increasing usage by researchers as a rapid method for detection and distinguishing between a live and dead bacteria. This is a fast method frequently used to determine both viable and total counts of bacteria and monitors bacterial cell viability as a function of the membrane integrity. The kits consists of a combination of two nucleic acid stains, the red fluorochrome (propidium iodide) (PI) is a large molecule that can pass through damaged plasma membranes into cells, while the small molecule green fluorochrome (SYTO 9) penetrates all bacterial membranes (Montañez-Izquierdo et al., 2012; Alonso et al., 2002). A bacterial viability kit which detects the presence of an intact membrane potential based on the differential accumulation of two dyes; component A (SYTO 9 dye, 3.34 mM) and component B (Propidium iodide, 20 mM). The excitation and emission maximum for these dyes are about 480/500 nm for SYTO 9 stain and 490/635 nm for propidium iodide (Probes, 2004). These kits are a rapid and convenient method that are appropriate for routine measurements. When the cell membrane is damaged, PI enters the cell and produces red fluorescence, which causes reduction in an intensity of SYTO 9 fluorescence and labels cells with damaged membranes. Thus, cells with an undamaged membrane will stain green while cells with a disrupted membrane that are considered to be non-viable will stain red and this is taken as evidence of the viable state of the bacteria, but - as discussed above in relation to the VBNC state - there is still some controversy on how to define dead

bacteria (Swaminathan and Gerner-Smidt, 2007; Montañez-Izquierdo *et al.*, 2012).

Kits containing propidium iodide and SYTO9 are the most commonly used, and have been used to assess bacterial viability in biofilms, distinguish between pathogenic and non-pathogenic bacteria and to perform a rapid viable count of water-borne bacteria (Yang *et al.*, 2009; Hoefel *et al.*, 2003). They have also been used to study different bacterial species in pure culture and in various environments and food products (Berney *et al.*, 2007; Leuko *et al.*, 2004). The technique can be applied rapidly because no extended incubation period is required to allow for colony growth, and quantitative results in some case can be gained within 1 h using a fluorescent microscope or flowcytometer, or can also be used as an alternative methods to monitor membrane integrity (Olszewska *et al.*, 2015). The technique is also versatile in that it has been successfully applied to a number of bacterial species, most of which are known pathogens (Berney *et al.*, 2007; Stocks, 2004).

Besides all of these indirect methods of measuring cell damage, one important parameter required to understand the mechanism of action of antimicrobial agents on microbes is visualising the morphological structure after treating with antimicrobial agents to determine how these agents could be impacting on microorganisms (Kitajima, 1998).

Objectives: Previous research in this laboratory and in the literature had indicated that liquorice showed a good antimicrobial activity against different types of bacteria as mentioned in Section 1.5.1. So far, liquorice has been extensively studied for its efficacy as an antimicrobial but the antibacterial mechanism of waste material recovered from the end of the production of liquorice has not been specifically reported. Hence the first aim of this Chapter was to investigate the

potential of one particular liquorice extract as natural antimicrobial compound, in particular against Gram-positive bacterial pathogens, including *Listeria monocytogenes* and *Staphylococcus aureus*. At the beginning of this work a product was supplied by a UK company (Phytoquest, UK) that was a waste product of the production of liquorice flavouring and had the potential to be used as a food additive to control these organisms. The first questions to be addressed were (a) what is the range of effectiveness? and (b) what is the minimum inhibitory concentrations required?. The mode of action would also be investigated by determining whether there are any visible effects on the membrane integrity and cell wall morphology.

The second aim was to use a range of techniques to examine the viability of bacteria after liquorice extract treatment. The Live/Dead BacLight[™] bacterial viability assay (Live/Dead staining) and viable count were employed to assess the extract treated with *Listeria* strains and *Bacillus*.

The main body of the methods are described in Chapter 2, when necessary some of these details are repeated to make it easier to follow the experiments presented.

3.2 Results

3.2.1 Test solubility of liquorice in ethanol solvent

Before starting the experiments it was important to confirm the percentage of solvent required to solubilise the compound. Phytoquest had provided information that liquorice powder should be dissolved in ethanol, but this is antimicrobial in itself and therefore it was important to determine the lowest concentration of solvent that could be used. Therefore different mixtures of ethanol/water were prepared in order to determine what composition was required to dissolve of liquorice powder completely. Compositions of ethanol/water (30:70, 40:60, 50:50, 60:40, 70:30 v/v) were mixed with different weights of liquorice powder (1.0, 2.0 and 3.0 mg) and the bottles were sealed with Nesco film to avoid evaporation of the solvent and then the samples were incubated at 37 °C for less than 30 min. The results indicated that ethanol/water (70:30, v/v) was the optimum concentration solvent as it was able to dissolve 3 mg of liquorice in 1 ml of solvent completely. Thus, in all further experiments in this thesis were carried out as using 70 % (v/v) ethanol to prepare the initial liquorice extracts.

3.2.2 Establishing the growth conditions required for different Grampositive and Gram-negative bacteria

The aim of this study was to investigate the effect of a different concentrations of liquorice extract on the growth rate of bacteria, these were initially tested by using a broth dilution method. Testing was conducted using organisms that contribute to infectious diseases and food spoilage including *E. faecalis*, *S. aureus*, *B. subtilis*, *L. innocua*, *L. monocytogenes*, *E. coli* and *P. aeruginosa* (see Table 2.2 for details). These organisms were chosen because of their harmful effects on human health and food spoilage including Gram-negative and Gram-positive bacteria, both spore-forming and non-spore forming.

First it was necessary to determine the appropriate method to prepare cultures of the test bacteria to allow convenient growth of different strains of bacteria being used as a part of this study and also to be able to monitor growth using optical density. Overnight cultures of the Gram-negative and Gram-positive bacteria including *E. faecalis*, *S. aureus*, *B. subtilis*, *L. innocua*, *L. monocytogenes*, *E. coli* and *P. aeruginosa* were prepared as described earlier in Section 2.8, then different dilutions were made (1:10, 1:50, 1:100) of these overnight cultures into different broth media (Tryptone Soya Broth, Nutrient Broth, Mueller Hinton Broth and Brain Heart Infusion Broth). Due to the fact that the different types of bacteria in this study had very different growth rates, it was first necessary to determine what dilution of the overnight culture would produce an optimal growth curve using the automated Tecan microtiter plate reader and monitoring optical density at

 OD_{600nm} . In particular it was important that the different stages of growth such as lag phase, exponential phase and stationary phase were visible. Samples (200 µl) of the different dilutions of the different cultures were transferred into a at least 4 wells of a sterile 96-well polystyrene microtiter plates which was then incubated at 37 °C for 24 h. Cell growth was monitored by using OD_{600nm} with readings taken every 10 min for 24 h.

The result showed that all the tested strains grew well in most of the different media and some representative results for two *Listeria* species, *S. aureus* and *E .coli* are shown in Figure 3.2. It was found that using BHI broth good growth was seen, but that all the strains had very different growth rates. Table 3.1 shows a typical result of the initial OD_{600nm} values achieved reached by the different test bacteria after overnight incubation. Hence some of these bacteria required extended incubation times to achieve a high enough cell density to allow the fresh cultures to be prepared to an initial OD_{600nm} of approximately 0.05 which reproducibly allowed the different phases of growth to be identified when samples were incubated in the Tecan plate reader and several test experiments were performed to establish the growth times required to achieve this.

Type of bacteria	Gram	Optical Density of	
	reaction	cultures	
Bacillus subtilis	+ve	0.381	
Enterococcus faecalis	+ve	0.286	
Listeria monocytogenes	+ve	0.101	
Listeria innocua	+ve	0.102	
Staphylococcus aureus	+ve	1.482	
Pseudomonas aeruginosa	-ve	0.585	
Escherichia coli	-ve	1.412	

Table 3.1. Growth of bacteria used in this experiment

Table showing typical results of the OD_{600nm} values for the different types of bacteria after overnight growth. These experiments were performed to establish the growth times required for each organism.

Figure 3.2. Growth curves for different types of Gram-negative and Grampositive bacteria



Growth curves of *L. monocytogenes*, *L. innocua*, *S. aureus* and *E. coli*. A colony of each bacteria was used to inoculate 5 ml BHI broth which was incubated at 37 °C overnight (18-24 h). The OD_{600nm} of the cultures was measured. The overnight growth was back-diluted into fresh BHI broth [●] 1:10, [●] 1:50, [▲] 1:100 and [▼] control contain BHI broth. Cultures were incubated at 37 °C in a shaking incubator (150 rpm), and optical density measurements were taken after 24 h to monitor the optical density change for all of bacteria tested.

3.2.3 Bacterial challenge experiments

The first point that needed to be established was whether the extract that had been provided was an active antimicrobial and what range of microbes it was effective against. The second point to be determined was the optimum concentration of the liquorice extract that would inhibit growth of the bacteria. Briefly, overnight cultures of different bacteria were grown in BHI broth, the cultures were adjusted in BHI broth to give a final density of 1X10⁸ cfu ml⁻¹ (approximately $OD_{600nm} = 0.1$), different concentrations of the liquorice extract (10, 20, 30, 40 and 50 μ g ml⁻¹) were prepared in BHI broth then 100 μ l were transferred into 96 well plate (Figure 3.3). After that 100 μ l samples of cultures at optical density 0.1 were transferred into the wells of a 96-well microtitre plate to give a final OD 0.05, two replicates for each concentration were used in addition to control, then the plate was incubated statically at 37 °C for 24 h in Tecan reader and optical density readings were taken throughout the incubation period. As control the bacteria were also grown in BHI broth without liquorice extract. The growth rate for all Gram-positive bacteria was calculated using the formula as was earlier described in Section 2.9.

Figure 3.3. Layout of samples in microtitre plate



Overnight cultures were prepared with different concentrations of liquorice extract e.g. 10, 20, 30, 40 and 50 μ g ml⁻¹, then incubated at 37 °C for 24 h. Using a microtitre plate to allow more concentrations to be tested, also allowed replicate samples of different bacteria tested.

The results shown in Figure 3.4 indicated that the liquorice extract was highly effective at inhibiting the growth of all the Gram-positive bacteria tested. At a concentration of 40 and 50 µg ml⁻¹, growth of *L. innocua*, *L. monocytogenes*, *S. aureus, E. faecalis* and *B. subtilis* were entirely inhibited. At 30 µg ml⁻¹ delayed growth was seen for all strains except *B. subtilis*, while a concentration 20 µg ml⁻¹ the growth was low rate for almost strains compared to growth seen in the presence of 10 µg ml⁻¹ liquorice extract (Figure 3.4). In contrast growth of none of the tested Gram-negative bacteria was inhibited by the liquorice extract at any of these concentrations (Figure 3.5) and, accordingly there was no significant difference (p > 0.05) between the growth rate of these organisms at all concentrations of liquorice extract tested. The specific growth rate (µ) was determined from the slope of the curve and the data expressed per hour. As expected, bacteria grown in BHI alone had the highest growth rate, while bacteria

grown in different concentrations of liquorice extract had variable growth rates, depending on the concentration of the extract used (Figure 3.4). However the growth rates achieved in the presence of 10 μ g ml⁻¹ of liquorice extract was almost the same in the BHI broth without extract particularly for *Staph aureus*, *E. faecalis* and *B. subtilis* (Table 3.2 and Figure 3.4) and in this case the effect of liquorice extract can be seen as an increase of the lag phase accompanied by a decrease on the specific growth rate (μ) that may result in slower but continual growth, as in for concentrations of 20 and 30 μ g ml⁻¹ or total growth inhibition as in a concentration of 40, 50 μ g ml⁻¹ (Figure 3.4). Thus, from the results obtained from this experiment, two concentrations of liquorice extract (12.5 and 50 μ g ml⁻¹) were selected to carry out further experiments as representative of concentration that would either give total growth inhibition (50 μ g ml⁻¹) or that would be predicted to cause sub-lethal injury (12.5 μ g ml⁻¹) to Gram-positive bacteria.

Figure 3.4. The effect of different concentrations of liquorice extract on growth of Gram-positive bacteria



Cultures of *L. innocua, L. monocytogenes, S. aureus, E. faecalis* and *B. subtilis* were grown in BHI broth at 37 °C for 24 h in the presence of different concentrations of liquorice extract (10 [\blacksquare], 20 [\blacktriangle], 30 [\blacktriangledown], 40 [\diamond] and 50 [\bigcirc] µg ml⁻¹). A control sample [\bigcirc] containing broth only and control culture [\square] was also included. Growth was monitored by optical density measurement at 600nm. Data are given as means ± standard deviations from two independent experiments, each performed in triplicates (n=3).

Bacteria Strains	Growth rate			
	Control	10 µg ml ⁻¹	20 µg ml⁻¹	30 µg ml⁻¹
L. innocua	0.34 h⁻¹	0.22 h ⁻¹	0.16 h ⁻¹	0.08 h ⁻¹
L. monocytogenes	0.50 h⁻¹	0.25 h⁻¹	0.24 h⁻¹	0.24 h ⁻¹
Staph. aureus	0.50 h⁻¹	0.44 h⁻¹	0.38 h⁻¹	0.40 h ⁻¹
E. faecalis	0.60 h⁻¹	0.59 h⁻¹	0.52 h ⁻¹	0.38 h ⁻¹
B. subtilis	0.48 h⁻¹	0.46 h ⁻¹	0.43 h ⁻¹	

Table 3.2. The growth rate of Gram-positive bacteria

Table shown that the calculated of growth rate of Gram- positive bacteria which grown for overnight in BHI broth then treated with different concentration of liquorice extract 10, 20, 30, 40, 50 μ g ml⁻¹, then incubated at 37 °C for 24 h.

Figure 3.5. The effect of liquorice extract on the growth of Gram-negative bacteria



Cultures of *E. coli* and *P. aeruginosa* were grown at 37 °C for 24 h in the presence of different concentrations of liquorice extract (10 [\blacksquare], 20 [\blacktriangle], 30 [\blacktriangledown], 40 [\bullet] and 50 [\bullet] µg ml⁻¹). A control sample [\bullet] containing broth only was also included. Growth was monitored by optical density measurement at OD_{600nm}. Data are given as means ± standard deviations from two independent experiments, each performed in triplicates (n=3). There is no effect of liquorice extract on the growth of Gram-negative bacteria.

3.2.4 Demonstrating the spectrum of activity of liquorice extract at active concentrations

To confirm the choice from the previous results of the two concentrations of liquorice extract (12.5 and 50 μ g ml⁻¹) were correct, these were used to carry out further experiments. In addition to these two test conditions, control experiments were included in this experiment to confirm the inhibition of growth of bacteria was due to liquorice extract rather than solvent. Therefore a control was prepared containing 1 % (v/v) ethanol which was calculated to be the highest concentration of ethanol present in any of the samples (i.e. 50 μ g ml⁻¹ sample). Biological evaluation of the extract was carried out by screening for antimicrobial activity against a range of bacteria L. innocua, S. aureus, B. subtilis, E. coli and S. Typhimurium to confirm that the extract had antimicrobial activity. Briefly, overnight broth cultures of Gram-positive and Gram-negative bacteria were adjusted to a population of 10⁷ cfu ml⁻¹ in fresh brain heart infusion (BHI) broth media then two concentrations of the liquorice extract (12.5 and 50 μ g ml⁻¹) were inoculated with cultured bacteria to an OD_{600nm} of 0.05, 200 µl of samples was transferred into microplate and incubated at optimum growth temperature for bacterial strain and growth was monitored over time using optical density at OD_{600nm} every 30 min over 24 hours, using a microplate reader. The MIC was confirmed as the lowest concentration which no growth was revealed.

It was again found that the extract only affected the growth of Grampositive bacteria since the Gram-negative bacteria *E. coli* and *P. aeruginosa* tested showed normal levels of growth (Figure 3.6 a and b) for any of the test conditions. The extract was found to specifically inhibit only Gram-positive bacteria (Figure 3.6 c, e& f), growth inhibition was seen the growth was completely inhibited at a concentration of 50 µg ml⁻¹, whereas at 12.5 µg ml⁻¹ growth rate was reduced, but some growth was still detected. The types of bacteria tested included Grampositive rods (*L. monocytogenes* (non-spore former) and *B. subtilis* (spore-former) and Gram-positive cocci (*S. aureus* and *E. faecalis*) showing that the extract inhibited the growth of a wide range of different cell types. The fact that all the different cell types were affected in the same way indicates a shared mode of action rather than a specific effect on one type of cell. However, the important points to note here are that optical density does not report on the viability of the cells detected. Experiments now needed to determine whether at a concentration of 50 μ g ml⁻¹ cell growth is just inhibited or whether the cells are killed.

Figure 3.6. The effect of liquorice extract on the growth of Gram-negative bacteria (a) & (b) and Gram-positive bacteria (c)-(e)-(f)



Liquorice extract with a two concentration of 12.5 [\blacksquare] & 50 [\blacktriangle] µg ml⁻¹ was strong effect on Gram-positive bacteria. For *L. innocua* a concentration of 50 µg ml⁻¹ inhibited growth, for both *S. aureus* and *B. subtilis*, 12.5 [\blacksquare] µg ml⁻¹ showed a little inhibition, and 50 [\blacktriangle] µg ml⁻¹ inhibited growth, at these concentrations there is no effect on the growth of either *E. coli* or *S.* Typhimurium and the results were the same as before, all bacteria at 1 % [\bullet] ethanol solvent was grown normally, this confirmed that the effect due to the extract.

3.2.5 Effect of liquorice extract on bacterial viability

Since bacterial growth inhibition does not necessarily indicate bacterial death, the previous experiment in Section 3.2.4 cannot distinguish between bactericidal and bacteriostatic effects. Thus, the effects on bacterial growth in liquid cultures treated with liquorice extract was studied through both optical density measurements and viable counts in order to determine if the effect of the liquorice extract on the Gram-positive organisms was bactericidal or bacteriostatic. L. innocua was chosen as a model organism, being non-pathogenic and non-spore forming. The MIC is interpreted as the lowest concentration that inhibits visible microbial growth and is expressed in terms of μg ml⁻¹, whereas the MBC is interpreted as the lowest concentration that can completely kill the bacteria. To do this the same method was used as previously described but this time both the viable count and optical density were monitored. Briefly, three flasks were prepared as before in Section 2.11 with final concentrations of 12.5 μ g ml⁻¹ and 50 µg ml⁻¹ of liquorice extract and also a control sample only containing the ethanol solvent. These flasks were inoculated with L. innocua at an OD_{600nm} of 0.05 (approx. 1×10^7 cfu ml⁻¹) and were incubated at 37 °C for 30 min in shaking incubator (150 rpm), and samples were taken every 30 min for OD_{600nm} and viable count determination for approximately 3 h. For the viable count, 10-fold dilutions in maximal recovery diluent (MRD) were prepared and plated on the BHI agar plates using Miles and Misra technique as a described in Section 2.12 and the results were recorded after the plates were incubated overnight at 37 $^\circ C$ and the results are shown in Figure 3.7.

The first most obvious result was that the bacteria grew on the agar plates to produce colonies, and therefore although growth was totally inhibited in the sample containing 50 μ g ml⁻¹ (Figure 3.7 a) there was no effect of the liquorice extract on the *L. innocua* viability (Figure 3.7 b), and the number of viable cells did not decline over time indicating that the effect of the extract was bacteriostatic

rather than bactericidal. Comparing the effects of the difference concentrations used, the results for viable count were the same as those for the optical density with some inhibition occurring when 12.5 μ g ml⁻¹ of the liquorice extract was added. This was particularly characterised by an increased lag time when viable count was monitored. Again strong inhibition of growth with 50 μ g ml⁻¹ was seen and confirmed that the MBC of the liquorice extract was at approximately 50 μ g ml⁻¹, and this level would taken as the MBC level in future experiments.

Using a Mann-Whitney test it was shown that the inhibition of growth did not decrease significantly (P > 0.05) in samples treated with 12.5 μ g ml⁻¹, which means that a concentration of 12.5 μ g ml⁻¹ is not enough to inhibit *L. innocua* growth. In contrast, as seen in Figure 3.7, a concentration of 50 μ g ml⁻¹ of liquorice extract inhibited the growth of *L. innocua* significantly an effect (p < 0.0001). The significant difference in the growth rates determined using the viable count method was confirmed using ANOVA Tukey's test and the results were the same as those gained using optical density measurements, with significantly slower growth occurring when 12.5 μ g ml⁻¹ of the liquorice extract was added to the cultures.

Figure 3.7. The effect of liquorice extract on the growth (a) and the viable count (b) of *Listeria innocua*



Graphs comparing growth and viable count for *L. innocua* exposed to liquorice extract. Panel (a) shows the optical density and panel (b) shows the results gained using viable count after incubation of *L. innocua* with two concentrations of liquorice extract 12.5 [•] and 50 [•] μ g ml⁻¹ for various time intervals. The activity is expressed as log₁₀ cfu ml⁻¹ observed at each 30 min compared to the control sample that contained 1 % (v/v) [•] ethanol solvent. Data from two independent experiments, each performed in triplicates (n=3). Results are expressed as mean \pm SD and the comparison is made by Tukey's post hoc test to calculate the P-values. Significant difference between a concentration 50 μ g ml⁻¹ and 1 % (v/v) ethanol solvent is indicated as p < 0.0001.

3.2.6 Assessment of cell morphology using Gram-staining

As many of the bacteriostatic agents affect cell wall synthesis and cell division, experiments were carried out to monitor the cell morphology after the application of the liquorice. In this case two different Gram-positive bacteria were chosen. *L. innocua* as a representative of the *Listeria* genus which are non-spore forming bacteria and *B. subtilis* which does form spores in case there were differences seen in the effect on these two difference groups. In this case Gramstain and microscopy were used to detect the effects of the extract on bacterial cell wall integrity. Cultures were prepared as described in Section 2.11 and then three 100 ml flasks for each type of bacteria were prepared containing a total volume of 20 ml of culture. The first flask (control) was supplemented with 1 % (v/v) ethanol to rule out solvent effects, the second flask was supplements with 12.5 μ g ml⁻¹ of liquorice extract. The flasks were incubated at 37 °C for 24 h in shaking incubator (150 rpm), and samples were taken at the beginning and the end of experiment.

Samples of both *L. innocua* and *B. subtilis* were then transferred onto a microscope slide and distributed by a loop around the centre before the Gram stain procedure was carried out (Section 2.13). Samples were viewed using a light microscope to see if any change in cell morphology could be detected and the results are presented in Figure 3.8.

There was a slight change in cell morphology detected in the treated samples in that the cell shape was shorter and rounder than those seen in the control samples, typical of agents that result in inhibition of cell division. However it was also found that samples treated with the liquorice extract were less able to take up the Crystal Violet stain and therefore appeared paler in colour (Figure 3.8). Measurements were carried on these images to quantify the change in average cell length and the result as shown in Table 3.3 and this confirmed that

for *Listeria* after 48 h exposure to the 12.5 μ g ml⁻¹ concentration, the cell length was the same (0.52 μ m) as that for the samples treated with 50 μ g ml⁻¹ (0.51 μ m) whereas after 24 h the cell length at a concentration 12.5 μ g ml⁻¹ was less markedly affected (0.8 μ m). The fact that the same pattern was seen for both types of bacteria, and the fact that the growth of all the Gram-positive bacteria tested to date was inhibited, suggest that the primary target of the liquorice extract is something common to Gram-positive bacteria and that it may inhibit cell wall synthesis leading to disruption of the normal synthesis of the peptidoglycan layer.

Figure 3.8. Effect of liquorice extract on cell morphology of Gram-positive bacteria

L. innocua





Control 1 % (v/v) ethanol 12.5 μ g ml⁻¹ of extract



50 µg ml⁻¹ of extract

B. subtilis



Control 1 % (v/v) ethanol 12.5 μ g ml⁻¹ of extract 50 μ g ml⁻¹ of extract

Gram-stained samples of *L. innocua* and *B. subtilis* were taken after 24 h incubation (x 1000) with two concentration of liquorice extract 12.5 & 50 μ g ml⁻¹ in addition to control sample contain 1 % (v/v) ethanol solvent. As images were captured using a light microscope (Nikon) connected to a Canon camera (Canon EOS 700D, Japan) after Gram-staining using a x100 oil immersion lens.

Table 3.3. Observed and calculated mean cell dimensions (μm) of *L. innocua* and *B. subtilis*

Time (h)	Treatment with	L. innocua	B. subtilis
	liquorice extract	Av. Length	Av. Length
		(µm) n=10	(µm) n=10
24 h	Control	1 µm	0.99 µm
	12.5 µg ml ⁻¹	0.8 µm	0.61 µm
	50 µg ml⁻¹	0.52 µm	0.59 µm
48 h	Control	0.9 µm	
	12.5 μg ml ⁻¹	0.52 µm	
	50 µg ml⁻¹	0.51 µm	

Overnight cultures of *L. innocua* and *B. subtilis* treated with a two concentrations of liquorice extract (12.5& 50 μ g ml⁻¹) as well as 1 % of ethanol solvent. A Gram stain procedure was carried out on these cells and examine under microscope and images were taken and measured. Images from the light microscope were all produced at the same magnification from the original images to allow differences in cell length to be determined hence cell sizes are given in μ m. Target cells were identified by dividing the field into 4 equal sections. The average cell dimensions for 10 individual bacteria per field were measured. Cells showing irregular deformation were excluded from the measurement.

3.2.7 Confirmation of the fluorescent phenotype of *Listeria* strains carrying synthetic *gfp:luxABCDE* operon

Fluorescent microscopy is a very useful for studying the effects of different antimicrobials at the cellular level. Green fluorescent protein (GFP) provides an easily detectable phenotype, so it has been used to label many different microorganisms. Many of the strains used in this Chapter which had been transformed to contained the gfp gene from the jelly fish Aquoria victoria (Chattoraj et al., 1996) and therefore the expression of this gene in these bacteria was tested to see if it was sufficient to allow direct visualisation of cells without the need for fluorescent dyes. In addition these strains also carried the full bacterial lux operon to allow bioluminescence to be produced. As a range of different strains of *Listeria* that had been transformed to a *lux* phenotype were available, it was also decided to investigate whether any differences would be seen if different species of the same bacteria were used. Hence the effect of strain variation was investigated by using different strains of *L. monocytogenes* in addition to L. innocua. Three wild-type strains of L. monocytogenes were used, including serovars 1/2a (L. monocytogenes EGD, L. monocytogenes 10403S) and 4b (L. monocytogenes 23074) which represent the major causes of human listeriosis. In addition, two mutant strains which either over express the virulence genes (L. monocytogenes NCTC 7973; prfA* mutant) or cannot express the virulence genes (*L. monocytogenes* NCTC 10357; $\Delta prfA$) were used (see Table 2.2 & 2.3 for details).

The cultures were recovered from storage at -80 °C and purified to single colonies on BHI agar. Once pure cultures were obtained the next goal was to confirm that the emission of light could be detected and the level of Gfp protein produced by these strains was sufficient to allow detection of single cells. A complete list of strains and the details of the plasmids they contained are listed in Tables 2.2, 2.3& 2.4.

3.2.7.1 Confirmation of light emission (Luminometry)

First of all to confirm that the *Listeria* strains were emitting light, fresh cultures were prepared (Section 2.4.1) and 3 ml samples of overnight cultures were placed in a Luminometer (Model TD-20e Luminometer, Turner designs) the reading was recorded in (Table 3.4). L. innocua without a plasmid was used as a negative control. However, for these stationary phase cultures the light levels were very low, thus to obtain an exponentially growing culture, 1 ml of the overnight culture was transferred into a sterile flask containing 49 ml of BHI broth (1:50) and incubated again for 3-4 h at 37 °C in shaking incubator (150 rpm), samples of cultures (3 ml) were again transferred into the luminometer and the light levels measured. This time the results showed there was a 10-100 fold increase in the light levels detected which is consistent with the fact that the luciferase enzyme requires the cell to be producing reduced FMNH₂ through respiration or fermentation and levels of metabolism are much higher in exponentially growing cells than in cells in stationary phase. However a consistent pattern of results was seen in that the same strains still had the highest rate of light production both before and after dilution which were LM 10403S and LM 23074 (Table 3.4). It was interesting to note that L. innocua showed the greatest increase in light (approximately 900-fold increase) despite the fact it contained the same plasmid as the majority of the other isolates. In these plasmids (Bs10:Dual:pUNK1) the reporter genes are linked to the ribosomal S10 subunit promoter from *B. subtilis* which is constitutively expressed in Listeria (Dr P Hill, University of Nottingham, pers comm.) so it would be expected that light levels would simply reflect levels of metabolism. The strains chosen also included a *prfA** mutant LM 7973 which over expresses the virulence genes (Eiting et al., 2005) and a deletion mutant of prfA Δ LM 10357 that does not express the virulence genes at all (de las Heras et al., 2011). LM 10357 contained a slightly different plasmid that used the XylA promoter which is also constitutively expressed in *Listeria* (Perehinec et al., 2007), but the level of induction seen was very similar in these two mutant strains

suggesting that the expression of the virulence genes does not influence the level of light produced. Interestingly in the wild type strains LM 23074 and LM 10403S both produced the highest level of light in stationary phase and the lowest change in light levels in exponential phase (less than 10-fold induction), whereas EGD and *L. innocua* produced the highest changes in light level. This result again suggests that light levels are not affected by the expression of the virulence genes since *L. innocua* is a not a pathogen and does not have these genes.

	Light Emitted (RLU)			
Strains and	Characteristic	Stationary	Exponential	Fold
plasmids		phase	phase cells	increase
		cells		
L. innocua (control)	wt	0.088	0.088	0
L. innocua	wt	2.038	1828	897
(Bs10:Dual:pUNK1)				
LM EGD	wt	1.475	693	470
(Bs10:Dual:pUNK1)	serotype 1/2a			
LM 10403S	wt	318.4	1910	6.0
(Bs10:Dual:pUNK1)	serotype 1/2a			
LM 23074	wt	388.2	1472	3.8
(Bs10:Dual:pUNK1)	serotype 4b			
LM 7973	PrfA* mutant	8.355	609.5	73
(Bs10:Dual:pUNK1)	serotype 4b			
LM 10357	PrfA∆ mutant	1.578	103.6	66
(XyIA:Dual:pUNK1)	serotype 4b			

Table 3.4. Effect of growth phase on light emission by Listeria

*wt: wild type

Listeria strains were grown at 37 °C in BHI broth that was supplemented with 5 μ g ml⁻¹ of erythromycin for strains that contained plasmids. After 24 h incubation the light emitted was measured and then the cultures were diluted and incubated at 37 °C for a further for 3 h before the light emitted by these strains was measured again. Light is recorded as Relative Light Units (RLU) per sample. The fold increase was calculated by dividing the light levels at time point 2 by the light levels at time point 1.

3.2.7.2 Confirmation of fluorescence phenotype (GFP)

Listeria strains carrying the plasmids listed in Table 3.4 were then examined to confirm the expression of the Gfp protein from the dual reporter operon (*qfp:luxABCDE*; Perehinec *et al* 2007). To do this individual colonies from a plate were inoculated into samples (200 μ l) of BHI broth supplemented with 5 μ g ml⁻¹ erythromycin in a 96 well plate, using a sterile toothpick. The plate was incubated at 37 °C for 24 h then fluorescence readings taken using a Tecan plate reader. The results (Table 3.5) showed that all the strains tested had the ability to express Green Fluorescent Protein (GFP) but a range of different levels of fluorescence was detected depending on both the strain and the individual colony used to inoculate the wells. This correlated with the fact that when colonies appeared to fluoresce more than others when imaged under the Biospace lab photon imager in Section 2.14.2. In this case the strain that produced the highest average levels of fluorescence was LM 23074, but given the large standard deviation of the values recorded in each group this is not significantly higher than any of the other strains. In these plasmids the reporter genes are both expressed from the same promoter, so it might be expected that the pattern of expression would be the same for both markers. However, unlike bioluminescence, fluorescence is not linked to the metabolic state of the cell and therefore these results indicated that in all strains the promoter activity is approximately the same, but physiological differences in the cells affect the amount of light produced. The variation in the results for each set of strains also indicates that the level of reporter produced in each individual cell is guite variable.

Samples were transferred into individual Eppendorf tubes and the cells were collected by centrifugation (13,000 rpm for 3 min) and resuspended in 200 μ I MRD by vortexing and then transferred into new microtiter plate to measure the fluorescence levels which are shown in (Table 3.6).
	L. inr	посиа	LM	EGD	LM 10	0403S	LM 2	3074	LM 2	7973	LM 1	0357
Α	32246	25281	29561	24670	21543	26593	33569	25595	29682	25689	26378	23165
В	31572	24968	17700	22151	26444	26787	42890	27107	20754	21961	23850	20219
С	33827	31478	35698	33372	30210	27895	36051	26815	35504	26877	23332	20937
D	34293	21410	25125	19758	24579	20422	34519	24867	27420	20334	18857	18156
E	34814	23239	34198	26908	34042	28358	34592	25678	33624	21984	20125	17571
F	26204	22220	27070	20536	20539	20048	32258	21363	26956	19779	20026	16441
G	33995	27426	27839	25737	26362	27657	37886	32205	31010	24038	18902	20706
Н	36641	25536	34876	21767	35376	24360	37840	24530	38078	25732	32970	24466
A	AV RFU	21631		26685		26326		31110		26839		29072
	SD	±3938		±5466		±4350		±5907		±5363		±4879

Table 3.5. Measurements of green fluorescent protein (Gfp) for all Listeria strains

Listeria strains carrying plasmids encoding the *gfp* (Table 3.4) were grown in BHI broth supplemented with 5 µg ml⁻¹ Erythromycin then incubated at 37 °C for 24 h. The plate was then moved to a Tecan plate reader to measure the fluorescence using an excitation wavelength of 485 nm and an emission wavelength of 535 nm. The gain was set at Optimal (10) and the number of reads taken for each sample was 10 (data in each cell represents the average of 10 readings). Results are reported as Relative Fluorescence Units (RFU) and the average and SD values for each group are shown at the bottom of the table.

Strain	L. innocua	LM EGD	LM 10403S	LM 23074	LM 7973	LM 10357
Α	7889	14551	17798	4836	8522	6538
В	6484	5728	22176	7486	6111	11522
С	5446	11824	34596	6894	5181	5230
D	6625	6207	24110	5900	5443	7198
E	12733	6299	40321	5490	5625	6432
F	6955	10527	42623	5162	7524	5198
G	6581	12527	38365	6294	5250	9718
н	3904	4024	4120	3762	4147	4112
AV RFU	7077	8961	28014	5728	5975	6994
SD	±2403	±3607	±12450	±1109	±1314	±2322

Table 3.6. GFP for *Listeria* strains

Cells were grown in BHI broth supplement with 5 µg ml⁻¹ of erythromycin and incubated at 37 °C (overnight from the cultures in Table 3.5), the cultures were transferred into Eppendorf tubes and centrifuged and resuspended in MRD and placed in a Tecan microtitre plate. The plate was then moved to a Tecan plate reader to measure the fluorescence using an excitation wavelength of 485 nm and an emission wavelength of 535 nm. The gain was set at Optimal (10) and the number of reads taken for each sample was 10 (data in each cell represents the average of 10 readings). Results are reported as Relative Fluorescence Units (RFU) and the average and SD values for each group are shown at the bottom of the table.

The Listeria strains (L. innocua, LM EGD, LM 10403S, LM 23074, LM 7973 and LM 10357) that exhibited highest fluorescence from the results shown in Table 3.6 were subcultured on a BHI agar plate supplemented with 5 μ g ml⁻¹ erythromycin to generate fresh colonies. To determine if the level of fluorescence produced was sufficient to detect individual cells, microscope slides were then prepared by picking up a colony and emulsifying the cells in a drop of PBS (approximately 40 µl) and then covered with cover slip and samples viewed using a fluorescence microscope. The results indicated that all the strains tested produced detectable levels of fluorescence and appeared as individual green cells (Figure 3.9). This experiment confirmed that all strains were producing the Gfp protein before starting treatment with liquorice extract. Thus, the next step was to determine if GFP was still being produce by the cells after treatment with the liquorice extract which was known to be exerting stress on the cells resulting in growth inhibition. One key point was to also demonstrate that when differences in cell shape were detected after prolonged treatment that this was due to changes in cell morphology rather than contamination of the culture with different bacteria that outgrew the test strains. This experiment will be described later in Section 3.2.11 and 3.2.12.

Figure 3.9. Detection of Green fluorescence in individual cells of Listeria



Cells were grown in BHI broth contain 5 µg ml⁻¹ Ery for 24 h at 37 °C, and were then harvested and resuspended in MRD. The most fluorescent strains were subcultured on agar plates at 37 °C for 24 h, and samples of the fresh colonies used to confirm that the individual cells were visible using fluorescence microscopy. These images representative results for two *Listeria* strains (a) LM 10403S and (b) LM 10357. Samples were imaged using florescence microscope (Nikon).

3.2.8 Use of bioluminescence to investigate the effect of liquorice extract on *Listeria* strains

To further investigate of the antimicrobial action of the liquorice extract, the *Listeria* strains containing the bacterial *lux* genes were then treated with the extract. However, these genes are encoded on plasmids and therefore to ensure that all the cells in a population retain the plasmid, antibiotic selection is needed and therefore additional controls will be needed to ensure that loss of light production is not due to plasmid loss. The same set of strains of *Listeria* described in the previous Section were again used (see Table 2.2 & 2.3 for details).

To carry out these experiments, the different strains of *Listeria* were recovered from frozen stocks by growth on BHI agar plates and then a colony was transferred into 5 ml of BHI broth supplemented with Erm (5 μ g ml⁻¹) for plasmid selection. These were grown for 24 h at 37 °C in a shaking incubator (150 rpm), before being diluted in fresh BHI broth supplemented with Erm (5 μ g ml⁻¹) to an

 OD_{600nm} of approximately 0.05. Samples were then supplemented with two concentrations of the liquorice extract (12.5 µg ml⁻¹ or 50 µg ml⁻¹) and 1 % (v/v) ethanol was used as a control. Due to the possibility of contamination when carrying out experiments using a microplate reader, an uninoculated control sample was also included (purple line, Figure 3.10) to show that any change in optical density or light recorded was due to the organism inoculated into the broth.

The results from the microtitre plate experiment showed the same pattern as before, in that a concentration of 50 μ g ml⁻¹ totally inhibited growth whereas at 12.5 µg ml⁻¹ very little inhibition of growth was seen in Figure 3.10 confirming that this concentration of the extract is above the MBC. The pattern of results was exactly the same for all strains and species of *Listeria* tested, showing that this effect was the same for both species and all strains of Listeria tested. However, when the levels of light produced from each of the strains were examined (Figure (3.11), it was clear that even when growth was not being affected by the addition of the extract at 12.5 μ g ml⁻¹ (see Figure 3.10), a reduction in the light output could be seen, although this was more pronounced in some strains than others (see Figure 3.11). For comparison, the results are presented on a log scale in the same way as the optical density measurements as they taken during the growth of the cultures when cell mass was increasing. This result indicated that the cells were experiencing some stress at the sub-lethal injury level as the level of light is representative of the overall metabolic state of the cells. It was also noted that the metabolic level of the strains challenged with 50 μ g ml⁻¹ was very similar to the background control, suggesting that in addition to effects on cell division, the cell membrane may be affected.





Growth curves of *Listeria* strains carrying *lux* gene were grown at 37 C for 24 h with two concentrations of liquorice extract [\blacksquare] 12.5, [\blacktriangle] 50 µg ml⁻¹ and 1 % (v/v) [\bullet] ethanol solvent as well as a control sample [\blacktriangledown] containing broth. Data are given as means ± standard deviations from four independent experiments, each performed in 4 replicate (n=4).

Figure 3.11. Effect of liquorice extract on (luminescence) light emitted by (a) LM 10357, (b) LM EGD, (c) LM 10403S, (d) LM 7973 and (e) LM 23074 containing plasmid pSB3008 and (f) *L. innocua*(pSB3007)



Graphs showing light output from the bacterial cells grown in present and absence of liquorice extract 12.5 [\blacksquare], 50 [\blacktriangle] µg ml⁻¹, [\bullet] 1 % (v/v) ethanol solvent and [\checkmark] uninoculated broth to confirm sterility. The bioluminescence intensity of the cells shown is the average for the replicate of light output by *Listeria* strains (n= 4 replicates).

Figure 3.12. Effect of liquorice extract on (a) LM 10357, (b) LM EGD, (c) LM 10403S, (d) LM 7973 and (e) LM 23074 containing plasmid pSB3008 and (f) *L. innocua*(pSB3007) reported as RLU per cell mass unit



Metabolic levels were monitored by measuring bioluminescence levels of strains carrying the bacterial *luxABCDE* genes. Light output (RLU) was divided by optical density to normalize light levels produced from samples containing different numbers of cells. Incubation of *Listeria strains* with two concentrations of liquorice extract 12.5 [\blacksquare], 50 [\blacktriangle] µg ml⁻¹ and 1 % [\bigcirc] (v/v) of ethanol solvent, a control sample [\checkmark] containing broth only was also included.

Since there were now both variations in growth rate and light output recorded it was important to normalize the bioluminescence data according to the number of cells in the sample (i.e. 10 cells producing 1 unit of light would give the same signal as 1 cell producing 10 units of light). To address this, the bioluminescence data was divided by the optical density data, and the results are presented in Figure 3.12 and as this data now represents the light per cell mass it is plotted on a linear scale. This treatment of the data that adjusts for differences in cell number in each sample more clearly shows that the cells treated with 12.5 µg ml⁻¹ of the extract were experiencing sub-lethal injury resulting in overall lower levels of metabolic activity. Since cell division was not affected in these cells as the growth rate was close to that of the control, this suggests that at sub-lethal levels the liquorice extract may be affecting either cell membrane integrity or other enzymes required for the production of FMNH₂ within the cells.

3.2.9 Confirmation of the presence of plasmid after treatment with and without liquorice extract

One question that remained was whether the presence of the antibiotic required for the selection of the plasmids was providing an additional stress on the cells that accounted for the results seen in Section 3.2.8. To determine whether the presence of the antibiotic was required to maintain the *lux* plasmids inside the cells, the stability of the plasmid was determined by growing *L. monocytogenes* strain 10403S(pSB3008) in the presence and absence of the selective antibiotic (Erm 5 µg ml⁻¹). To do this an overnight culture was prepared as a described in Section 2.4.1, and then the inoculum of a fresh broth culture was prepared, with and without 5 µg ml⁻¹ erythromycin then supplemented with 1 % (v/v) ethanol (control), 12.5 µg ml⁻¹ liquorice extract or 50 µg ml⁻¹ of liquorice extract. Samples of each culture (200 µl) were transferred into a microtitre plate and incubated at

37 °C for 24 h, and rates of plasmid-loss were investigated using viable count as a described in Section 2.12.

The stability of the plasmid was determined by comparing the viable count of samples taken from the experiment and plating on agar with and without the antibiotic Erythromycin (Figure 3.13) and the results are expressed as the percentage of cells retaining the plasmid (viable count on BHI supplemented with Erm) with total cell number being determining by the viable count on BHI agar. When grown in the absence of the antibiotic, the rates of plasmid loss were very low (92-80 %; Table 3.7) indicating that the plasmids themselves were very stable. However plasmid stability decreased with increasing concentration of the liquorice extract, suggesting that it was exerting a metabolic stress on the cells that encouraged plasmid loss. In the presence of the antibiotic in the growth medium, the plasmid was more stable (as expected) but even then plasmid loss was seen when the cells were grown in the presence of the liquorice extract (93-97 %; Table 3.8). Consequently, it was clear that the addition of the antibiotic alone is not causing a major stress on the bacterial cells. Although a concentration of 50 μ g ml⁻¹ inhibited the growth of bacteria, the bacteria could recover again as demonstrated by the formation of colonies in the viable count experiments after the stress was removed (Figure 3.13).

BHI agarListeria selectiveBHI agar plusnedia5 μg ml² Ery1 % (v/v) EtoH12.5 μg ml²50 μg ml²

Figure 3.13. Viable count for *Listeria monocytogenes* 10403S(pSB3008) on different agar plates

Listeria monocytogenes 10403S(pSB3008) treated with 12.5 and 50 μ g ml⁻¹ liquorice extract and 1 % (v/v) ethanol solvent, then incubated at 37 °C for 24 h then diluted by serial dilution and plated on BHI agar plate, Listeria selective media and BHI agar plate with 5 μ g ml⁻¹ Ery using Miles and Misra technique for a viable count to confirm the present of plasmid.

Table 3.7. Plasmid stability in LM 10403S without selection

BHI Agar with 5 μ g ml ⁻¹ Erm	BHI Agar without Erm	% Bacteria
= Plasmid Present in Cell (cfu ml ⁻¹)	= viable count (cfu ml ⁻¹)	with Plasmid
2.7x10 ⁹	3.2x10 ⁹	92 %
2.54x10 ⁹	3.06x10 ⁹	82 %
1x10 ⁶	1x10 ⁶	80 %
	BHI Agar with 5 μ g ml ⁻¹ Erm = Plasmid Present in Cell (cfu ml ⁻¹) 2.7x10 ⁹ 2.54x10 ⁹ 1x10 ⁶	BHI Agar with 5 μ g ml ⁻¹ ErmBHI Agar without Erm= Plasmid Present in Cell (cfu ml ⁻¹)= viable count (cfu ml ⁻¹)2.7x10 ⁹ $3.2x10^9$ 2.54x10 ⁹ $3.06x10^9$ 1x10 ⁶ $1x10^6$

*Erm: erythromycin

Table 3.8. Plasmid stability in LM 10403S with selection (ERM)

LM 10403S (BS10, luxABCDE, gfp) with	BHI Agar with 5 μ g ml ⁻¹ Erm =	BHI Agar without Erm	% Bacteria
5 μ g ml ⁻¹ Erm in the Growth Media	Plasmid Present in Cell (cfu ml ⁻¹)	= viable count (cfu ml ⁻¹)	with Plasmid
Control ethanol solvent	1.43×10 ¹⁰	1.34x10 ¹⁰	100 %
12.5 µg ml ⁻¹ liquorice extract	2.64x10 ⁹	2.83x10 ⁹	93 %
50 μ g ml ⁻¹ liquorice extract	9.7x10 ⁷	9.9x10 ⁷	97 %

Overnight culture of LM 10403S was prepared with and without erythromycin in the growth media and incubated for 24 h at 37 °C in Tecan, then serial dilutions was created and plated on the BHI agar plate with and without 5 µg ml⁻¹ Erm to confirm the stability of plasmid.

To see if the extract was also affecting the metabolic state of other bacteria, the effect of the extract on the Gram-positive bacterium *S. aureus* and the Gram-negative spoilage organism *P. fluorescens* bacteria were also tested as bioluminescent derivatives of these organisms were available in the laboratory culture collection. In particular it was important to determine whether, even if there was no evidence of growth inhibition in the Gram-negative bacteria, whether they did experience sub-lethal injury and reduced metabolic levels. For both these strains *S. aureus* and *P. fluorescens*, plasmid stability was not an issue since they were chromosomally tagged with a *lux* operon (*luxABCDE*) that was modified to function in each of them to give a highly bioluminescent phenotype.

The experiment was prepared as described previously in Section 3.2.8 and the growth and bioluminescence of *S. aureus* and *P. fluorescens* were compared by monitoring light and optical density simultaneous and comparing levels of bioluminescence produced (in relative light units [RLU] (Figure 3.14) after exposure to the liquorice extract. The measured light emission was high for P. *fluorescens* for both concentrations of the extract, indicating that there was no significant effect on metabolism. In contrast S. aureus showed a dramatic fall in light levels for both concentrations of the extract, and there was no difference between the basal level of light recorded and those measured when a concentration of 50 μ g ml⁻¹ of liquorice extract was added. There was however some recovery of the light levels during the incubation seen when 12.5 µg ml⁻¹ was used, and the increase in light corresponded to the time point where the cells left the lag phase and began to grow. There results confirmed that the extract does not have antimicrobial activity against Gram-negative bacteria, but in addition to affecting the cell wall formation of Gram-positive bacteria also seems to have an effect on cellular metabolism indicative of membrane damage.



Ligth emission rate (log₁₀)

1000

100

Ligth emission rate (log₁₀)

1000

100

Figure 3.14. Effect of liquorice extract on growth and metabolism of (a) *S. aureus* and (b) *P. fluorescence* chromosomally tagged *lux*



Overnight cultures of *S. aureus* and *P. fluorescens* chromosomally tagged with *lux* were used to inoculate fresh broth to an OD_{600nm} of 0.05 then cultures prepared with two concentrations of liquorice extract 12.5 [•], 50 [▲] µg ml⁻¹, 1 % [•] (v/v) of ethanol. A control sample [▼] containing broth only was also included. Samples incubated at 37 °C and growth monitored by optical density. Bioluminescence was monitored simultaneously and data presented as RLU/OD_{600nm}. Data are given as means ± standard deviations from two independent experiments, each performed in triplicate (n=3).

Because the antimicrobial effect was seen on all Gram-positive bacteria tested, an interesting question to ask is if it also affects acid-fast bacteria which are a part of Gram-positive group but have a very different cell wall structure. Specifically *Mycobacteria* have a complex cell wall structure containing many proteins, lipids, and carbohydrates, many of which are found only in these bacteria and have been identified as excellent targets for further investigations to discover anti-Mycobacterial compounds. So, just to see if the extract would also effect *Mycobacterium*, a growth experiment was carried out and the results are shown in Figure 3.15. As seen for the other Gram-positive bacteria, a concentration of 50 μ g ml⁻¹ was able to totally inhibit the growth of *Mycobacterium smegmatis*. Although this is a fast growing member of the *Mycobacterium* genus, the structure of the cell wall is very similar to that of the slow growing pathogens such as M. tuberculosis. Hence it is likely that this extract would also be effective against pathogenic mycobacterial species, although this needs to be formally demonstrated. Given the differences in the cell wall structures between these bacteria and other Gram-positive bacteria, the target affect must be a highly conserved element fundamental to these bacterial cell types.

Figure 3.15. The effect of liquorice extract on the growth of *Mycobacterium smegmatis*



Culture of *Mycobacterium smegmatis* was grown at 37 °C for 24 h in the presence of liquorice extract 12.5 [•], 50 [▲] µg ml⁻¹ and [•] 1 % (v/v) ethanol solvent. A control sample [▼] containing broth only was also included. Growth was monitored by optical density measurement at 600nm. Data are given as means ± standard deviations from two independent experiments, each performed in duplicate (n=3). Significant difference between a concentration 50 µg ml⁻¹ and 1 % (v/v) ethanol solvent is indicated as p < 0.0001.

3.2.10 Microscopic assay of membrane integrity after exposure to liquorice extract

In Section 3.2.8, the liquorice extract was shown to have an effect on metabolic activity, and one reason for this could be that it was having an effect on membrane integrity. Therefore the purpose of this study was to confirm whether there was any effect of liquorice extract on membrane integrity of Gram-positive bacteria.

Effects on membrane integrity were monitored using a LIVE/DEAD® viability stain and fluorescence microscopy according to the method given by Molecular Probes (2004) with slight modifications. Briefly, overnight cultures of L. monocytogenes, L. innocua and B. subtilis were prepared and aliquots inoculated into sterile 250 ml Erlenmeyer flasks containing 100 ml of BHI broth. The inoculum of the broth culture was adjusted until the bacterial suspension reached an OD_{600nm} equal to 0.05. Three individual flasks were inoculated with each strain; the first flask (control) contained 1 % (v/v) ethanol, the second flask contained 12.5 μ g ml⁻¹ of liquorice extract and the third flask contained 50 μ g ml⁻¹ of liquorice extract. To allow sufficient cell mass to be recovered to allow the cells to be detected by microscopy, samples of the cultures (6 ml) were transferred into a sterile disposable centrifuge tube using a 10 ml pipette. Cells were harvested by centrifugation (10,000 x g for 15 min), the supernatant was discarded and the pellet was re-suspended in 60 μ l of the recommended resuspension buffer (0.85 % NaCl) before mixing with LIVE/DEAD[®] BacLight[™] Bacterial Viability stain. Samples were taken immediately after the flasks were prepared (time = 0) and the flasks containing the remaining cultures were incubated at 37 °C in a shaking incubator (150 rpm) and further samples were taken after 3, 6 and 24 h of incubation. At each time point the cells were concentrated before staining with the LIVE/DEAD[®] BacLight[™] Bacterial Viability Kit as described in Section 2.15.

The fluorescence microscope images (Figure 3.16) showed that for all control cultures over time there was an increase in cell number which would be expected in growing cultures and most of the cells taken at all time points showed predominantly green fluorescence (stained with SYTO9 dye), indicating that even though the cultures had reached stationary phase after 24 h incubation the cells were still metabolically active. In the *Bacillus* culture there was some evidence of some orange/red cells present after 24 h incubation, which may be due to cells that have initiated sporulation when the cells entered stationary phase. However from these results with the control there was no evidence of any effect of the ethanol on membrane integrity.

In contrast the cells treated with the extract at 50 µg ml⁻¹ showed a large number of cells that were unable to remove the propidium iodide dye, as they were stained red or orange, a characteristic of cells with membrane damage. This was evident even at the first sampling time point but the number of red/orange cells increased over time (Figure 3.16). In addition the intensity of the red stain also appeared to increase with time so, three sub-populations (dead, viable and injured cells) could be distinguished using this viability kit.

In the samples containing the 12.5 μ g ml⁻¹ of extract, the number of red cells accumulating was lower in each of the samples, and for both *Listeria* species tested, did not seem to increase over time suggesting that the cells may have been able to adapt to this challenge. These results are consistent with the pattern of growth seen in Section 3.2.8 when strains were challenged with 12.5 μ g ml⁻¹ of the extract, where there seemed to growth at near normal rates after an extended period. In contrast for *Bacillus* treated with 12.5 μ g ml⁻¹ of extract, the number of red/orange cells increased over time perhaps indicating that for this sporulating organism the cells responded in a different way to the initial challenge and initiated sporulation rather than using energy to adapt to the antimicrobial challenge.

Figure 3.16. The effect of liquorice extract on membrane integrity after applying LIVE/DEAD[®] BacLight[™] Kit



(a) Listeria monocytogenes

(b) Listeria innocua



(c) Bacillus subtilis



LIVE/DEAD[®] BacLight[™] staining kit applied to three bacterial samples. Culture of *L. monocytogenes, L. innocua* and *B. subtilis* were grown in BHI broth overnight at 37 °C to an OD_{660nm} of 0.05. The cells were incubated with liquorice extract (12.5 & 50 µg ml⁻¹) and 1 % (v/v) ethanol solvent for 24 h at 37 °C. The samples were then stained with the LIVE/DEAD BacLight Bacterial Kit-L-7012. 5 µl aliquot was examined by fluorescence microscopy. Control cells (incubated without extract) appear green whereas the cells incubated with extract were fewer in number and appeared red at the 24-h point. In general, it shows that 3–6 h exposure to extract seems to be a susceptible time period after which the rate of cell damage seems rapid in positive bacterial strains tested in this study.

These results were consistent with the results in the previous Section 3.2.8 (Figure 3.12) where a decrease in metabolic activity was detected using bioluminescent strains of bacteria, and provides more evidence that after treatment with liquorice extract and there is inhibition of growth rather than killing of the bacteria. It also suggests that the effect on the metabolic activity for these organisms may be due to an effect on the membrane integrity. However the results in Section 3.2.8 also showed that cell growth was inhibited and that cell structure was changed and therefore there could also be effects on cell division.

Comparing these results with the growth curves of these strains (see Figure 3.17), it is seen that 50 μ g ml⁻¹ of liquorice extract totally inhibited the growth of *Listeria* and *Bacillus* and therefore we expect a high number of damaged cells to appear. However, as shown in Chapter 3, the treatment is bacteriostatic rather than bactericidal and therefore it would not be expected that all of the cells in the population were non-viable. However at a concentration of 12.5 μ g ml⁻¹ the growth of *Bacillus* was lower than the control sample whereas at this concentration slightly inhibited the growth of *Listeria* compared to the control. Hence the pattern of growth appears to correspond well with the fluorescent images since at this concentration the *Listeria* cells were more green and *Bacillus* samples contained more orange/red cells. This result confirmed the earlier suggestion that, as reported for Gfp, fluorescence intensity can be used as an indirect measure of viable cell number as fluorescence levels per cell are relatively constant, unlike bioluminescence levels (Lehtinen *et al.*, 2006)

Figure 3.17. Comparison of effects of liquorice on growth and viability staining



This figure represented the growth curve of *Listeria* and *Bacillus* strains which was grown in BHI broth with two concentrations of liquorice extract [•] 12.5 μ g ml⁻¹ and [•] 50 μ g ml⁻¹ and BHI broth as a control [•], the X axis is indicate to the time and the Y axis is indicate to the optical density, the images were taken after treatment of cells with and without liquorice extract and stained with Live/Dead stain, again showed that the *Listeria* was grown more than the *Bacillus* as see the growth curve matched with the images taken after dye of cells with Live/Dead stain.

3.2.11 Use of *gpf*⁺ marked strains to confirm *Listeria* identification

In Section 3.2.6, changes in cell wall structure and cell shape were detected using the Gram stain method. However these cultures were incubated overnight before the cells were visualized, and Gram-staining requires fixing and treating the cells with dyes which may lead to artefacts if the cells are sub-lethally damaged and then treated with further chemicals. In addition it is difficult to prove that the change in cell staining is not due to contamination of the culture by cells that out compete the strain of interest. Both of these possibilities could be addressed using the strains that had been transformed to a fluorescent phenotype.

The experiment described in Section 3.2.7 confirmed that the Gfp was present in all *Listeria* strains before starting treatment with liquorice extract, thus to confirm that the changes in cell morphology were occurring in the cells being treated rather than apparent changes due to the presence of a contaminating microbe, in this Section it was decided to confirm that GFP still remained inside the cells after application of liquorice extract as proof that the changes in cell morphology were occurring in *Listeria* cells and were not due to contamination. To use fluorescence to monitor the effect of the liquorice extract, *L. monocytogenes* 23074(pSB3008) was cultivated in 10 ml of brain heart infusion broth, supplemented with Erythromycin (5 μ g ml⁻¹) to select for the plasmid, in a shaking incubator (150 rpm) at 37 °C. The overnight culture of bacteria was adjusted to approximately 1 X 10^7 cells ml⁻¹ (OD ~ 0.05) and then two concentrations of liquorice extract (12.5 and 50 µg ml⁻¹) were added into the culture broth and mixed well. As before 1 % (v/v) ethanol was used as a control and a control sample containing uninoculated BHI broth only was also included. Eight replicate samples (200 µl) for each treatment were transferred into a 96-well microtiter plate which was incubated in a Tecan microplate reader for 24 h at 37 °C.

After incubation, to confirm that the Gfp protein was being expressed at detectable levels, the first stage of the Gram staining procedure was used to

prepare samples for microscopy. Specifically the samples from the microtitre wells were transferred into an Eppendorf tube then centrifuged at 13,000 x g for 2 min, then the broth was removed and Phosphate Buffered Saline (PBS w/v) was added into the pellet. The samples were mixed thoroughly by using a vortex mixer, then a loop was used to transfer a small amount of culture on to a slide which was left until dry. PBS solution (20 μ I) was added to permeablise the unfixed cells. The sample was then covered with a coverslip and imaged using the x1000 oil immersion lens under the fluorescence microscope to detect Gfp fluorescence.

The results showed that even though the cells were grown with antibiotics to ensure that they contained the plasmid, however, not all cells had the ability to make enough GFP to detect fluorescence, as only a small proportion of *Listeria* cells in the population were seen to fluoresce when examined by fluorescence microscope (Figure 3.18). This is consistent with the results seen in Section 3.2.7 when the fluorescent phenotype of the strains carrying the dual operon were characterised. In this case when individual colonies were grown in a microtitre plate to assess the variability on fluorescence levels, although there was not much difference in the average fluorescence of the replicates, there was a large variation in the fluorescent output of the individual samples. Since they were all incubated until they reached stationary phase, this suggests that the different samples contained a variable number of cells that did not give a fluorescent phenotype.

Despite this limitation it was still important to confirm some of the cells detected with altered morphology had the ability to fluorescence, so samples were viewed using the fluorescence microscope after treatment with 12.5 and 50 µg ml⁻¹ of liquorice extract to confirm if any of the cells were detectably green. Only a small proportion of *Listeria* strain population was seen to fluoresce when examined by fluorescence microscope (Figure 3.18), but the presence of the *gfp* gene protein confirmed that cells with altered cell walls are not contaminants. However it was not clear whether the cells that gave a fluorescent signal were a specific sub-

population of the culture or were just a random selection of cells that had retained sufficient Gfp protein to be detectable. There was some evidence that this was the case since none of the fluorescent cells seen in the sample treated with 50 µg ml⁻ ¹ shown in Figure 3.18 were elongated in any way compared to those in the control sample. So perhaps the cells that retained the Gfp phenotype were those cells in the population that were green when detected using the BacLight[™] stain (see Figure 3.16).

Figure 3.18. Detection of fluorescence in *L. monocytogenes* 23074(pSB3008) culture after incubation with and without liquorice extract



1 % (v/v) EtoH

12.5 µg ml⁻¹

50 µg ml⁻¹

Fluorescence of *L. monocytogenes* 23074(pSB3008) culture after incubation for 24 h at 37 °C with a two concentrations (12.5 and 50 μ g ml⁻¹) of liquorice extract as well as 1 % (v/v) ethanol solvent, the green cells were seen under the fluorescence microscope at all treatments to detect Gfp fluorescence.

3.2.12 Morphology changes and cell elongation

As the fluorescent marker was not stable enough to allow the cultures of Listeria to be directly visualised, it was decided to investigate morphological changes and cell elongation in *L. monocytogenes* caused by liquorice extract when the cells exposed to stress using the BacLight[™] stain as all cells should be visible. Overnight culture of Listeria monocytogenes 23074 was diluted to 0.05 of optical density then two flasks were prepared with 12.5 and 50 µg ml⁻¹ of liquorice extract as well as 1 % (v/v) ethanol solvent. Sample (200 ml) was transferred into microtiter plate. The plate was subsequently incubated for 24 h at 37 °C. Finally, the cells were harvested and exposed to a dye LIVE/DEAD[®] BacLight[™] as a described in Section 3.2.10. In order to visualize the potential existence of septa in elongated cells, a loop of culture was transferred into slide and the changes in the morphology of cells (cell elongation) were observed under fluorescent microscopes, the images of *Listeria* that had been treated with 50 μ g ml⁻¹ of liquorice extract, after 24 h of incubation are presented in Figure 3.19. Cells treated with liquorice extract for 24 h showed various morphological alteration and damage to the outer surface of the cells as well as clumping of the cells. Numerous cells appeared longer compared to control.

Figure 3.19. Effect confirmed on the elongation of *Listeria monocytogenes* 23074(pSB3008) after 24 h treatment with 50 μ g ml⁻¹



Control



50 μg ml⁻¹ liquorice extract

Listeria monocytogenes 23074(pSB3008) was exposed to 50 µg ml⁻¹ of liquorice extract and after 24 h of incubation time, the cells were harvested and stained with Live/Dead stain. Under the fluorescence microscope the presence of elongated cells was very evident and the presence of structures that resembled septa (examples indicated with arrow) were evidence between normal size cells, while the untreated sample showed just normal cell structures.

In this study, long filament formations were observed when *Listeria* exposed to liquorice extract that indicate changes in cell length. When bacteria are exposed to stress conditions, the can induce the formation of filaments (Liu *et al.*, 2014). In the literature there are many reports that *Listeria monocytogenes* does induce filament formation when the environment became more challenging, in particular when food preservatives are applied that inhibit cell growth (Isom *et al.*, 1995; Zarei *et al.*, 2012; Hazeleger *et al.*, 2006).

Filaments were apparent at 50 μ g ml⁻¹, the most predominant filaments occurred after 12 h treatment (Figure 3.20 c). Along with the presence of filaments, normal coccoid rods were observed under all conditions of the treatment with 12.5 μ g ml⁻¹ (Figure 3.20 b). There were no morphological changes observed in BHI media with 1 % (v/v) ethanol solvent (Figure 3.20 a). Hence changes in bacterial morphology were associated with *L. monocytogenes* cells being treated with liquorice and it was found that the extract induced alterations in the morphology of bacterial cells proportionate to concentration of the extract and time of exposure.

Figure 3.20. Filament formation in *L. monocytogenes* cells grown in BHI broth with and without liquorice extract



1 % (v/v) ethanol solvent

12.5 µg ml⁻¹

50 µg ml⁻¹

Images showing morphology of treated culture of *Listeria monocytogenes* when treated with 1 % (v/v) ethanol solvent and a two concentration (12.5 and 50 μ g ml⁻¹) of liquorice extracts. Samples shown in panel (a) untreated culture no morphological changes (b) 12.5 μ g ml⁻¹ over 24 h showed the cells were green with normal coccoid rods (c) with 50 μ g ml⁻¹ of liquorice extract the dead cells stained red and the live cells stained green and cells appeared longer compared to control. Images were taken by a fluorescence microscope at 100x total magnification with camera (Canon).

3.3 Discussion

The desire of the public for more natural food preservatives has driven an interest in novel antimicrobial agents derived from plants. As is well known, plants synthesize different groups of secondary metabolites (phytochemicals) which contribute to defence mechanisms, as well as it is recognized that some of these molecules have useful effects such as acting as antimicrobial agents. The current research has produced a more comprehensive picture of the effect of one such natural compound produce by plants on the growth of Gram-positive bacteria revealing an effect on the cell wall, indicated by changes in cell shape as well as the inability to retain the dyes present in Gram-stain reagents.

All Gram-negative bacteria tested in this study were comparably resistant to liquorice extract. In contrast the liquorice extract had a specific effect against Gram-positive bacteria (Figure 3.6 c, e& f). These results are not totally consistent with studies performed by other researchers which described antimicrobial activity of liquorice extract against both Gram-positive and Gram-negative bacteria. For instance Karami et al. (2013) showed ethanol extracts of Glycyrrhiza glabra root collected at different times of the year exhibited potent antimicrobial activity against S. enterditis, E. coli and S. aureus with the MICs of 0.8 mg ml⁻¹ and MBCs of 0.9 mg ml⁻¹ whereas *B. cereus* was the most resistant species with MICs of 1.0 mg ml⁻¹ and MBCs of 3.0 mg ml⁻¹. Gupta *et al*. (2013) also studied the antimicrobial activity of a methanolic extract of G. glabra against Escherichia coli, Vibrio cholerae, Staphylococcus aureus, Bacillus cereus and Bacillus subtilis strains using well-agar diffusion method and they also found that the extract had antibacterial activity against all the bacteria tested. In contrast these results agreed with the report by Tsukiyama et al. (2002) in which the extract of roots of liquorice (G. inflata), licochalcone A showed effects against all Gram-positive bacteria tested with MICs of 2 to 15 μ g ml⁻¹, and the extract especially was effective against all Bacillus spp., with MICs of 2 to 3 μ g ml⁻¹, but was not effective against Gram-

negative bacteria or eukaryotes. These results, and the results gained in this study, show that Gram-negative bacteria are more resistant to the extracts than the tested Gram-positive bacteria, and this probably could most likely be explained by the differences in the cell wall structure, since the lipopolysaccharide layer in the outer membrane of Gram-negative bacteria is known to serve as an active permeability barrier to many environmental substances, specifically acting as a protective barrier against hydrophobic compounds (Puupponen-Pimiä *et al.*, 2001).

The differences in the results gained in these different studies might also be attributable to the fact that the content of these components in Glycyrrhiza vary depending on many factors including different plant species, plant age, seasonal differences, geographical changes and time of harvest and collect plant, and all of these divers factors can affect their biological activity (Fiore et al., 2008). Known phytochemical groups with antimicrobial activity from liquorice extracts can include alkaloids, saponins, flavonoids, tannins, glycosides and phenols (Meghashri, 2009). Thus, the effect of liquorice extracts against Gram-positive and Gram-negative bacteria may be due to the presence or absence of certain effective biological components or a difference in the quantity of active compounds responsible for bio- efficiency. However the extract provided by Phytoquest Ltd for these experiments was not a direct extract of liquorice but was prepared from the waste material left over after liquorice flavourings were prepared for food products and therefore this extract may contain a different set of bioactive compounds. According to the analysis provided by Phytoquest Ltd, the active constituents in the waste material were Licoricidin and Licorisoflavan A.

Natural preservatives have been used as alternatives to traditional chemical preservatives; however, a limited number have been commercially developed and many remain to be investigated as sources of safer and effective antimicrobials. In this study, we have been investigating the antimicrobial activity

of an extract of *Glycyrrhiza uralensis* (liquorice) that was provided as waste material from the production of liquorice flavourings for the food industry, and to investigate if this retained the expected antimicrobial activity so it could be used as a natural preservative. Testing different concentrations of extract is important when evaluating antimicrobial activity and in this case it revealed that sub-inhibitory concentrations of the extract still had an effect on growth rate and cell metabolism (Figure 3.12). Since food preservatives can be used in combination at this sub-inhibitory levels to produce an overall inhibitory effect hurdle technology; see (Singh and Shalini, 2016), then this information may be useful if this extract is applied in a real food application.

L. monocytogenes is quite fastidious and requires a range of different amino acids and vitamins that it cannot synthesize by itself, so generally it is grown in rich media containing these compounds (Jarvis, 2016). Thus BHI is the most commonly used non-selective media for cultivation of *Listeria* species which is considered rich culture medium that provides all of these growth factors (Jones and D'Orazio, 2013). Therefore in these experiments all bacteria were grown in this medium but while it was found that using brain heart infusion broth most of the bacteria exhibited good growth, the growth rates achieved were very different (see Table 3.1) which then required extended incubation times to achieve a high enough cell density to allow all of the cultures to be prepared the same starting concentration. Since the growth optimal of the bacteria tested were all very similar, these differences in growth rate indicate that not all of the strains being compared are in the same physiological state and this needs to be factored in when assessing antimicrobial effects as the additional stress may make the cells more sensitive to the extract.

Several techniques using different bacterial physiological indicators such as MIC, MBC and metabolism have been used in this study in order to evaluate the extract as antimicrobial agent. Traditionally determining the MBC is performed

using viable count. However, performing these experiments using shake flasks is very laborious and therefore developing a new method to rapidly screen bacteria utilizing a microtitre culture dish and recording cell density automatically using a plate reader allowed a number of concentrations of the compound and simultaneous testing against a number of different bacterial types to be carried out and allow more data to be rapidly generated. However, a limitation of this approach is that when growth inhibition occurs the question remains of whether the effect of the antimicrobial is bacteriostatic or bactericidal. This signifies the need for further tests to further interpret these results. In this case parallel experiments using viable count showed that the effect of the extract was bacteriostatic rather than bactericidal and that the effective concentration is 50 μ g ml⁻¹ (see Figure 3.7). This is a very low concentration of material, and therefore it may be practical to use it to inhibit Gram-positive bacteria in some foods. According to Kim et al. (2002), in agreement with our study, the MIC values for glycyrrhizin, a-glycyrrhetinic acid and β -Glycyrrhetinic acid isolated from *Glycyrrhiza glabra* had antibacterial activity at 7.6 and 12.5 μ g ml⁻¹ against Bacillus subtilis and Staphylococcus epidermidis, yet it was not effective against Gram-negative bacteria *Escherichia coli*, *Proteus vulgaris* and fungi.

The emergence of resistance of bacterial strains to antibiotics may be due to the excessive usage of antibiotics and transmission of resistance within and between individuals, where the genes causing resistance can be transferred between different strains of microorganism, then recipient organisms will also become resistant (Cetin-Karaca, 2011). Therefore, there is an urgent need to find solutions to overcome bacterial resistance. Hence, natural alternatives have been also sought for use as new antimicrobial agents as alternatives to antibiotics. Antimicrobial phytochemicals have mechanisms of action similar to antibiotic action in terms of impact and includes inhibition of nucleic acid synthesis (Simões *et al.*, 2012), cell wall synthesis, cell membrane damage and inhibition of

respiration (Yap *et al.*, 2014; Mc Dermott *et al.*, 2003) and it's important to understand the antimicrobial properties of plant species, that could help to contribute to the development of drugs against microorganisms of medical interest.

A Gram stain performed on L. monocytogenes, L. innocua and B. subtilis demonstrated the altered on cell morphology as compared to control group (Figure 3.8), and indicated that Gram-positive bacteria may be more susceptible to the liquorice extract than Gram-negative bacteria due an effect on cell wall synthesis. The fact that a range of different cell types were tested (Gram-positive rods, Gram-positive cocci and spore-forming organisms), and all showed equal sensitivity, suggests that there is a common target in all these different types of Gram-positive bacteria. Using *L. monocytogenes* strains and *L. innocua* as model organisms, the nature of the growth inhibition seen was further investigated, and it was clear that cell growth, including cell division, was affected. At sub-lethal concentrations, even when the growth rate was not particularly affected (Figure 3.10), there was evidence of sub-lethal injury and cell stress (Figure 3.12). Thus, it has been demonstrated that liquorice extract is able to inhibit the growth of Gram-positive bacteria such as L. monocytogenes, L. innocua, S. aureus, E. faecalis and B. subtilis (Figure 3.4). In addition an effect on the growth of Mycobacterium smegmatis (Figure 3.15), which is considered genetically to be a member of the Gram-positive group was seen, even though the mycobacteria do not give a typical Gram stain result because of the high lipid content of its cell wall (Fair and Tor, 2014).

The fact that all types of Gram-positive bacteria tested to date were inhibited, suggests that the primary target of the liquorice extract is something common to Gram-positive bacteria and that it may inhibit cell wall synthesis leading to disruption of the normal synthesis of the peptidoglycan layer. This raises a question of whether the effect on the cell wall may also there is effect on the

membrane integrity through the effect on the metabolic activity, so, strains carrying *lux* genes were chosen to confirm the effect on the membrane integrity.

Monitoring the bioluminescence levels of the genetically engineered lux derivatives of these bacteria, it was clear that even when growth was not affected , a reduction in the light output could be seen. The bacterial *lux* genes are used to monitor cell injury due to the fact that light levels produced by cultures reflects their metabolic state, and agents that inhibit cell metabolism or affect membrane permeability result in decrease levels of bioluminescence. Since it has been shown that light levels produced by cultures reflects their metabolic state, agents that inhibit cell metabolism or affect membrane permeability result in reduced levels of bioluminescence (Robinson et al., 2011). This was confirmed by study reported by (Gregor et al., 2018; Unge et al., 1999), the bioluminescence emitted is directly related to the number of viable cells, since $FMNH_2$ is only produced by metabolically active cells thus allowing an estimation of the sub-lethal effects of antimicrobial agents on the viability of target bacteria to be monitored. Although this approach has been used before by many workers, some caution must be taken to ensure that reduction in light levels are not simply due to the loss of the plasmid from the strains. In this case, we found that the stress caused by treating the bacteria with the liquorice extract did exert some additional metabolic pressure on the cells, leading to plasmid loss, unless antibiotic was added to the media, but there was no evidence of a difference in results gained with and without antibiotic to select for the plasmid (Table 3.7 and Table 3.8) and therefore in this case we can be confident that the loss of light was due to sub-lethal injury.

Treatment with 50 µg ml⁻¹ of liquorice extract completely inhibited the growth of all of the Gram-positive bacteria tested, but when plated on to media without the extract the treated cells were able to grow normally indicating that effect was bacteriostatic rather than bactericidal. It is interesting to note that without the use of the bioluminescent reporter genes, the separate effect on cell

metabolism would not have been evident, and therefore this shows that using these tools is a powerful way to better identify stresses that have an effect on metabolic activity through damaging the membrane integrity or affecting enzymes associated with energy generation. So, more studies are needed to confirm that the extract may cause damage to the cell membrane.

Another area of caution is that strains of bacteria such as *Listeria* that have been cultured for long periods of time in the laboratory may have lost some of their fitness. To address this, we tested a range of different isolates, including examples of the two serovars that are responsible for the majority of human infections. Again no difference was seen between the sensitivity of different strains indicating that the results gained were very reproducible and not dependent on the test strain used. In addition, we tested two well characterised mutants of *L. monocytogenes*, one which over expressed the virulence genes (NCTC 7973 prfA*) and another than is incapable of expressing the virulence genes due to a deletion of the virulence gene regulator (NCTC 10357; Δ pfrA). The results gained with these strains were also identical to the results gained with wild-type *L. monocytogenes* and *L. innocua*, which has a complete absence of any of the virulence genes found in *Listeria*. Again these results suggest that the target site is found in all cell types and is not associated with the virulence traits of these bacteria.

The mechanism of action of the liquorice extract was further investigated using the LIVE/DEAD[®] BacLight[™] Bacterial Viability Kit for microscopy and quantitative evaluation of treatment. Bacteria was treated with an inhibitory (50 µg ml⁻¹) or sub-inhibitory (12.5 µg ml⁻¹) concentration of liquorice extract. Exposure to the MIC for this extract (50 µg ml⁻¹) resulted in total inhibition of bacteria growth and samples imaged using the LIVE/DEAD stain showed that the cells were unable to actively remove the fluorescent dye from the cytoplasm, suggesting that membrane integrity was affected. Exposure to a sub-lethal level

of the extract (12.5 μ g ml⁻¹) resulted in some growth inhibition and a lower percentage of cells showing evidence of membrane damage. However it must be noted that the cells require a membrane potential to exclude the PI from the cell, and if their metabolism is affected it is also possible that they are not able to efficiently remove the PI from the cell rather than there being a direct damaging effect on the membrane. Consequently, disrupting the membrane lead to cell death, if the cell is not able to repair the damage

The staining of bacterial cells detected three various cell populations: living, compromised, and dead cells as a result separation of the cell occur according to the different viability stages when dealing with two types of stain. These changes in fluorescence intensity are associated with the different viability stages of *Listeria* and Bacillus cells when were revealed by using a sub-inhibitory concentrations of liquorice. On this occasion Listeria grow more than the Bacillus which means Bacillus was more affected than the *Listeria*, as illustrated in Figure 3.17. However previously it was seen that a concentration of 12.5 µg ml⁻¹ of the extract did cause growth inhibition of Listeria and resulted in lower levels of metabolism. This suggests that either there is some variability in the level of active ingredient in the liquorice extract and different solutions prepared on different days had different potency or that this level of the extract is very close to the threshold where no observable effect is seen so small variations in the concentration during preparation of the samples means that it is easy to drop below that threshold. However, at a concentration of 50 μ g ml⁻¹ the growth of Gram-positive bacteria was consistently inhibited and cell injury was evident even immediately after treatment (appearance of red cells in 0 time point) (Figure 3.16 a, b& c), although the number of orange cells detected over the time of exposure increased suggesting that the cells were being slowly killed by the extract. Thus, the remaining population of *L. monocytogenes* after liquorice extract treatment could be described by the decline of metabolic activity of cells, especially those treated
with 50 µg ml⁻¹ of extract. The fact that an increasing proportion of red-stained cells were detected over time confirmed that the membrane was being damaged but for both *Listeria* and *Bacillus* samples treated with the extract, there was also an increase in cell length and the appearance of chains of cells were seen during the 24 h of exposure, suggesting that cell division was also being affected.

Moreover, there was evidence of septa in the elongated cells in L. monocytogenes after 24 h of the stress being applied (Figure 3.19) and the length of these filaments continuously increases with time and, progressively, filaments with 4 or more cells are formed (Figures 3.16 and 3.20). Filament formation is considered to be caused by problems in cell division as well as changes in gene and protein expression (Bereksi et al., 2002; Kieboom et al., 2002; Margolin, 2000). These filamentous cells are longer than single cells but have approximately the same width (Giotis et al., 2007). Hence cells that become elongated increase their surface-to-volume ratio which can occur in response to environmental changes or stress (Harris and Theriot, 2016). These results directly confirmed that L. monocytogenes undergoes significant changes in cell morphology under the stress. We have found that a concentration of 50 μ g ml⁻¹ has shown inhibition of growth for Listeria and Bacillus, as shown by results from experiments after 24 h, cells elongate and filaments are formed. One point to note about the dyes that are typically employed to assess membrane integrity, is that they may be ineffective against cells with a hardy membrane or cell wall, such as spores (Luna et al., 2002; Bosshard et al., 2002). Although the Bacillus has the ability to form spores that make it very resistant to different stress in the environment such as antibiotics, temperature and dyes than other species of bacteria non-spore forming bacteria (Piktel et al., 2017; Roszak and Colwell, 1987), in this study we found that the vegetative cells were susceptible to liquorice extract and this limitation was not relevant for this study.

Several adverse environmental conditions, such as exposure to acids, high CO₂ concentration, high osmolality, growth at non-optimum temperatures, low pH and antimicrobial agents, are known to lead to morphological changes in bacterial cells as reaction of these stresses. Thus, for several foodborne pathogens including Salmonella, Escherichia coli, Bacillus, Clostridium and L. monocytogenes, it has been reported that stress conditions may cause elongation of cells (Gill et al., 2007; Hazeleger et al., 2006; Everis and Betts, 2001; Jones et al., 2013). Specifically for *Listeria* the formation of elongated (filamentous) cells has been reported on exposure to low pH 5.0 at 37 °C (Bereksi et al., 2002; Brzin, 1973; Brzin, 1975; Isom *et al.*, 1995; Jørgensen *et al.*, 1995), osmotic stress (Jørgensen et al., 1995), above-optimum growth temperature, (i.e. 42.5 °C; Rowan and Anderson, 1998), and the presence of antimicrobial agents such as trimethoprim and co-trimoxazole (Minkowski et al., 2001). Isom et al. (1995) observed that filament formation by L. monocytogenes in cultures containing high levels of NaCl as environmental challenge increased, the length of the filaments increased. In addition, cells of L. monocytogenes grown in the presence of 9 % NaCl were became more thermotolerant than cells were grown in low-salt medium which exposure to the same concentration of NaCl for up to 4 h (Jørgensen et al., 1995). These findings are in line with the results seen when the *Listeria* cells were treated with liquorice extract, indicating that the cells are experiencing cell a kind of stress. In particular the fact that staining treated samples with the BacLight[™] kit revealed the presence of septa within the filaments in *L. monocytogenes*, consistent with observations that during cell filamentation, it is the cell separation event that is inhibited. Consistent with this idea on the images of the stained *Listeria* cells the cytoplasmic material did seem to be separated into individual cells suggesting that the filaments actually consisted of several normal size cells that have not completed the cell division event.

Several reports indicate that, in some of these cases, removal of such stresses results in the slow return to the normal cell forms in around 24 h, further supporting the idea that the filamentation of *L. monocytogenes* may be an adaptive response to adverse conditions (Brzin, 1973; Isom et al., 1995). However, the mechanisms involved in these changes are unclear (Gardan et al., 2003). Since the cells plated out from these experiments were able to form normal colonies, this suggest that of the filamented *L. monocytogenes* cells have the ability to revert back to the normal form within one day of incubation at 37 °C when the stress was removed. This is in agreement with the findings by (Brzin, 1973; Brzin, 1975), which showed that *L. monocytogenes* had the ability to form filamentous only as long as the bacteria was grown on media containing high concentrations of NaCl. Following subculture to media without the appropriate high concentrations of NaCl, cell division resumed and normal rods were produced. These findings support the hypothesis that morphological filamentation is a direct result of bacterial adaptation to adverse environmental conditions. Another study conducted by Giotis et al. (2007) reported that L. monocytogenes 10403S was exposed to sublethal alkaline stress (pH 9.5) led to formed filaments or elongated chains, the cell length was increased over the time, during 1-3 h the cells showed quick develop in filaments/chains. However, these changes were not observed in control samples. When alkali stressed cells were moved into neutral medium, filamentous/chain forms returned to normal cellular morphology. Our results seem to support the observations of Kusumaningrum (2003) that these elongated cells are actually on the verge of division and, when transferred to more favourable conditions, will divide rapidly in single cells and start growing. If this happens in practice, it could have a significant impact on food safety, for instance when elongated pathogens are present. If that food is subsequently kept at conditions where growth is possible, the filaments will split up rapidly into many cells, resulting in a highly contaminated food product.

3.4 Conclusion

The antibacterial activity of liquorice extract was screened for evidence of growth inhibition against eight species of Gram-negative and Gram-positive bacteria, including L. monocytogenes, L. innocua, S. aureus, E. faecalis and B. subtilis. The Gram-negative bacteria tested include P. aeruginosa, E. coli and S. Typhimurium but none of these were affected by the extract. In contrast, for all of the Gram-positive bacteria tested, growth was inhibited as monitored using optical density. However parallel studies using viable count indicated that the cells were not killed meaning that the extract was bacteriostatic rather than bactericidal. The MIC and MBC of the extract were also determined, and a concentration of 50 μ g ml⁻¹ was found to have a strong bacteriostatic effect on Gram-positive bacteria. Microscopic analysis indicated that there were changes in cell shape suggesting the cell wall was affected. In addition, the use of a reporter strain of Listeria transformed with the bioluminescence genes *luxABCDE* indicated that cell energy levels were reduced when treated with either 12.5 or 50 μ g ml⁻¹ of the extract, with the reduction in light output being proportional to the concentration of the extract used. Together these results suggest that the extract is inhibiting the growth of Gram-positive bacteria only by damaging the cell wall and/or membrane, all measurements were performed the optical density, bioluminescent and green fluorescence protein (GFP) as a quick and reliable techniques for assessing the bacteriostatic, bactericidal or an effects on the metabolic activity during damaging the membrane integrity by using the liquorice extract as the antimicrobial against Gram-positive bacteria. This study confirms that potential of plant-derived molecules could use as a sustainable source of new broad-spectrum antimicrobial products.

Together these results indicate that the liquorice extract targets a site in the cell wall of Gram-positive bacteria and it has potential as a natural food preservative either for high risk *Listeria* products or for products where spoilage occurs due to Gram-positive bacteria. This study contributes to an increasing

number of reports on the occurrence of high numbers of filamentous bacteria under stress conditions and indicates that such morphological changes might play a role in bacterial defense and/or virulence. The next question to be asked was whether this extract could be used to control microbial growth in real food system and whether there was evidence that it was safe for human consumption.

CHAPTER 4

INVESTIGATION OF MECHANISMS OF ACTION OF LIQUORICE EXTRACT

AS A FOOD PRESERVATIVE

Abstract

Recently consumers have expressed a preference for more natural foods, but still expect high levels of food safety. This has led to a search for natural antimicrobials that can be used to control the growth of food-borne pathogens in foods. Currently, plant products are being considered as a significant alternative source for novel antimicrobial drugs against antibiotic-resistant microorganism and as preservatives of food.

It was shown in the last Chapter that a liquorice plant extract, prepared from a waste product from the preparation of edible liquorice, is able to inhibit the growth of *Listeria*, resulting in lower levels of metabolic activity. In addition we showed that it was able to inhibit the growth of some other Gram-positive pathogens. In this Chapter an investigation of the effect of liquorice extract as a natural antimicrobial agent on the growth of *L. monocytogenes* 10403S cells applied to the surface of sliced deli meat was carried out. Two initial inoculum concentrations were used in this experiment, 10⁶ and 10⁸ cfu ml⁻¹. Liquorice exhibited potent activity against *L. monocytogenes*, but the observed effects were dependent on the size of inoculum in the culture medium, and at low inoculum, inhibition of *Listeria* growth was seen for up to 10 days. Although, growth inhibition was seen with a high inoculum, out growth was seen after extended incubation. Hence the liquorice extract has potential as a natural food preservative and could be of use to improve the safety of RTE product, in particular could be of use to improve the safety of sliced deli meats.

4.1 Introduction

In the previous Chapters, it was found that the liquorice extract has antimicrobial activity against the Gram-positive bacterium *Listeria*. *L. monocytogenes* contamination is a particular problem associated with sliced deli meats, but other Gram-positive bacteria, such as *Bacillus*, *Brochothrix* and Lactic

Acid Bacteria commonly contribute to the spoilage of cooked meats (Møretrø and Langsrud, 2017) and therefore if the growth of all of these bacteria were inhibited in product it would be of benefit to the food industry and consumer. Having established that the effect of the liquorice extract was bacteriostatic, and that it caused damage to the cell wall and cell membrane, further studies were now required to determine if this extract could be used in a practical sense in a real food system as a preservative. There are several techniques utilized to investigate the antimicrobial activity of food preservatives, but they are very variable depending on the type of product used. In this case, to further investigate the mechanism of action of the liquorice extract, and towards validating its application as a food preservative, a sliced meat model was chosen.

Microbial Food safety is a major concern for consumers throughout the world, as well as regulatory agencies and food industries. This is due to foodborne disease, as well as spoilage, where growth of microbes cause unwanted changes in food that leads to the alteration of food quality and causes sensory quality deterioration. For this reason there is a growing consumer demand for natural preservatives instead of chemical compounds. Thus there is growing attention by researchers to the discovery of natural antimicrobials to use in food products to extend shelf life as well as to control or inhibit microbial growth (Goni *et al.*, 2009; Fattouch *et al.*, 200).

L. monocytogenes is widely distributed in meat processing plants (Hof, 2003), and many listeriosis outbreaks have been reported due to the consumption of ready-to-eat (RTE) meat products (Barmpalia *et al.*, 2005). Within the factory environment, *L. monocytogenes* is considered as a potential cause of contaminated food products, in particular during the slicing of meats, if proper controls are not in place (Swaminathan and Gerner-Smidt, 2007). Although, pathogenic organisms can be destroyed by the application of high temperature to food products, products can be contaminated by exposure to the environment

after the processing during peeling, slicing and repackaging processes (Sofos and Geornaras, 2010). Thus, food industries have increased attention in developing methods that prevent the growth of, or can kill, this pathogen in RTE meats. In RTE meat products where this bacterium cannot grow, for instance in salami, a limit of 2 log cfu g⁻¹ of *L. monocytogenes* is accepted in both the European Commission (Commission, 2005) and Health Canada (Luber *et al.*, 2011). However, the United States Department of Agriculture (USDA) has adopted a policy of zero-tolerance for pathogens in all RTE meats irrespective of the growth potential.

Not taking the precautions necessary to prevent contamination during the cutting and packing of meat is one of the reasons that processed products become contaminated with microbes, especially with *L. monocytogenes* which has the ability to grow and survive in the food factory environment and therefore there is a greater risk that this bacterium being present at levels likely to result in cross-contamination events if sufficient control measures are not used (Swaminathan and Gerner-Smidt, 2007). In particular, growth of the organism within slicing machines can lead to high levels of contamination of product by directly distributing *L. monocytogenes* on to the surface directly prior to packaging when there are no further Controls at Critical Points (CCPs) used to inactivate this organism (Beresford *et al.*, 2001; Frye *et al.*, 2002; Gormley *et al.*, 2010) and it can then grow during storage even in the presence of high salt and low pH (Gandhi and Chikindas, 2007). In addition, sliced meat products are stored in the refrigerator and as a psychotroph *L. monocytogenes* has the ability to adapt to cold temperatures and carry on growing.

Although outbreaks of listeriosis have been associated with many different types of sliced meats, bologna-style sausage has been recognized as a particular risk for transmission of *Listeria* since there is nothing in the formulation that prevents growth of the organism if it gets into product (Rodrigues *et al.*, 2017).

This has been demonstrated recently by the implication of a bologna-style sausage in the largest recorded outbreak of listeriosis in the world which occurred in South Africa in 2017-early 2018 (WHO, 2018). Therefore when choosing a model system to determine if the liquorice extract can specifically inhibit the growth of *L. monocytogenes* in a real food system, bologna-style sausages were chosen.

For the reasons outlined above, all sectors of the food industry tend to control the growth of organisms through the addition of chemical preservatives. The use of natural antimicrobials as an alternative to traditional, chemical food preservatives has become increasingly popular. Viuda-Martos *et al* (2010) reported that the addition of orange dietary fiber (1 %), rosemary essential oil (0.02 %) and thyme essential oil (0.02 %), combined with specific storage conditions, exhibit very desired effects to preserve the oxidative stability of mortadella, a bologna-type sausage by reducing microbial growth

The ingredients of bologna recipes (Polish Doktorska) used in this study inlude various meat and non-meat ingredients including mechanically separated turkey (44 %), water, turkey skin, soya protein, pork fat (3 %), potato starch, salt, acidity regulator: sodium citrates, stabiliser. triphosphates, antioxidant: sodium erythorbate, spices (contains mustard), spice extracts, flavouring, thickener. carrageenan, flavoure enhancer: monosodium glutamate, preservatives: sodium nitrite.

Objective of this study: The purpose of the research presented in this Chapter was to further investigate the potential of the liquorice extract as a natural alternatives to control *L. monocytogenes* growth in a real food system, using bologna-style sausage as a model to investigate whether it has potential as a food preservative for sliced meats.

4.2 Results

4.2.1 Evaluating liquorice extract activity on bacteria inoculated on to sausage

The aim of this of experiments was to assess the effect of liquorice extract on the growth of *L. monocytogenes* in a model food system. The inoculation of bologna style sausage with *L. monocytogenes* was done using the dipping of sliced sausage samples method according to method described by Lara-Lledó *et al* (2012) with some modifications. Specifically in this study the effect of packaging was not being examined and so the slices were not stored under different conditions after inoculation. For these experiments retail samples of whole bologna style sausage was purchased from a local supermarket (see Section 2.16)

To prepare the samples *L. monocytogenes* 10403S was grown in BHI broth and incubated at 37 °C in shaking incubator 150 rpm for 18 h. The cells were then collected by centrifugation (2,500 x *g* for 10 min at room temperature) and resuspended in 0.1 % (w/v) BPW to a density of 1 x 10^9 cfu ml⁻¹ and then diluted further diluted to the required inoculum level. BPW was used as a neutral buffer that would not support the growth of the bacteria but would allow them to recover from any damage incurred during the preparation of the inoculum.

Initially a high inoculum of *L. monocytogenes* was used so the cells were further diluted 1:10 in BPW to reach 10^8 cfu ml⁻¹. The bologna-style sausage was sliced (~3 mm) and inoculated by dipping for 30 s in to samples of the inoculum that had been supplemented with either 50 µg ml⁻¹ liquorice extract or 1 % (v/v) ethanol (Figure 4.1) and then the slices were allowed to dry on a petri dish lid in a Class II microbiological safety cabinet (MSC) for 30 min at room temperature to remove excess liquid for inoculum attachment. It was expected that this method would achieve an inoculation of approximately 10^6 cells onto the surface of the sausage slices.

Figure 4.1. Method used to inoculate *L. monocytogenes* onto slices of bologna-style sausage



A schematic of the method used to apply liquorice extract and *L. monocytogenes* to the bologna sausages slices to study the effect of the extract on the growth during storage. The cells were diluted into BPW at the appropriate concentration and then the suspension supplemented with either ethanol as the solvent control or the liquorice extract dissolved in ethanol.

To determine if the extract could inhibit the growth of *Listeria* on the surface of the sausage, the inoculated slices were stored at 6 °C and samples (n = 3) were removed for sampling at day 0 (day slices were prepared) and thereafter at day 5, 10 and 15 of incubation. To sample the slices, each one was transferred into an individual sterile stomacher bags (177 mm x 305 mm) and mixed with 180 ml 0·1 % (w/v) BPW and stomached for 60 s at 300 RPM to release the *Listeria* into the liquid phase for sampling. This process was repeated on each day of sampling. After stomaching, appropriate serial dilutions were prepared with MRD and then the level of *Listeria* in the samples was determined by plating appropriate dilutions on PALCAM agar plates using Miles and Misra technique as a described in (Section 2.12) using 5 replicate 20 µl samples per dilution (total sample volume = 100 µl). Selective enumeration of *L. monocytogenes* was required since the

product being used was a commercial sausage product and was not sterile, therefore selective agar was used to specifically enumerate this organism. After allowing the plates to dry, colonies were enumerated after incubation at 37 °C for 24 h (Section 2.12). The experiment performed in triplicate and the values were expressed as a mean value with standard deviations (M \pm SD).

Figure 4.2. Bologna-style sausage inoculated with high numbers of *Listeria monocytogenes*



Cell suspensions (1 ml) of *L. monocytogenes* was diluted 1: 10 in buffered peptone water (BPW) in two flasks, one flask contains a concentration of 50 μ g ml⁻¹ liquorice extract and second flask contains a 1 % (v/v) ethanol solvent served as a control, slices inoculated by dipping in both solutions, then the slices storage at 6 °C at different time (0, 5, 10 and 15 days). Three individual samples on each sampling day and for each treatment were used. Data are given as means ± standard deviations, each performed in triplicates (n=3).

Since outgrowth of the cells was seen after a period of incubation (Figure 4.2), it was hypothesized that the high cell number used in this experiment might result in breakdown of the inhibitory substances and allowing a delayed growth of cells after some time. This would reflect the findings reported by other authors that inoculum size is important when evaluating antimicrobials. For instance Silva-Angulo *et al.* (2015) showed that the lag phase of *L. monocytogenes* exposed to citral was shortened when high cell inoculum was used. Therefore, as some inhibition of growth was seen using the high cell inoculum, the experiment was redesigned using a lower cell concentration (10⁶ cfu ml⁻¹) in the inoculum.

In this case the initial number of cells detected on the slices of sausage was lower at approximately 10^4 (see Table 4.1). In the absence of the liquorice extract, an increase in *L. monocytogenes* of $1 \log_{10}$ was seen in the control samples during incubation indicating that *Listeria* was able to grow under the experimental conditions used. After day 5, differences in *L. monocytogenes* levels were observed between the control and the sample treated with 50 µg ml⁻¹ liquorice extract (Figure 4.3). From day 5 to day 15 the control exhibited the highest level of growth (p < 0.0001) compared to the treatment. By day 10 and 15 bacterial populations in the sample treated with 50 µg ml⁻¹ and 1.1 log₁₀ cfu ml⁻¹, respectively (Table 4.1, Figure 4.3).

Table 4.1. The number of bacteria emerging on the slices of sausage inoculated with lower numbers of *L. monocytogenes*, counted after 24 h incubation, and converted to \log_{10} cfu ml⁻¹

Time	cfu ml ⁻¹ (mean ± SD)		
(Day)	L. monocytogenes with 1 %	L. monocytogenes with 50 µg	
	(v/v) Ethanol solvent	ml ⁻¹ of liquorice extract	
0	4.29 ± 0.15	4.18 ± 0.16	
5	4.17 ± 0.13	4.21 ± 0.12	
10	4.59 ± 0.14	3.63 ± 0.12	
15	4.71 ± 0.16	3.61 ± 0.23	

Cell suspensions (1 ml) of *L. monocytogenes* were diluted 1: 100 in buffered peptone water (BPW) in two flasks, one flask contains a concentration of 50 μ g ml⁻¹ liquorice extract and second flask contains a 1 % (v/v) ethanol solvent to yield 10⁶ cfu ml⁻¹, then slices of bologna inoculated by dipping in both solutions, one served as a control 1 % (v/v) and other as treatment with liquorice extract, then the slices storage at 6 °C at different time (0, 5, 10 and 15 days). Three individual samples on each sampling day and for each treatment were used. Data are given as means \pm standard deviations, each performed in triplicates (n=3).

Figure 4.3 illustrates the growth of *L. monocytogenes* and the results showed that the extract was able to completely inhibit the growth of *Listeria* throughout the cold storage, whereas an increase in 1 log₁₀ was seen in the control samples. By adding liquorice extract the number of *L. monocytogenes* on bologna slices initially surface inoculated to contain 10^4 *L. monocytogenes* cfu ml⁻¹ decreased 0.6 log₁₀ after 15 day of storage in the presence of 50 µg ml⁻¹ of liquorice extract, compared with a 0.42 log₁₀ increase for the slices without liquorice extract (Table 4.1). Analysis of the results indicated that by day 10 of inoculation there was a clear statistical difference (p < 0.0001) between the

number of cells detected in the two samples (see Figure 4.3). Interestingly this result in a model food system differed from the early results as there was a reduction in viable count whereas on long term exposure in liquid culture experiments (see Section 3.2.5) the extract was found to be bacteriostatic rather than bactericidal.

Figure 4.3. Growth of *Listeria monocytogenes* on bologna-style sausage at 6°C



Bologna style-sausage was inoculated with *L. monocytogenes* with 50 µg ml⁻¹ liquorice extract prepared in 0.1 % BPW [\blacksquare] and with *L. monocytogenes* with 1 % (v/v) ethanol (solvent control) prepared in 0.1 % BPW [\bullet]. Two way-ANOVA was used to evaluate the treatment by using Sidak's multiple comparisons test to determine the significant differences among treatments by day with the level of significance (P value) set at p < 0.0001. Growth of *L. monocytogenes* on bologna incubated for 15 day at 6 °C, **** indicates the time when growth was significantly higher than inoculation levels.

The results gained here showed that in addition to being effective at controlling the growth of *Listeria* in liquid cultures, it could also be used to control growth in a real food system. It was interesting to note that when the bacterial cells were exposed to the extract in this system, some loss of viability was seen over time. As this was a commercial product, it is possible that the antimicrobials used to produce the product were creating an additional stress on the *Listeria* cells. Bologna-style sausages are non-fermented emulsions and to extend the shelf life of these low acid food products, preservatives are often added. In particular potassium and sodium nitrite are added which, are popular in this type of product because, in addition to being antimicrobial nitrites (NO₂⁻) combine with myoglobins present in the meat product to form nitrosylmyoglobins, which are converted to nitrosylhemochromes during cooking and provides the characteristic pink color in these ready-to-eat meat products.

4.3 Discussion

We have previously shown that a liquorice plant extract, prepared from a waste product from the preparation of edible liquorice was able to inhibit the growth of some Gram-positive pathogens and there was evidence that the mode of action of liquorice extract is based on its ability to both disrupt cell wall formation and effect membrane integrity, leading to inhibition of growth for those pathogenic bacteria including *Listeria* strains and resulting in lower levels of metabolic activity.

In vitro, the extract was applied on bologna sausage as a model real food to show whether it could be used to control the growth of *Listeria*. In the initial experiments using a high cell density cells, although was seen inhibition, after extended incubation outgrowth was seen (Figure 4.2). This is consistent with results reported by others using organic food preservatives where the substance is bacteriostatic and can potentially be metabolised by the cells in the culture. It

is interesting to note that even at high cell number (approximately 10⁷ cfu ml⁻¹) in the liquid cultures, even after extended incubation, there was no evidence of metabolism and outgrowth. This suggests that in a real food system the extract may be less available due to partitioning into the different food components, such as fats, which would effectively reduce the concentration of the extract and allow outgrowth, as seen with the lower concentration of the extract used in the *in vitro* experiments.

To try and carry out a more realistic evaluation of the potential application of the extract, the method was modified to reduce the initial inoculum but still provide enough cells to allow the *Listeria* to be enumerated without the need for enrichment procedures. When a low inoculum (10⁶) was used, good inhibition of *Listeria* which confirmed the hypothesis that the high cell number was leading to metabolism of the extract, resulting in outgrowth of the cells after extended incubation at low temperature, and according to a study by Cos *et al.* (2006), a high inoculum size (e.g. 10⁷ cfu ml⁻¹) will increase the chances for false-negatives, while a too low inoculum size (e.g. 10² cfu ml⁻¹) will create many false-positives as cell numbers will fall below the limit of detection of plating methods. Another study conducted in *E. coli* by Somolinos *et al* (2010) demonstrated that a large initial inoculum resulted in a smaller amount of inactivation by citral, however it must be noted that such very high levels of *L. monocytogenes* are very unlikely to present in retail RTE foods and therefore the outgrowth seen at very high inoculum levels is less of a concern in real food systems.

In the previous Chapters the effect of the extract was shown to be bacteriostatic, and once the compound was removed the bacteria started to grow again. However in the low dose experiment the number of *Listeria* cells declined overtime. As known in the food industry, salt and nitrates are commonly used as a food preservatives and an antimicrobial agent to inhibit the growth of bacteria in ready-to-eat (RTE) bologna styles sausages contributing to an extended shelf

life of these meat products (Desmond, 2006; Ruusunen and Puolanne, 2005). Salt affects the bacteria by causing damage on the maintenance of osmotic balance between the cytoplasmic and intracellular environments (Burgess *et al.*, 2016), and this creates stress on the cell membrane and cell metabolism. Hence during exposure to the extract in the real food system this may have contributed to the increase in the rate of cell death seen when tested in the bologna-style sausages.

The efficacy of naturally derived antimicrobials on the growth of *L*. monocytogenes has been reported in several other studies, such as that as reported by Lara-Lledó *et al.* (2012) the inhibition of *L. monocytogenes* on bologna sausages coated by an antimicrobial film containing mustard extract or sinigrin and stored at 4 °C for 72 day, and showed that on meat packed with pure sinigrin film, *L. monocytogenes* was not inhibited. In contrast, at 17 d storage, yellow mustard extract film caused a significant delay ($p \le 0.05$) in the growth of *L.* monocytogenes and the viability of *L. monocytogenes* reduced in bologna sausages during the storage period. In addition, no viable *L. monocytogenes* was recovered after enrichment at 52 and 70 d, it is highly likely that the inhibitory effects of oriental mustard observed were bactericidal.

In study conducted by Naas *et al.* (2013) reported that the effect of nisin (produce by *Lactococcus lactis* subsp. *lactis* during fermentation) when combined with different types of packaging used for turkey bologna slices was enhanced and reducing the initial population of *Listeria monocytogenes* by 1.5 to 2 log. Thus, when extra hurdles are applied to control growth of *L. monocytogenes* in RTE meat a synergistic effect is seen that helps to improve the safety in RTE meats during refrigerated storage. Thus, the use of this kind of natural compounds can have effective antimicrobial activity and may eliminate *L. monocytogenes* from food products if used as a food preservative in sliced meats.

L. monocytogenes can be present in a wide variety of fresh produce, thus it is important to minimize the risk of the growth of this pathogen in order to

promote food safety and consumer health protection. Previous studies of chemical compounds isolated from liquorice plants had revealed antitumor, antibacterial, immunomodulatory and antioxidant activities (Karahan *et al.*, 2016). Now having established that the liquorice extract could be used in a real food system. However, before using the extract as food preservative we need to confirm the safety and toxicity of the extract as it is known that not all compounds produced by plants are safe. Therefore, the next step required was to check the safety of the product by carrying out mutagenicity and toxicity tests.

4.4 Conclusion

In this research, it was found that liquorice extract inhibited the growth of food-borne pathogenic bacteria, such as *L. monocytogenes*. Therefore this novel antimicrobial extract could be of value to inhibit the growth of *L. monocytogenes* and other Gram-positive spoilage organisms when applied to the surface of sliced deli meats and hence could be of use to improve the safety of sliced deli meats. Thus, application of liquorice extract at 50 µg ml⁻¹ was confirmed to be an effective way to control *L. monocytogenes* growth on the slices of bologna sausages.

CHAPTER 5

EVALUATION OF THE SAFETY AND TOXICITY OF LIQUORICE EXTRACT

Abstract

To confirm that the liquorice extract is safe for application as a food preservative, the potential cytotoxicity and mutagenicity of the liquorice extract was studied *in vitro* using Ames and MTT tests. In the Ames test, any mutagenic effects were determined using a concentration of 50 µg ml⁻¹ of liquorice extract and the ability of extract to induce reverse mutations was evaluated. Using *Salmonella* Typhimurium strains TA100, TA102 and TA1535, the results showed that there were no increases in the number of revertant colonies at this concentration indicating that no mutagenic activity was detected. In contrast, the results showed that using positive controls there were a 6-7-63 fold increase in the number of revertant colonies et the number of revertant colonies that there were this indicates that there is no evidence that the extract is mutagenic.

A second set of experiments was carried out to investigate whether the liquorice extract had any cytotoxic effects. Using yeast cells as a simple model eukaryotic cell type, it was found that yeast growth was not affected by liquorice extract. Moving on to a human cell line model, MTT (3-(4, 5 dimetylthiazol-2-yl)-2, 5-diphenyltetrazolium bromide) assays were used to detect any cytotoxicity. In this case it was found that the liquorice extract did have a cytotoxic effect on Caco-2 cell lines after 24 h exposure to liquorice extract at a concentration of 50 µg ml⁻¹. This was in contrast to the control sample as well as samples treated with 12.5 µg ml⁻¹ of liquorice extract which had no cytotoxic activity on the Caco-2 cells. However the level of cytotoxicity detected was minimal and was considered unlikely to cause damage to human gut cells if consumed by humans rather than being directly applied to colonic cancer cell culture samples. Therefore these preliminary safety studies indicated that this extract could be taken forward as a candidate novel food preservative, and there is also some evidence that the material could be used as a therapeutic agent.

5.1 Introduction

As reviewed in Chapter 1, not all compounds-derived from plants are safe, and several studies indicated that there are some compounds in plants that have toxic and carcinogenic effects which has led to an emphasis on studying the genetic risks of plant compounds (Boada et al., 2016). Accordingly, the use of medicinal plants, crude plant extracts and isolated compounds for food preservation and therapeutic purposes require an evaluation of their cytotoxic and mutagenic potential (Cavalcanti et al., 2006). Plant extracts exhibiting explicit mutagenic properties are considered as unsafe since compounds which exhibit a mutagenic activity could possess carcinogenic properties and thus require further investigation before continuing to be used for use in human products (Verschaeve and Van Staden, 2008). Accordingly in vitro toxicity testing has acquired significant role as a tool for the assessment of the safety of different drugs and chemicals. In this respect, it is important to note that many researchers have reported that various natural compounds isolated from different species of the liquorice plant (Glycyrrhiza spp.) belonging to the Fabaceae family possess cytotoxic activities against multiple cancer cell lines (Fukuchi et al., 2016; Lee et al., 2013; Sheela et al., 2006; Bassyouni et al., 2012). Moreover, liquorice root has been identified by the National Cancer Institute within a group of herbs which possess cancer-preventive properties (Caragay, 1992). This family represents a group known to have high biodiversity and also for the production of an abundance of in secondary metabolites (Wink, 2013), and because of this has been considered especially important for human health in the form of dietary supplements or as therapeutic treatments (Dixon and Sumner, 2003). However the investigation of liquorice extracts for their potential applications to contain other bioactivities is still important.

Mutagenicity bioassays have been confirmed as a significant tool in environmental monitoring and assessment of pollution, with more than 200 short-

term bioassays were utilized through using micro-organisms, insects or plants which have been developed and applied to assist in the identification of factors that pose genotoxic risks (Ansari and Malik, 2009). Among them, the Ames test was specifically designed to detect if a chemical is a mutagen by detecting its ability to introduce changes into bacterial DNA. This test is able to detect different mutagens which affect single nucleic acid bases by either altering or deleting them or those chemicals that cause insertion of bases into a DNA sequence, and any such changes in eukaryotic cells may result in the formation of cancer cells (Ames et al., 1975). Ames employs particular strains of the Salmonella Typhimurium were developed by Bruce Ames and form the basis of the "reverse" mutation assays as tools to reveal specific types of mutation (Mortelmans and Zeiger, 2000; Ames, 1971). The Ames strains carry defined mutations in the histidine biosynthetic operon, thus they need histidine for growth and the strains will not normally grow unless the histidine is supplied in growth medium. Each of the different mutations is designed to be reacting to mutagens that act via different mechanisms, and therefore they are susceptible to a broad variety of substances. The basis of the reverse mutation assay is that in the presence of a test product containing a mutagenic chemical compound, a second mutation will be caused in the faulty genes that will convert it back to the functional state. This allows the bacteria to growing in a medium that contains limiting amounts of histidine to form colonies, with the number of colonies formed being proportional to the mutagenic potential of the chemical (Mortelmans and Zeiger, 2000). Over time, the Ames test has become recognized by many governmental and non-governmental organizations as a standard screening test (it has been adopted as an ISO test; ISO, 2012) and is widely used as an initial screen since the Ames test is an inexpensive and flexible method.

In addition to determining whether the compounds are mutagenic, it is also important to determine whether chemicals have any cytotoxic effects on human

cells since bacterial cells have a different structure and cell composition and chemicals may interact with different target sites eukaryotic cells that do not exist in bacterial cells. Under physiological conditions intestinal cells have a high exposure to dietary compounds and for this reason Caco-2 cells are often chosen as a model system to assess cytotoxicity. This cell line was derived from a human colon carcinoma and when grown in culture, it subjects enterocytic differentiation in order to form a monolayer and its functional and structure much like to the human small intestine epithelium, it is known that the responses of live cells are complicated by the membrane permeability (Asano *et al.*, 2003; Baker *et al.*, 1995). In addition a study conducted by Asano *et al.* (2003) has shown that Caco-2 cells show good absorption of some flavonoids from liquorice and therefore this is an appropriate test system to use for assessing this new liquorice extract.

The overall aim of this study was to carry out a preliminary safety assessment of the liquorice extract using *in vitro* tests to confirm that it is safe for subsequent application as a food preservative. The specific objectives were to assess the cytotoxicity and mutagenicity effects of liquorice extract *in vitro* to confirm that it is safe for subsequent application using the Ames test (Section 2.17) and the MTT assay (Section 2.18.3).

5.2 Results

5.2.1 Evaluating the mutagenicity and toxicity of liquorice extract

The safety of liquorice extract *in vitro* was investigated using the Ames test as the indication of the mutagenicity of the extract. The *Salmonella* Typhimurium strains used were TA100, TA102 and TA1535 (Table 5.1). The *rfa* mutation present in all of these strains shortens the LPS molecules on the outside of the cell to help facilitate the uptake of the larger test compounds by the cells. A deletion mutation in the region of the chromosome between *uvrB* and *bio* genes is also present in all strains except TA102 and this region includes the *uvrB* gene which is required for

the accurate excision DNA repair mechanism. This mutation then allows more DNA lesions to be repaired via the error-prone DNA repair pathway and the deletion in this region also make the cells dependent on the presence of biotin in the media (see review by Tejs, 2008). The presence of the plasmid pKM101 in TA100 and TA98 results in enhanced chemical and UV-induced mutagenesis due to the over expression of the recombination DNA repair pathway that is encoded by this plasmid and also confers ampicillin resistance on the cells. Plasmid pAQ1 which contains the mutation *hisG*428 was introduced into strain TA102 to increase the number of copies of the target mutation, and reversion of any one of these will lead to a reversion to wild type, hence increasing the sensitivity to mutagenic agents. Finally this strain also retains the *uvrB* gene which makes it more sensitive to DNA cross-linking agents as it is able to use the excision DNA repair mechanism induced by these types of chemicals.

Given the phenotypes described above, in the case where the extract is mutagenic, the strains will undergo mutations that revert the mutations in the histidine biosynthetic operon to wild-type (histidine-independence) when exposed in the presence of chemical. If there is no evidence of mutagenic potential this is the first evidence that the extract was safe and could be used as a food preservative. However before using these strains it is also important to carry out tests to confirm that the isolate still retains the expected plasmids which can be easily lost from these strains.

Table 5.1. Genotype of the most common	y used Salmonella test strains*
--	---------------------------------

Mutation (strain)	(bio chlD uvrB gal)	LPS defect	Plasmid
TA1535 hisG46	Deletion mutation	rfa	No plasmid
TA100 hisG46	Deletion mutation	rfa	pKM101
TA98 hisD3052	Deletion mutation	rfa	pKM101
TA1537 hisC3076	Deletion mutation	rfa	No plasmid
TA102 hisG428	Wild type	rfa	pKM101, pAQ1

Histidine operon mutations

(*): This table reproduced from Mortelmans and Zeiger (2000)

5.2.1.1 Confirmation of the presence of plasmids in Salmonella strains

The presence of plasmid pKM101 was tested by streaking a loop-full of culture across a BHI agar plate supplemented with 24 μ g ml⁻¹ ampicillin and incubating the samples for 24 h at 37 °C. The result (growth) confirmed that the strains that were expected to contain these plasmids were able to grow and form colonies on these plates and this indicative of plasmid presence (Table 5.2). Similarly the presence of plasmid pAQ1 was tested by streaking a loop-full of culture across a BHI agar plate supplemented with 24 μ g ml⁻¹ ampicillin and 2 μ g ml⁻¹ of tetracycline. Plates were incubated for 24 h at 37 °C. The results confirmed that the strain that were expected to contain both plasmids was able to grow and form colonies on these plates and this indicative of plasmid presence in (Table 5.2).

Characteristics	Type of genetic	Result	
Characteristics	change		
Tetracycline registance	Presence of plasmid	Growth in the presence of	
retracycline resistance	pAQ1	tetracycline	
Americillin vesietenes	Presence of plasmid	Growth in the presence of	
Ampicillin resistance	pKM101	ampicillin	

Table 5.2. The genotype of the Salmonella Typhimurium TA100 & TA102

Salmonella strains (TA100 &TA102) were grown in the plates supplemented with antibiotics to confirm the presence of plasmid

5.2.1.2 Ames test

For the current study, to assess the safety of liquorice extract, a preliminary mutagenic screen was carried out using the Ames test using the *Salmonella* Typhimurium strains TA102, TA100 and TA1535 (see Tables 2.2, 2.3 and 5.1 for details). In these experiments trace amounts of histidine and biotin were added to a molten agar to allow all the *Salmonella* strains grow on the plate of minimum agar medium for a defined number of divisions. Thus, in case there is no chemical compound cause mutagenicity, the *Salmonella* will stop dividing once the histidine and biotin are exhausted. Otherwise, the *Salmonella* will continue division which indicates a DNA mutation had occurred to reverse the histine-dependant growth phenotype. Therefore, after exposure to the chemicals, the number of cells that are able to grown into colonies are counted, and this number reflects the number of *S*. Typhimurium bacteria that undergo a reverse mutation from His⁻ to His⁺.

Cultures of the strains were prepared as described in Section 2.4.2 and then 0.1 ml of bacterial suspension was added to 2 ml of top agar in a bijou bottle, along with the test substances. The liquorice extract was added to a concentration

of 50 µg ml⁻¹ and as a control 1 % (v/v) ethanol was added to separate agar samples. The mixture was then poured over the surface of minimal glucose (MG) agar plate (Section 2.17 and Figure 5.1). As a positive control, sterile filter paper disc (6 mm) were placed in the centre of an MG agar plate that had been seeded with the Ames strain using the soft agar method described and then 10 µl of the test mutagenic solution was dispensed onto the surface of the disc (see Table 2.6 for details). The standard mutagens used as positive controls in experiments were sodium azide for TA100 and TA1535, and Mitomycin C for TA102. The plates were incubated at 37 °C for 48 h and the His⁺ revertant colonies were counted manually. All experiments were carried out in triplicate. After incubation the colonies were counted on each plate, the average number of colonies on the three plates was recorded and the results expressed as the number of revertant colonies per plate and compared with the negative control plates. Toxicity was detected as a reduction in the number of histidine revertants (His⁺) or as an alteration in the auxotrophic background (i.e., background lawn).

Figure 5.1. Method for preparing agar plates for Ames test



2 ml molten top agar was combined with each of the cultures of bacteria \pm 50 µg ml⁻¹ of liquorice extract and then poured onto the surface of a minimal glucose (MG) agar plate (Adapted from Mortelmans and Zeiger, 2000).

The results of the reverse mutation test using three bacterial strains are summarized in Figure 5.2. Data showed no significant increase in the number of revertant colonies as a result of liquorice treatment on the three strains of *S*. Typhimurium (TA100, TA102 and TA1535). In contrast the number of revertant colonies in positive controls increased significantly (p < 0.0001) compared to the negative control while the number of colonies recorded for the 50 µg ml⁻¹ liquorice extract treatment and the negative control were not statistically different (P > 0.05) (Figure 5.2). Therefore the treatment with 50 µg ml⁻¹ of liquorice extract was not considered to be mutagenic compared to the positive controls of both sodium azide & mitomycin C (Figure 5.2, a & b). Interestingly there was a slightly higher increase in the number of revertants when the TA100 and TA102 cells were exposed to ethanol in the negative control samples. Although generally not considered to be mutagenic in the Ames test (Vedmaurthy *et al.*, 2012), it is known

that when converted to acetaldehyde ethanol can cause both DNA cross linking damage and induce transition/transversion mutations in DNA, which are the types of mutations detected by TA102 and TA100, respectively. Although 1 % ethanol was used as a control, this is higher than the level of ethanol present in the sample containing 50 μ g ml⁻¹ of the extract, since as described in Chapter 3, this was the maximum amount of ethanol present in any of the different concentrations tested and therefore was used as a standard throughout these experiments. The fact that the increase in mutation rate was only 2.7-fold for TA102 and 1.4-fold for TA100 would mean that the dilution of the ethanol in the liquorice extract would take the concentration of this below a threshold where an effect could be detected. Generally only eukaryotic cells are able to convert ethanol to acetaldehyde efficiently through the action of alcohol dehydrogenase, which is believed to be the underlying mechanism for the carcinogenic effect of alcohol consumption in humans (Mizumoto et al., 2017), but the pattern of results seen here would be consistent with the presence of low levels of acetaldehyde in these samples due to an unknown source. Despite this anomaly, the results of the reverse mutation test using three bacterial strains proved that dose of 50 μ g ml⁻¹ of liquorice extract was not mutagenic to *S*. Typhimurium.

Figure 5.2. Effect of treatment of liquorice extract, sodium azide and Mitomycin C treatment on revertant of histidine auxotrophy of the Ames strain S. Typhimurium TA100, TA102 and TA1535

TA1535



50 µg ml⁻¹ of liquorice extract

(a) Salmonella strains TA102, TA100 and TA1535 were treated with 50 μ g ml⁻¹ and sodium azide (NaN₃)or Mitomycin C ($C_{15}H_{18}N_4O_5$) as a positive control and 1 % (v/v) ethanol solvent as a negative control, after 48 h incubation time the number of revertant colonies were counted, (b) Values presented are averages of three replicates of each bacterial strain. The results indicated that positive control high significance with ****p < 0.0001 compared to negative control and 50 µg ml^{-1} of treatment and the differences were not statistically significant (P > 0.05) between 50 µg ml⁻¹ of liquorice extract and negative control. Positive control (sodium azide) for TA100, TA1535, positive control (Mitomycin C) for TA102, negative control (1 % (v/v) ethanol solvent with culture).

5.2.2 Evaluating cytotoxicity using a cell viability with MTT assay

In previous Chapters, we confirmed that the extract significantly inhibited the growth of Gram-positive bacteria. *In vitro* experiments were performed using Caco-2 cells for the cytotoxicity to confirm that the liquorice extract has the activity as anticancer through its effect on the viability of Caco-2 cells. The viability of the Caco-2 cells was determined by using the MTT assay, in which the reagent (*3*-(*4*,*5*-dimethylthiazol-2-yl)-2,*5*-diphenyltetrazolium bromide) tetrazolium is used as a monitor of cell viability. This test is based on the cleavage of a tetrazolium salt by mitochondrial dehydrogenases and this occurs in viable cells (Khairunnisa and Karthik, 2014). In this test soluble MTT bromide is converted by mitochondrial enzyme activity of viable cells into an insoluble coloured formazan dye which can be measured by resolubilize the formazan and the concentration of the dye produced determined spectrophotometrically at 570nm (Figure 5.3).

Figure 5.3. Structures of MTT and coloured formazan product



Viable cells converted MTT bromide into a purple coloured formazan product which is insoluble, the absorption can be measured at 570 nm using a spectrophotometer after cells are solubilized (Adapted from Riss *et al.*, 2013). As previously mentioned Caco-2 cells were selected for this study as they had previously been shown to exhibit an excellent absorption of some flavonoids from liquorice (Asano *et al.*, 2003). To carry out these experiments Caco-2 cells were maintained in DMEM medium as a described in Section 2.18.1. The medium was replaced every two days, and the cells were passaged at 80 to 90 % confluence using trypsin-EDTA solution. Cells were stained with trypan blue, and cell number and viability were determined using a haemocytometer as a described in Section 2.18.2.

The MTT assay was then performed using the protocol described by Riss *et al.* (2016) with a slight modification. Briefly, Caco-2 cells were seeded at a density of 0.3×10^5 cells per well by adding 5 ml of DMEM containing the cells to each well in a 6-well plate. The cells were then treated with two different concentrations of liquorice extract (12.5 and 50 µg ml⁻¹) and as negative controls they were also treated with 1 % (v/v) ethanol and 20 % sodium dodecyl sulfate (SDS) was used as a positive control. A second negative control was prepared by treating the cells with 4 % vinegar. It was decided to use vinegar as a negative control since in a study by Bozorgi *et al.* (2017), vinegar was used to extract Squill [Drimia maritima (L.) Stearn] and the results did not show toxicity against cancer cells including HT-29 and Caco-2 . In addition controls were prepared by adding cells to the wells (*i.e.* the well contained medium only with cells). Triplicate wells were prepared for each individual treatment and samples were kept in the incubator for 24 h at a humidified atmosphere of 95 % air and 5 % CO₂ at 37 °C.

After 24 h, culture medium was removed by aspiration and the cells washed with 5 ml of PBS to remove dead cells and cellular debris. The PBS was then replaced with fresh DMEM medium containing yellow tetrazolium salt and samples were re-incubated for an additional 3 h at 37 °C to allow the tetrazolium salt to be converted to the purple formazan dye. Finally the supernatant was removed, and the residual purple formazan product was solubilized in 2 ml solubilization buffer,

the plate was gently shaken and re-incubated for 15-30 min to dissolve the purple formazan dye crystals formed and to stop the reaction. Finally 1 ml samples were transferred into cuvettes and the absorbance was measured at OD_{570nm}. The percentage cell viability was calculated from triplicate results using the following formula as a described by (Rezk *et al.*, 2015). Absorbance values higher than the control indicated that the cells had retained viability, while values lower than the control indicates that cell death or inhibition of respiration had occurred.

$Cell \, viability \, (\%) = \frac{Absorbance \, of \, treated \, cells}{Absorbance \, of \, control \, cells} \times 100$

The experiment was performed in triplicates and repeated at least three times in independent experiments, and the number of cells was expressed by averaging of triplicate readings in each well (Table 5.3; Figure 5.5) and the results gained at each step of the assay are shown in Figure 5.4.

The results showed that the cell control culture contained very similar levels of viable cells to that detected in the presence of 1 % ethanol (96 % viability retained; Table 5.3; Figure 5.5) and that there was no significant difference (p= 0.15) between the results gained using the solvent control and the liquorice extract at a concentration of 12.5 µg ml⁻¹ indicating that at this concentration no cytotoxic effects were detected. Interestingly the control sample challenged with 4 % vinegar showed no effect on the viability of the Caco-2 cells – in fact there was a slight increase in the cell viability recorded (111 %; Table 5.3). Acetic acid is commonly used as a component of the solvent for the insoluble formazan dye (Berridge *et al.*, 2005), and therefore it is possible that this apparent increase in cell viability just represented a better solubilisation of the dye produced by the cells in these wells rather than a stimulation of cell growth or enzyme activity.



Figure 5.4. Caco-2 cell line with different concentration of liquorice extract and positive and negative control

Cell viability was measured by detecting purple formazan that was metabolized from MTT by mitochondrial dehydrogenases, which are active only in live cells. Panel (a) samples taken at zero time, (b) samples after 24 h incubation, (c) samples after fresh DMEM medium containing yellow tetrazolium salt was added and re-incubated for an additional 3 h at 37 °C and (d) samples after added solubilization solution and incubated for 20 to 30 min at 37 °C.
However when the cells were treated with the liquorice extract at a concentration of 50 μ g ml⁻¹ the cell viability recorded was only 2-fold higher than that for the SDS positive control sample. This result was significantly lower than the negative control samples (p < 0.0001) indicated that there was damage to the Caco-2 cells caused by exposure to this concentration of the extract.

Treatment	Average	% Cell Viability
Control (cells)	1.47	100
Control 1 % (v/v) EtOH	1.40	96
12.5 µg ml ⁻¹ Extract	1.41	96
50 µg ml ⁻¹ Extract	0.33	22
Vinegar (4 %)	1.63	111
SDS (20 %)	0.14	10

Table 5.3. The cell viability MTT Assay

Caco-2 cells treated with and without liquorice extract as well as vinegar and SDS then incubated for 24 h at a humidified atmosphere of 95 % air and 5 % CO₂ at 37 °C. The experiment was performed in triplicates and repeated at least three times in independent experiments, and the number of cells was expressed by averaging of triplicate readings in each well. Caco-2 cells have a good relationship between absorbance and the number of cells was observed (r = 0.9971), and the extract caused more than 50 % growth inhibition which considered as a cytotoxic agent. The findings showed that the reduction in cell viability caused by the liquorice extract was statistically significant (P < 0.0001).

Figure 5.5. Histogram showing results of MTT assays



The cytotoxicity effect of liquorice extract on Caco-2 tumour cells. Cells were treated with and without the extract at two concentrations (12.5 and 50 μ g ml⁻¹), then readings were taken after 24 h using spectrophotometer at OD_{570nm}. Data is given as means ± standard deviations from three independent experiments, each performed in triplicates (n=3). Differences between means were evaluated using one-way ANOVA analysis; and values of p < 0.0001 were considered statistically significant.

5.2.3 Evaluation of toxicity of liquorice extracts on yeast

Since the MTT assay had indicated that there was a cytotoxic effect seen when the higher concentration of the liquorice extract was used in the MTT assay, it was decided to test the effects of the cytotoxicity of the liquorice extract against *Saccharomyces cerevisiae* which was chosen as another eukaryotic cell. The cellular structure and functional of *Saccharomyces cerevisiae* much resemblance to cells of higher-level organisms, *S. cerevisiae* is also increasingly utilized in the toxicological evaluation of chemicals (Kasemets *et al.*, 2009). To confirm if there is any an effect on growth of yeast, two strains of *S. cerevisiae* were used in this study are listed in (Table 2.2) including a lab isolate strain (S288C) and one wild yeast (NCYC 363).

To carry out these experiments a single colony of yeast strains S288C and NCYC 363 were inoculated in yeast extract peptone dextrose (YPD; Section 2.5) broth and incubated at 30 °C for 48 h. Cells were harvested by centrifugation and re-suspended in 20 ml distilled H₂O, this process was repeated twice to ensure all the YPD broth was removed from the culture and then the cells were finally resuspended in 10 ml. To determine the number of viable cells present in the sample, the cell suspension was diluted and then stained with methylene blue (Section 2.20) so that the viable count could be determined microscopically using a Neubaure Haemocytometer (Section 2.18.2)

The number of viable cells in the suspension was then calculated and the cells further diluted in YPD broth to a density of 1×10^6 cfu ml⁻¹. Three flasks were then prepared which had been supplemented with 12.5, 50 µg ml⁻¹ liquorice extract or 1 % (v/v) ethanol solvent as a described in Section 2.20. Samples (200 µl) of these different cultures were transferred into plate 96 well which was then incubated in a microplate reader (TECAN) at 25 °C for 1800 cycles of orbital (30 min) shaking followed by OD_{600nm} measurements for 48 h.

The result are shown in Figure 5.6 indicated that the growth of both strains was the same as the control sample containing (1 % v/v) ethanol at both concentrations of the liquorice extract tested (12.5 and 50 µg ml⁻¹). Although ethanol is known as an antimicrobial, the percentage of ethanol used in these experiments is well below the level known to effect on the growth of yeast as ethanol-producing yeast can tolerate levels of ethanol of up to 8 % in the culture media without affecting growth (Kalathenos and Russell, 2003). Hence in this case, the yeast appeared to be resistant to the antimicrobial effects of the extract seen when testing Gram-positive bacteria as there was no significant difference between the control and the samples treated with the liquorice extract. Hence for this eukaryotic cell type, no toxicity was detected.



Figure 5.6. The growth of yeast in the presence of liquorice extract

The growth curves of *S. cerevisiae* of yeast cultures NCYC 363 and S288c grown at 25 °C for 48 h in a culture with and with liquorice extract at concentrations of $[\bullet]$ 12.5 µg ml⁻¹ and $[\bullet]$ 50 µg ml⁻¹ and the control sample $[\bullet]$ containing 1 % (v/v) ethanol. Growth was monitored by optical density at 600nm. Data shown are given as means ± standard deviation (SD) for 12 replicates for every concentration; SD values are very small and are very close to the data points and therefore are not visible on the graphs.

5.3 Discussion

Current research is focussing on natural molecules and natural products that are produced by plants as novel food preservatives since they can be sourced relatively easily and many are selected for evaluation based on their medicinal use as traditional medicine for many years (Mussarat et al., 2014). However, compounds or extracts with a specific activity at a non-toxic dose need further evaluation to define the toxicity status. Thus, toxicological properties in this study were defined by using three methods, for mutagenicity (Ames test) and toxicity (MTT test and yeast growth inhibition). The Ames test was chosen as a rapid and simple method to perform, and this test is specifically designed to reveal a wide range of chemical substances that can cause genetic damage which turn lead to gene mutation (Mortelmans and Zeiger, 2000). However even in the absence of a mutagen a small number of spontaneous mutations will occur in the high number of bacterial cells present in the test sample leading to the formation of a low number of colonies due to the occurrence of random mutations in the population (Mortelmans and Zeiger, 2000; Wessner et al., 2000; Ames et al., 1973a; Ames et al., 1973b). This was seen in the experiments reported here, but clearly when the strains were exposed to both positive control substances (sodium azide and mitomycin C) the rate of mutations increased dramatically.

The Ames test is considered a convenient tool for quantifying the genotoxicity of composite herbal mixtures that are consumed to improve human health. Although it is generally recognised that identifying the level of genotoxic compounds present in the herbs is dependent on the use of several *Salmonella* strains for the Ames assay (Maron and Ames, 1983). As known, chemicals can cause damage in germ lines leading to fertility problems and causing mutations in future generations, in addition they also are able to induce cancer. Gene mutations can occur with only a single base changes (base-pair substitution mutants), and others contained additions or deletions of one or more bases

(frameshift mutants) (Mortelmans and Zeiger, 2000) as a described in (Table 5.1) and therefore different Ames strains were used to allow these different types of mutations to be detected. However, a positive result does not necessarily indicate that the compound is a carcinogen, the Ames assay just confirms whether or not a compound is mutagenic (Zeiger, 2001).

In the experiments described in this Chapter, no statistically significant difference (P > 0.05) was observed between the effects of the liquorice extract and ethanol solvent, and indeed the ethanol alone seemed to have a more mutagenic effect than the extract showing the importance of including controls in these experiments to rule out the possibility that the solvents used are adversely affecting the test cells. Importantly there has been no reported toxicity of liquorice in the previous studies and therefore the results gained are in agreement with the published literature. For instance Moosavi et al. (2013) reported that Abrus precatorius seeds belonged to the family Fabaceae - and known commonly as Indian liquorice and contains glycyrrhizin and are used for medicinal purposes were not mutagenic to the TA100 strain of S. Typhimurium when tested in the suggested dose range of 0.2-0.5 mg ml⁻¹. However this study would have been more convincing if a range of the different Ames strains were used to cover all of the possible mechanisms of DNA damage that can be detected. Based on the results obtained in this study in Figure 5.2 it is confirmed that liquorice extract is not mutagenic in the Salmonella Typhimurium reverse mutation assay (Ames test).

Fruits and vegetables considered as a plant-based diet which contains a large number of molecules that have the chemopreventive potential to fight against cancer evolution. It has been discovered that there is a strong correlation between diet and cancer; where the dietary constituents that inhibit mutagenesis and/or carcinogenesis play an important role in the protection from cancer such

as vitamins. Previous studies have documented the anti-cancer properties of plants due to their high content in polyphenols (Fresco *et al.*, 2010). Other compounds including carotenoids, flavanoids, isoflavones and catechins are known to reduce the risk from several forms of human cancers (Kris-Etherton *et al.*, 2002; Gagandeep *et al.*, 2005). In a study conducted by Xuesheng *et al.* (2009), it was reported that liquorice could be used to minimize the production of toxic intermediates when included in some mixtures containing poisonous herbs, but this would probably be related to the well-known antioxidant effect of liquorice extracts (Kim *et al.*, 2012; Tohma and Gulçin, 2010; Martins *et al.*, 2015).

The cytotoxicity of liquorice extract was determined using MTT Assay against Caco-2 cell line. The MTT colorimetric assay has been used for assessing the reduction in the cell culture viability in many studies of plant extracts (Wang *et al.*, 2015). Anti-cancer activity for plants has also been evaluated by many researchers using methods such as MTT assay (Carvalho *et al.*, 2010; Cheshomi *et al.*, 2016). Thus, liquorice extract was tested in this study using cultures of human Caco-2 cells, although the results are often equally interpreted as evidence of anticancer activity. Liquorice has already been studied extensively as a potential anticancer or cancer chemopreventive efficacy.

The results of this study showed that the extract at a concentration of 12.5 μ g ml⁻¹, the cells have higher cell viability of more than 90 % indicated that the compound at this concentration was non-toxic to the cells. However, after 24 h exposure, the cell numbers were significantly reduced (p < 0.0001) at a concentration of 50 μ g ml⁻¹ (Table 5.3). Thus, the results of this study showed that the extract exhibited cytotoxic effects toward cancer cells at a concentration of 50 μ g ml⁻¹ by decreasing cell viability after 24 h incubation time. In liquorice, some of phytochemicals compounds have been reported to possess antineoplastic activities against different cancer cell lines. Several reports have confirmed that

various extracts of liquorice (*Glycyrrhiza* spp.) root exerted anticancer effects. The National Cancer Institute has been identified that liquorice root possesses cancerpreventive properties (Fiore *et al.*, 2005; Fiore *et al.*, 2008; Wang and Nixon, 2001). It has been used among patients with prostate cancer and it has been proposed that liquorice can be used efficiently for the cancer therapy without causing severe side-effects to the healthy body cells. However the preparation of herbal formulations to fight the cancerous cells without affecting healthy cells of the body requires a thorough understanding of the complicated synergistic interaction that can occur between the different constituents of anticancer herbs (Larkin, 1983; Saxe, 1987).

Licochalcone A, the main chalcone in liquorice, which possesses many potential biological activities including antitumor activities (Fukuchi *et al.*, 2016; Lee *et al.*, 2013; Fu *et al.*, 2004). It was found that chalcones, such as licochalcone E isolated from the roots of *G. inflata*, exhibited the highest potent cytotoxic effect compared with the known antitumor agents such as licochalcone A and isoliquiritigenin (Yoon *et al.*, 2005). Cytotoxicity of the active constituents isolated from *Glycyrrhiza uralensis* was evaluated against several human cancer cell lines including HepG2 (liver cancer), SW480 (colorectal cancer), A549 (lung cancer), and MCF7 (breast cancer) cells. Licoricidin was the most potent compound among the isoflavans tested, and some compounds showed apparent selectivity among different cell lines, and the liquorice ethyl acetate extract showed more cytotoxic effects between 76–99 % against all cell lines at a concentration of 25 µg ml⁻¹ (Ji *et al.*, 2016). The results of these studies did not definitively conclude whether or not the liquorice can be used as a therapeutic plant, especially for cancer protective.

When considering cytotoxic effects, our results are in accordance with those reported by Badr *et al*. (2013) where the crude methanolic extract of liquorice has

shown cytotoxic activities against intestinal carcinoma cell line (Caco-2) and prostate carcinoma cell line (PC-3) with different concentration of liquorice extract (0, 12.5, 25, 50 and 100 μ g ml⁻¹). In agreement with our results earlier studies conducted by Rao et al. (2014) reported that among of different plant extracts liquorice (Glycyrrhiza glabra) extract were evaluated for their cytotoxicity activity, different concentrations of plant extracts (50, 100 and 150 μ g ml⁻¹) were added to cultured HepG2 cells and incubated for 48 h, it was observed that all plant extracts were cytotoxic to the liver cancer cell line at all the tested concentrations by using MTT assay, it was noticed that the percentage of viable cells decreased as the concentration of extracts increased. On the other hand, when the same extracts were added to normal lymphocytes and examined after 48 h for cell viability by MTT assay, it was observed that the percentage viability was approaching to 100 in most of all the treatments and this is proof that liquorice is entirely safe for human consumption. Therefore although some evidence of cytotoxicity was detected at the higher concentration, and this may be undesirable in a food preservative, it could also be taken as yet more evidence of the presence of anticancer agents in these liquorice extracts. Hence in general, and in agreement with our results, the results of other published studies have concluded that extracts from *Glycyrrhiza* species are not toxic or genotoxic *in vitro* but did exhibited cytotoxicity of the against different cancer cell lines.

5.4 Conclusion

The fact that liquorice extract did not affect yeast cells and the Ames test revealed no mutagenic potential is very significant from a food safety standpoint should these antimicrobials be leveraged as bio-preservatives.

According to the results presented in this study and also in comparison with similar studies, it can be concluded that the liquorice extract has anticancer effects

for colon cancer treatment. However, further research is needed to explore effective compounds that present in the waste material of liquorice extract.

Set against this is the cytotoxicity observed when high concentrations were used, and this is in line with previous reports that liquorice plant helps to reduce the proliferation of tumour cells, such as the Caco-2 cell line. Studying the effects of these extracts on normal cell line is necessary to ascertain selectivity of cytotoxic activity to cancer cells and to determine if this extract could be safely used as a novel antimicrobial.

CHAPTER 6

GENERAL DISCUSSION AND FUTURE WORK

6.1 General discussion

Currently, the research on derivatives produced by plants has being gaining interest in the food industry due to the increasing demand by consumers for natural derivatives produced by plants instead of artificial additives that are considered unsafe and have adverse effects on the health of consumers (Essawi and Srour, 2000). These active constituents derived from plants have a long history of use in traditional medicine for the prevention and treating of illness and infections and immensely acceptable by individuals in order to understanding that they are secure (Guarrera, 2005; Kurek et al., 2010). There are two interesting points to mention here. First – as mentioned in the introduction – plant products are also sources of many toxic compounds and medicinal herbs, including liquorice, are known to have adverse effects when consumed above a safe threshold. Therefore it is somewhat counter intuitive that consumers would see these as safer alternatives to well characterised and purified preservatives. However may be the fact that these new ingredients have less of an impact on the organoleptic properties of the food that make then attractive to the consumer. The second point to mention is that many of the common food preservatives include organic acids, such as acetic and propionic acids. These weak acids are naturally formed by many fermentation processes and therefore would be present in fermented foods, which remain acceptable to the consumer on the grounds that they are traditional. Hence this extensive attention in these plant derived compounds as replacements for microbial control is being driven in part as a marketing tool to allow food producers to make their products more attractive rather than because the existing antimicrobials are not performing well. In a recent review of "European consumer healthiness evaluation of 'Free-from' labelled food" (Hartmann et al., 2018); a major finding of the research was that consumers who were interested in 'free-from' labelled foods had a preference for naturalness as a feature of the product and were prepared to pay a price premium for products labelled as 'free-from'. Hence developing these products also makes good

economic sense for food producers. Nonetheless by developing these novel products the food industry also aims to protect producers protect human health through minimize the usage of chemical preservatives and also produce food with less acidic, with a lower salt content without compromising product quality in line with consumer demands (Naidu, 2000).

However, this work is not entirely driven by marketing since there is a real need for novel antimicrobials. There have been serious attempts by academia, industry and government agencies to control Listeria in foods and in food processing facilities which have so far not managed to find an answer to the problem posed to the food industry by this organism. Therefore new and improved methods to prevent the survival and growth of *Listeria* are still required. Different preservation techniques have been used in many studies in order to try and control of Listeria in foods, with a focus on achieving food safety without an effect on the sensory and nutritional qualities of foods (Gandhi and Chikindas, 2007). In line with this, the aim of this study was to evaluate the antimicrobial potential of liquorice extract that was provided as waste material from the production of liquorice flavourings for the food industry, with a specific idea of demonstrating that it was able to control the growth of this important food borne pathogen. The presented results in this study confirmed the fact that this extract did have significant antibacterial activity, specifically targeting Gram-positive bacteria and therefore is a good candidate to be tested as a strong antimicrobial agent to control both pathogens and Gram-positive spoilage organisms.

6.2 Inhibition of Gram-positive bacteria by liquorice extract

This study was first conducted *in vitro* to determine the antibacterial activity of liquorice extract of *Glycyrrhiza glabra* and confirm the promising role of a natural plant source as an antimicrobial that may help to reduce the current

problem of the resistance of many pathogens to chemicals traditionally used to preserve foods. To determine whether the extract had antimicrobial activity, a broad range of organisms consisting of Gram-positive and Gram-negative bacteria were selected to test the ability of liquorice extract to inhibit their growth. It was found that the extract only affected the growth of Gram-positive bacteria and at these concentrations all the Gram-negative bacteria tested showed normal levels of growth. Several previous studies have also found that the most bioactive compounds extracted from the plants were more active against Gram-positive bacteria than Gram-negative bacteria (Karahan *et al.*, 2016). In this study growth inhibition was noted for a range of different types of Gram-positive bacteria, including Gram-positive rod-shaped bacteria (L. monocytogenes; non-spore former and B. subtilis; spore-former) and Gram-positive cocci (S. aureus and E. faecalis), which might suggest that the extract exerted a similar mode of action on all of the bacteria tested. However the fact that the growth of the acid fast M. *smegmatis* was also inhibited was unexpected and suggests that while most types of Gram-positive bacteria were sensitive to this extract, it is unlikely that they all possess identical target sites that are affected by the same compound. This is also supported by the fact that as multiple effects were seen on the treated cells (membrane integrity and cell division). Perhaps the fact that the extract contains a mixture of bioactive compounds better explains the broad spectrum effects seen against Gram-positive bacteria and explains why some of the more purified compounds described in the literature did not have such a broad spectrum of activity.

Initial antimicrobial studies used different concentrations a range between $10-50 \ \mu g \ ml^{-1}$ to determine the antimicrobial activity of extract. An important point to note is that using optical density measurements alone, it was not possible to determine if the extract was having a bactericidal or bacteriostatic effect on the cells when growth was inhibited. Hence experiments were carried out using viable count to determine whether cell growth is just inhibited or whether the cells are

killed, and it was clear that the growth inhibition of the Gram-positive bacteria was reversible indicating that the extract was bacteriostatic. This suggests that the compound is not irreversibly binding to a target, and that the cell is possibly able to repair sub-lethal injury that is occurring, but the processes required to carry out this repair results in inhibition of bacterial growth. This idea is supported by the fact that it was found that a concentration of 50 μ g ml⁻¹ of liquorice extract was sufficient to inhibit the growth of Gram-positive bacteria, whereas at a lower concentration (12.5 μ g ml⁻¹) the growth of some of the bacteria tested was reduced but growth was still detected, and in some cases this was very close to the levels seen in the control sample indicating that this concentration was close to the break point (NOEL; No Observable Effect Level (Esdaile, 1995) for the extract at which no observable effect is seen. The pharmaceutical industry is very familiar with the concept of NOEL, in that it is accepted that cells or organisms treated with a drug will tolerate or overcome any adverse effects at concentrations below this threshold. In general food companies need to use preservatives at well above this level due to the heterogeneous nature of food products which can interact with the antimicrobials and affect their activity (Rai and Bai, 2014). Therefore these results suggest that 50 μ g ml⁻¹ is the minimum concentration that should be used as the in food application level.

It is interesting to note that other studies have identified specific extracts prepared from liquorice that produced a similar pattern of results to those seen in this study. Fukai *et al.* (2002b) investigated the antimicrobial activity of nineteen flavonoids commonly found in three species of liquorice namely, *G. glabra, G. inflata and G. uralensis* and found that they were active against the three Grampositive bacteria tested (*B. subtilis, S. aureus* and *M. luteus*) with MIC values in a similar range to the unpurified extract used in this study (MIC = $1.56-25 \mu g m l^{-1}$), but had not effect on the Gram negative bacteria tested (*E. coli, K. pneumonia* and *P. aeruginosa*). Similarly Kim *et al.* (2002) reported that the MIC of

glycyrrhizin, α -glycyrrhetinic acid and β -glycyrrhetinic acid isolated from *Glycyrrhiza glabra* was in the range of between 7.6 to 12.5 μ g ml⁻¹ for Grampositive bacteria such as *B. subtilis* and *S. epidermidis* but had no activity against *E. coli* and fungi. This group also used confocal microscopy to show that β glycyrrhetinic acid penetrated into the bacteria but did not cause membrane disruption suggesting that the mechanisms of action these compounds was not through disruption of the membrane potential or loss of the semi-permeable nature of the membrane itself. Since these different purified extracts had MIC values very close to those recorded in this study it suggests that the waste material used here contains a significant level of bioactive compounds, and the fact that multiple effects were seen on the cells suggests that there are different bioactive molecules which target different aspects of the Gram-positive bacterial cells. This makes the extract quite attractive for the food industry as it is a waste product that therefore could now be used as a food preservative and if it targets multiple sites in the bacterial cells they are less likely to be able to overcome inhibition of growth, eventually leading to cell death as occurs when Hurdle Technology is applied to food preservation. In fact there was evidence of this seen when the extract was applied to bologna style sausage, which created more stress conditions for the cells and the population of cells declined following a long period of exposure to the extract.

In contrast to the results gained using Gram-positive bacteria, no effect was detected on any Gram-negative bacteria tested. One suggestion is that the outer membrane of the Gram-negative bacteria helps prevent the passage of hydrophobic compounds into the cells (Zhang *et al.*, 2013). However the fact that other bioactive compounds prepared from liquorice can affect Gram-negative bacteria, like *H. pylori* as discussed in Chapter 1, suggests that this is not a full explanation. The fact that the highly hydrophobic *M. smegmatis* was also inhibited also argues against a model where the molecules are simply excluded from Gram-

negative cells because they have a hydrophobic surface layer. However cell structure and composition does play an important role in the sensitivity to antimicrobial agents (Hayrapetyan et al., 2012). The outer membrane is impermeable and does not allow the passage of large molecules but does allow limited diffusion of hydrophobic substances through its lipopolysaccharide covered surface (Zhang *et al.*, 2013). Whereas the cytoplasmic membrane acts to protect the internal contents of cell from leaking out by controlling the exit and entry of the substances into and out of the cell, the presence of the outer membrane in Gram-negative bacteria also provides an extra layer (the periplasmic space) which contains enzymes (including antibiotic degrading enzymes) and efflux pumps which contribute to the resistance of microorganisms to antibiotics by rapidly removing them from the cell (Levy, 2002; Poole, 2004) and to help the organism protect itself against antimicrobial challenge without affecting the exchange of material required for sustaining life (Delcour, 2009). As evidence was seen when very high levels of *Listeria* were added to the bologna style sausage that the bioactive molecules in the extract could be metabolized leading to outgrowth of the cells after prolonged incubation, perhaps the location of enzymes in the periplasmic space that can degrade the inhibitory compounds before they reach the target site might also help explain the difference in sensitivity seen.

However, although the basis of this result is not known, the finding is in line with the results of many other studies reported in the literature which showed that generally unpurified liquorice extracts affect Gram-positive bacteria more significantly than Gram-negative bacteria (Lin *et al.*, 1999; Parekh *et al.*, 2006; Parekh and Chanda, 2007). In general, plants have been shown to have much greater inhibition impact against Gram-positive than Gram-negative bacteria. The activity may be signal of the existence of broad-spectrum antimicrobial agents or metabolic toxins cause an inhibitory effect on both Gram-positive and Gramnegative bacteria (Srinivasan *et al.*, 2001). Over the last decades, bacterial

infectious disease becomes a significant problem and spread considerably among patients worldwide due to the high increase in antibiotic resistance (Dagan, 2003). Despite great progress in other areas of human medicine, infectious diseases caused by pathogenic bacteria are still a major threat to public health, in particular in developing countries due to the shortage of medicines and the emergence of widespread drug resistance (Okeke et al., 2005; Sosa et al., 2010). The emergence of antibiotic-resistant microorganisms has been attributed to the overuse or inappropriate use of antibiotics leading to serious problems in the treatment of infectious diseases (Nimmo et al., 2003). So increasingly research is being carried out to identify new antimicrobial substances derived from natural products which may play an important role in healthcare in the future (Shakya, 2016). Hence the fact that the extract had a broad spectrum of activity and is a food grade material could be of great significance. One suggestion is that it could be used to inhibit the growth of *Clostridium difficile* in the human gut in patients suffering with hospital acquired infections. This difficult to treat Gram-positive bacterium causes a major problem, particularly in patients recovering from extensive antibiotic therapy. A formulation of this extract given as a drink could help target these bacteria and allow normal gut flora to re-establish without the need for further extensive antibiotic therapy, and an extract that did not affect the Gram-negative bacteria in the gut may produce less problems of dysbiosis commonly associated with aggressive antibiotic treatments (Francino, 2016).

6.3 Evaluation of safety of liquorice extract as a food preservatives

The belief that natural products and herbs are safe and have no side effects and the possibility of using them in the long term is a misguided belief. This assumption is due to the lack of sufficient data on these herbs which confirm prove to be potentially hazardous, and therefore requires further studies to investigate their side effects and toxicity (Alves *et al.*, 2009). Thus, when plant compounds

are used for long term treatment, it is important to evaluate the toxic, cytotoxic and genotoxic damage caused by these plant compounds in order to reduce the possible risks of these agents (Rodeiro et al., 2006). Although natural products will continue to be extremely important as sources of medicinal agents, the literature also describes many plants containing mutagenic compounds, such as furocoumarins, tannins, anthraquinones, alkaloids, and flavonoids (Rietjens et al., 2005; Nesslany et al., 2009; da Rosa Guterres et al., 2013; Mininel et al., 2014) and therefore caution is required before these are introduced into human foods at significant levels. This evidence draws attention to the importance of studying the genetic risks of plant compounds, since the presence of mutagens in medicines can be dangerous to human health. In this study, Ames test and MTT assay were used to confirm if there are any side effects or mutagenicity for liquorice extract and the result obtained confirmed that there is no effect of liquorice on the cell line and there is no mutagenicity, so, this extract may play an important role toward the basic health requirements and may offer a new drugs use as antimicrobial and anticancer agents with significant activity against Gram-positive bacteria.

Although many of active constituents derived from plants may be safe and could be used as herbal therapies, other herbal products may be unsafe to use as the can be toxic to humans in high doses. This study investigated of the potential cytotoxicity of liquorice extract against Human Caucasian colon adenocarcinoma (Caco-2) cells using the MTT assay and the cytotoxicity activity of this extract was found to vary depending on concentration of liquorice as already found when determining its antimicrobial activity. Cytotoxicity was observed on Caco-2 cells treated with extract at a concentration of 50 μ g ml⁻¹ after 24 h incubation. However in this study a dose of 12.5 μ g ml⁻¹ showed no effect on the Caco-2 cells. Encouragingly Rao *et al.* (2014) found that liquorice (*Glycyrrhiza glabra*) extract had no effect on healthy cells when cells were exposed to different concentrations of the extract (50, 100 and 150 μ g ml⁻¹) but did see a toxicity against different

types of cancer cells. Therefore it will be important to evaluate the cytotoxicity of the extract used in this study against non-cancer cell lines, since a concentration of 12.5 µg ml⁻¹ seems to be close to the NOEL it would be difficult to use this effectively as an antimicrobial if a general effect on human cells was detected. Several herbs have been described as unsafe by The Food and Drug Administration, even in small doses, and for this reason they should not be utilized in either foods or beverages (Ekor, 2014). Thus, further research is required to investigate and understand the compounds responsible for cytotoxicity activity seen and their potential mechanisms.

However Glycyrrhizin and other liquorice compounds have been shown to have anti-carcinogenic properties and our results using a cancer cell line are in accordance with those reported by Badr et al. (2013) where the extract of liquorice has shown cytotoxic activities against intestinal carcinoma cell line (Caco-2) and prostate carcinoma cell line (PC-3) with different concentration of liquorice extract (0, 12.5, 25, 50 and 100 μ g ml⁻¹). In *in vitro* the anticancer activity was also observed against cancer cell lines, including HT-29 and MCF-7 cells treated with extract of G. glabra at different concentration (20, 200, and 2000 μ g ml⁻¹), after 24, 48 and 72 h incubation with a significant reduction observed in the viability of the treated cells (Nazmi et al., 2018). Similar to the previous study, (Khazraei-Moradian et al. (2017) reported that treating CT25, HT29, and Hek293 cells with different concentrations of liquorice extract (0, 20, 50, 100, and 200 μ g ml⁻¹) for 24 h showed decreases in the viability of cancer cell line significantly more than non-cancerous cell line and once the dose is increased, cell viability was reduced. In other similar studies it has been reported that liquorice and its main constituents decrease cell viability in different types of cancer cells such as gastrointestinal, leukemia, breast and prostate cancer cell lines (Hibasami et al., 2006; Lee et al., 2013; Shen et al., 2015). Although the exact mechanisms are still under investigation research has confirmed they inhibit tumor formation and growth of breast, skin and liver cancers (Vibha et al., 2009).

6.4 Conclusion

In conclusion, our study has demonstrated that liquorice extract exhibits antibacterial activity against all Gram-positive bacteria tested by using several techniques. A range of methods was used to investigate the mechanisms of action where Gram-stain was used to provide evidence that the extract affects the bacterial cell wall, shape change indicates cell division affected, bioluminescence showed that levels of metabolism were lower due to membrane damage. So, this work has found that a waste product from the food industry has potential as a novel antimicrobial product which could be used as a food additive to control bacteria either in food products.

Although liquorice extract at a concentration of 50 µg ml⁻¹ displayed *in vitro* cytotoxicity it was not mutagenic during Ames test. Hence, this plant species can be an important source of biologically active compounds and aid in the search for new effective and safe agents that act against spoilage bacteria and as food preservatives.

The antibacterial mechanism of liquorice extract in Gram-positive bacteria was investigated by examining morphology changed, membrane integrity, evaluating on bologna sausage and MTT assay. Based on the data obtained, it could be concluded that the antimicrobial effect of the extract against bacteria is attributed to its action of liquorice extract to Gram-positive bacteria cell membrane, which consequently increases bacteria cell membrane permeability and destroys the membrane integrity.

The results obtained in this thesis confirm the hypothesis that the herbal plants which traditionally are used in various forms in folk medicine since ancient times, due to the presence of antimicrobial compounds could be a potential source of antimicrobial compounds as well as a natural agents for food preservation.

6.5 Future study

This work has found that a waste product from the food industry has potential as a novel antimicrobial product which could be used as a food additive to control bacteria either in food products. In order to use this extract as a preservative to avoid contamination by Listeria, further studies are needed to determine an inhibitory dose and the qualitative aspects of the final product. These tests provides evidence that liquorice extract may have antimicrobial property at 50 µg ml⁻¹ concentration but more research on their mechanism of action are needed to understand the practicality of using them as an antimicrobial agent. In addition studies would be required to establish whether there would be any organoleptic changes to the foods if used at this level, as it has been noted that natural food preservatives can affect the acceptability of the food product to the consumer (Hintz et al., 2015). In addition it would be necessary to see how different food products interacted with the product to determine what type of foods it could be successfully applied to. Non-fermented meat products such as bologna style sausage, typically have high fat and protein content and have a pH between 6-7. In particular it would be interesting to see whether a change in pH of the product affected the ability of the extract to inhibit the growth of *Listeria* since many commonly used antimicrobials are only active in low pH foods (Boziaris and Nychas, 2006). In addition to its potential as a food preservation, an other area that could be investigated would be antioxidant activity which can also help to keep food in a palatable state. Sultana et al. (2010) and Li et al. (2016) have reported such studies when looking at the identification of the phytochemical properties of liquorice, and identifying how these could be applied to foods. Hence another area of study could be investigating these properties of the liquorice extract to prevent chemical food spoilage.

Currently in the UK, a "Novel Food" is defined as food that had not been consumed to a significant degree by humans in the EU before 15 May 1997, (Regulation (EU) 2015/2283). Before such foods can be marketed, pre-market

safety assessments are evaluated by the regulatory authorities, which in the UK is the Advisory Committee on Novel Foods and Processes (ACNFP) before a decision is made on EU-wide authorisation. To do this the producer must prepare a dossier on the new novel food, providing the information needed so the European Food Standards Agency can carry out a safety assessment. In this study evaluating the safety and toxicity of the extract was carried out *in vitro* using cultured human cell lines and also using a bacterial mutagenicity test (Ames test). If the extract is to be used for food preservation or medicinal purposes, it is essential that more safety studies are carried out *in vivo* using animal models to identify any toxicological effects of the liquorice extract as these would be needed before the extract could be used in any commercial food products.

Several reviews have reported that the efficacy of conventional antimicrobials can be improved by use in combination with plant extracts and there are some reports in the literature about the effect of liquorice plant extracts being used in combination with other plant extracts and showing that their action may be improved by synergistic interactions with other compounds. Haroun and Al-Kayali (2016) have been looking at how antibiotics interact with other natural plant extracts, however no published studies have investigated the effect of combination of liquorice extract with antibiotics against growth of bacteria. Since the results in this study have shown that this extract inhibits Gram-positive bacterial growth at concentrations similar to the MIC of antibiotics, it would be interesting to see if an additive effect can be seen if using combinations of antibiotics and extract. Although there have been a number of papers that have focused on activity of antimicrobial studies of liquorice plant against Gram-positive and Gram-negative bacteria (Gupta et al., 2013; Soulef et al., 2014), this has been noted in other studies where the components isolated from liquorice plant were found to have good activities against Mycobacterium tuberculosis (Gupta et al., 2008), however, very little research has been conducted on the effect of extract on Mycobacteria. This will be interesting to follow in terms of understanding the mode of action as

Mycobacteria have a very different cell wall structure to the other Gram-positive bacteria. Understanding what the active site is – or determining whether it is the same as that in Gram-positive bacteria - would be an interesting area of research to take forward.

In addition to the food applications, there are many examples in the literature where these types of extracts from plants have been shown to have potential anti-cancer activities. Therefore another area of study on the liquorice extract could be as a good source of alternative medicines for treating cancer. If work is performed to determine the active compounds in the liquorice extract, it may be possible to predict whether such activity is likely, based on the structure of active molecules that have previously been studies, and therefore, there is the possibility of using this material as a source of the active compounds that confer anticancer activities on liquorice extract. In addition, the long history of use of liquorice plant as a medicinal means there are also reports that suggest it contains many active constituents that have been reported to have antispasmodic, antitussive, anti-ulcer and anti-inflammatory effects. For instance a study conducted by Tanemoto et al. (2015) reported that active constituents isolated from G. uralensis and examined its anti-inflammatory effects. However, reports comparing the anti-inflammatory effects of these constituents are very few. Thus, isolation and purification of components to determine the active compounds in liquorice extract could be another interesting study to carry out.

Hence there are many possible applications of this extract that could be studied in the future, and the results presented in this thesis now provide a basis for future research that may reveal its potential uses in a number of different areas.

REFERENCES

AGGARWAL, H., GHOSH, J., RAO, A. & CHHOKAR, V. 2015. Evaluation of root and leaf extracts of *Glycrriza glabra* for antimicrobial activity. *Journal of Medical and Bioengineering Volume*, 4.

AJAGANNANAVAR, S. L., BATTUR, H., SHAMARAO, S., SIVAKUMAR, V., PATIL, P. U. & SHANAVAS, P. 2014. Effect of aqueous and alcoholic licorice (*glycyrrhiza glabra*) root extract against *streptococcus mutans* and *lactobacillus acidophilus* in comparison to chlorhexidine: an *in vitro* study. *Journal of International Oral Health: JIOH,* 6, 29.

AL-TEREHI, M., AL-SAADI, A. H., ZAIDAN, H., BEHJET, R. H. & HALEEM, Z. 2015a. Some plants extracts Synergism effects in Pathogenic bacteria. *International Journal of PharmTech Research*, 8, 158-164.

AL-TEREHI, M. N., AL SAADI, A. H., ZAIDAN, H. K., AL AMERI, Q. M. A. & EWADH, M. J. 2015b. *In vivo* study of antimutagenic and antioxidant activity of *Glycyrrhiza glabra* root extract. *Research in Pharmacy*, 2.

ALOCILJA, E. C. & RADKE, S. M. 2003. Market analysis of biosensors for food safety. *Biosensors and Bioelectronics*, 18, 841-846.

ALONSO, J. L., MASCELLARO, S., MORENO, Y., FERRÚS, M. A. & HERNÁNDEZ, J. 2002. Double-staining method for differentiation of morphological changes and membrane integrity of *Campylobacter coli* cells. *Applied and Environmental Microbiology*, 68, 5151-5154.

ALVES, A., VIDAL, L., KUSTER, R., LAGE, C. & LEITÃO, A. 2009. Genotoxic and mutagenic effects of Melissa officinalis (Erva Cidreira) extracts. *Open Toxicology Journal*, **3**, 58-69.

AMES, B. N. 1971. The detection of chemical mutagens with enteric bacteria. *Chemical mutagens.* Springer.

AMES, B. N., DURSTON, W. E., YAMASAKI, E. & LEE, F. D. 1973a. Carcinogens are mutagens: a simple test system combining liver homogenates for activation and bacteria for detection. *Proceedings of the National Academy of Sciences*, 70, 2281-2285.

AMES, B. N., LEE, F. D. & DURSTON, W. E. 1973b. An improved bacterial test system for the detection and classification of mutagens and carcinogens. *Proceedings of the National Academy of Sciences*, 70, 782-786.

AMES, B. N., MCCANN, J. & YAMASAKI, E. 1975. Methods for detecting carcinogens and mutagens with the *Salmonella*/mammalian-microsome mutagenicity test. *Mutation Research/Environmental Mutagenesis and Related Subjects*, 31, 347-363.

ANSARI, M. I. & MALIK, A. 2009. Genotoxicity of wastewaters used for irrigation of food crops. *Environmental Toxicology*, 24, 103-115.

APPENDINI, P. & HOTCHKISS, J. H. 2002. Review of antimicrobial food packaging. *Innovative Food Science & Emerging Technologies*, **3**, 113-126.

ARQUÉS, J. L., RODRÍGUEZ, E., NUÑEZ, M. & MEDINA, M. 2008. Inactivation of Gram-negative pathogens in refrigerated milk by reuterin in combination with nisin or the lactoperoxidase system. *European Food Research and Technology*, 227, 77-82.

ASANO, T., ISHIHARA, K., MOROTA, T., TAKEDA, S. & ABURADA, M. 2003. Permeability of the flavonoids liquiritigenin and its glycosides in licorice roots and davidigenin, a hydrogenated metabolite of liquiritigenin, using human intestinal cell line Caco-2. *Journal of Ethnopharmacology*, 89, 285-289.

ASHA, M. K., DEBRAJ, D., EDWIN, J. R., SRIKANTH, H., MURUGANANTHAM, N., DETHE, S. M., ANIRBAN, B., JAYA, B., DEEPAK, M. & AGARWAL, A. 2013. *In vitro* anti-*Helicobacter pylori* activity of a flavonoid rich extract of *Glycyrrhiza glabra* and its probable mechanisms of action. *Journal of Ethnopharmacology*, 145, 581-586.

ASL, M. N. & HOSSEINZADEH, H. 2008. Review of pharmacological effects of *Glycyrrhiza* sp. and its bioactive compounds. *Phytotherapy Research*, 22, 709-724.

AUTHORITY, E. F. S., PREVENTION, E. C. F. D. & CONTROL 2018. Multi-country outbreak of *Listeria monocytogenes* serogroup IV b, multi-locus sequence type 6, infections linked to frozen corn and possibly to other frozen vegetables–first update. *EFSA Supporting Publications*, 15, 1448E.

B COTA, B., M BERTOLLO, C. & M DE OLIVEIRA, D. 2013. Anti-allergic potential of herbs and herbal natural products-activities and patents. *Recent Patents on Endocrine, Metabolic & Immune Drug Discovery*, **7**, 26-56.

BADR, A., OMAR, N. & BADRIA, F. 2011. A laboratory evaluation of the antibacterial and cytotoxic effect of liquorice when used as root canal medicament. *International Endodontic Journal*, 44, 51-58.

BADR, S. E., SAKR, D. M., MAHFOUZ, S. A. & ABDELFATTAH, M. S. 2013. Licorice (*Glycyrrhiza glabra* L.): chemical composition and biological impacts. *Research Journal of Pharmaceutical, Biological and Chemical Sciences*, 4, 606-621.

BAKER, R. D., BAKER, S. S. & LAROSA, K. 1995. Polarized caco-2 cells. *Digestive Diseases and Sciences*, 40, 510-518.

BALOUIRI, M., SADIKI, M. & IBNSOUDA, S. K. 2016. Methods for *in vitro* evaluating antimicrobial activity: a review. *Journal of Pharmaceutical Analysis*, 6, 71-79.

BARAKETI, A., SALMIERI, S. & LACROIX, M. 2018. Foodborne pathogens detection: persevering worldwide challenge. *Biosensing Technologies for the Detection of Pathogens-A Prospective Way for Rapid Analysis.* InTech.

BARMPALIA, I. M., KOUTSOUMANIS, K. P., GEORNARAS, I., BELK, K. E., SCANGA, J. A., KENDALL, P. A., SMITH, G. C. & SOFOS, J. N. 2005. Effect of antimicrobials as ingredients of pork bologna for *Listeria monocytogenes* control during storage at 4 or 10 C. *Food Microbiology*, 22, 205-211.

BASSYOUNI, R. H., KAMEL, Z., MEGAHID, A. & SAMIR, E. 2012. Antimicrobial potential of licorice: leaves versus roots. *African Journal of Microbiology Research*, 6, 7485-7493.

BEALES, N. 2004. Adaptation of microorganisms to cold temperatures, weak acid preservatives, low pH, and osmotic stress: a review. *Comprehensive Reviews in Food Science and Food Safety*, **3**, 1-20.

BECHINGER, B. & GORR, S.-U. 2017. Antimicrobial peptides: mechanisms of action and resistance. *Journal of Dental Research*, 96, 254-260.

BEREKSI, N., GAVINI, F., BENEZECH, T. & FAILLE, C. 2002. Growth, morphology and surface properties of *Listeria monocytogenes* Scott A and LO28 under saline and acid environments. *Journal of Applied Microbiology*, 92, 556-565.

BERESFORD, M., ANDREW, P. & SHAMA, G. 2001. *Listeria monocytogenes* adheres to many materials found in food-processing environments. *Journal of Applied Microbiology*, 90, 1000-1005.

BERNEY, M., HAMMES, F., BOSSHARD, F., WEILENMANN, H.-U. & EGLI, T. 2007. Assessment and interpretation of bacterial viability by using the LIVE/DEAD BacLight Kit in combination with flow cytometry. *Applied and Environmental Microbiology*, 73, 3283-3290.

BERRIDGE, M. V., HERST, P. M. & TAN, A. S. 2005. Tetrazolium dyes as tools in cell biology: new insights into their cellular reduction. *Biotechnology Annual Review*, 11, 127-152.

BERTRAND, S., CEYSSENS, P., YDE, M., DIERICK, K., BOYEN, F., VANDERPAS, J., VANHOOF, R. & MATTHEUS, W. 2016. Diversity of *Listeria monocytogenes* strains of clinical and food chain origins in Belgium between 1985 and 2014. *PloS One*, 11, e0164283.

BETTENCOURT, P., PIRES, D., CARMO, N. & ANES, E. 2010. Application of confocal microscopy for quantification of intracellular mycobacteria in macrophages. *Microscopy: Science, Technology, Applications and Education,* 614.

BHATNAGAR, I. & KIM, S.-K. 2010. Immense essence of excellence: marine microbial bioactive compounds. *Marine Drugs*, 8, 2673-2701.

BIESTA-PETERS, E. G., REIJ, M. W., JOOSTEN, H., GORRIS, L. G. & ZWIETERING, M. H. 2010. Comparison of two optical-density-based methods and a plate count method for estimation of growth parameters of *Bacillus cereus*. *Applied and Environmental Microbiology*, 76, 1399-1405.

BISHOP, D. & HINRICHS, D. 1987. Adoptive transfer of immunity to *Listeria monocytogenes*. The influence of *in vitro* stimulation on lymphocyte subset requirements. *The Journal of Immunology*, 139, 2005-2009.

BOADA, L. D., HENRÍQUEZ-HERNÁNDEZ, L. & LUZARDO, O. 2016. The impact of red and processed meat consumption on cancer and other health outcomes: epidemiological evidences. *Food and Chemical Toxicology*, 92, 236-244.

BOSSHARD, F., BERNEY, M., SCHEIFELE, M., WEILENMANN, H.-U. & EGLI, T. 2009. Solar disinfection (SODIS) and subsequent dark storage of *Salmonella* Typhimurium and *Shigella flexneri* monitored by flow cytometry. *Microbiology*, 155, 1310-1317.

BOULOS, L., PREVOST, M., BARBEAU, B., COALLIER, J. & DESJARDINS, R. 1999. LIVE/DEAD[®] BacLight[™]: application of a new rapid staining method for direct enumeration of viable and total bacteria in drinking water. *Journal of Microbiological Methods*, 37, 77-86.

BOZIARIS, I. & NYCHAS, G.-J. 2006. Effect of nisin on growth boundaries of *Listeria monocytogenes* Scott A, at various temperatures, pH and water activities. *Food microbiology*, 23, 779-784.

BOZORGI, M., AMIN, G., OSTAD, S., SAMADI, N., NAZEM, E. & SHEKARCHI, M. 2017. Toxicological, chemical and antibacterial evaluation of squill vinegar, a useful product in Persian Traditional Medicine. *Research Journal of Pharmacognosy*, 4, 33-39.

BRZIN, B. 1973. The effect of NaCl on the morphology of *Listeria monocytogenes*. *Zentralblatt für Bakteriologie, Parasitenkunde, Infektionskrankheiten und Hygiene. Erste Abteilung Originale. Reihe A: Medizinische Mikrobiologie und Parasitologie,* 225, 80-84.

BRZIN, B. 1975. Further observations of changed growth of *Listeria* monocytogenes on salt agar. *Zentralblatt fur Bakteriologie, Parasitenkunde, Infektionskrankheiten und Hygiene. Erste Abteilung Originale. Reihe A: Medizinische Mikrobiologie und Parasitologie,* 232, 287-293.

BURGESS, C. M., GIANOTTI, A., GRUZDEV, N., HOLAH, J., KNØCHEL, S., LEHNER, A., MARGAS, E., ESSER, S. S., SELA, S. & TRESSE, O. 2016. The response of foodborne pathogens to osmotic and desiccation stresses in the food chain. *International Journal of Food Microbiology*, 221, 37-53.

BURT, S. 2004. Essential oils: their antibacterial properties and potential applications in foods—a review. *International Journal of Food Microbiology*, 94, 223-253.

BURT, S. A. & REINDERS, R. D. 2003. Antibacterial activity of selected plant essential oils against *Escherichia coli* O157: H7. *Letters in applied microbiology*, 36, 162-167.

BURT, S. A., VAN DER ZEE, R., KOETS, A. P., DE GRAAFF, A. M., VAN KNAPEN, F., GAASTRA, W., HAAGSMAN, H. P. & VELDHUIZEN, E. J. 2007. Carvacrol induces heat shock protein 60 and inhibits synthesis of flagellin in *Escherichia coli* O157: H7. *Applied and Environmental Microbiology*, 73, 4484-4490.

CARAGAY, A. B. 1992. Cancer-preventive foods and ingredients. *Food Technology* (USA).

CARVALHO, M. R., SILVA, B. M., SILVA, R., VALENTÃO, P. C., ANDRADE, P. B. & BASTOS, M. L. 2010. First report on *Cydonia oblonga* Miller anticancer potential: differential antiproliferative effect against human kidney and colon cancer cells. *Journal of Agricultural and food Chemistry*, 58, 3366-3370.

CAVALCANTI, B., COSTA-LOTUFO, L., MORAES, M., BURBANO, R., SILVEIRA, E., CUNHA, K., RAO, V., MOURA, D., ROSA, R. & HENRIQUES, J. 2006. Genotoxicity evaluation of kaurenoic acid, a bioactive diterpenoid present in Copaiba oil. *Food and Chemical Toxicology*, 44, 388-392.

CDC. 2011. Foodborne Diseases Centers for Outbreak Response Enhancement | FoodCORE | CDC. [online] Available at: https://www.cdc.gov/foodcore/index.html [Accessed 26 Mar. 2018].

CETIN-KARACA, H. 2011. Evaluation of natural antimicrobial phenolic compounds against foodborne pathogens. University of Kentucky Master's Theses. Paper 652. <u>https://uknowledge.uky.edu/gradschool_theses/652/</u>.

CEYLAN, E. & FUNG, D. Y. 2004. Antimicrobial activity of spices. *Journal of Rapid Methods & Automation in Microbiology*, 12, 1-55.

CHABUCK, Z. A. G., HADI, B. H. & HINDI, N. K. K. 2018. Evaluation of antimicrobial activity of different aquatic extracts against bacterial isolates from UTI in Babylon Province, Iraq. *Journal of Pure and Applied Microbiology*, 12, 693-700.

CHAMBERS, H. F. 2001. The changing epidemiology of *Staphylococcus aureus*? *Emerging Infectious Diseases*, 7, 178.

CHATTORAJ, M., KING, B. A., BUBLITZ, G. U. & BOXER, S. G. 1996. Ultra-fast excited state dynamics in green fluorescent protein: multiple states and proton transfer. *Proceedings of the National Academy of Sciences*, 93, 8362-8367.

CHEMBURU, S., WILKINS, E. & ABDEL-HAMID, I. 2005. Detection of pathogenic bacteria in food samples using highly-dispersed carbon particles. *Biosensors and Bioelectronics*, 21, 491-499.

CHÉRIGO, L., PEREDA-MIRANDA, R., FRAGOSO-SERRANO, M., JACOBO-HERRERA, N., KAATZ, G. W. & GIBBONS, S. 2008. Inhibitors of bacterial multidrug efflux pumps from the resin glycosides of *Ipomoea murucoides*. *Journal of Natural Products*, 71, 1037-1045.

CHESHOMI, H., ALDAGHI, L. S. & SERESHT, H. R. 2016. Cytotoxicity of the methanol extract of *Datura innoxia* petals on MCF-7 and HEK-293 Cell Lines. *Journal of Biomedical Science*, 1, e6623.

CHO, W. I., CHOI, J. B., LEE, K., CHUNG, M. S. & PYUN, Y. R. 2008. Antimicrobial activity of torilin isolated from *Torilis japonica* fruit against *Bacillus subtilis*. *Journal of Food Science*, 73, M37-M46.

CHOFFNES, E. R., RELMAN, D. A., OLSEN, L., HUTTON, R. & MACK, A. 2012. *Improving food safety through a one health approach: workshop summary*, National Academies Press.

CHOUHAN, S., SHARMA, K. & GULERIA, S. 2017. Antimicrobial activity of some essential oils—present status and future perspectives. *Medicines*, 4, 58.

COMMISSION, E. 2005. Commission Regulation (EC) No 2073/2005 of 15 November 2005 on microbiological criteria for foodstuffs. *Official Journal of the European Union*, 50, 1-26.

COS, P., VLIETINCK, A. J., BERGHE, D. V. & MAES, L. 2006. Anti-infective potential of natural products: how to develop a stronger *in vitro* 'proof-of-concept'. *Journal of Ethnopharmacology*, 106, 290-302.

COWAN, M. M. 1999. Plant products as antimicrobial agents. *Clinical Microbiology Reviews*, 12, 564-582.

DA ROSA GUTERRES, Z., DA SILVA, A. F. G., GARCEZ, W. S., GARCEZ, F. R., FERNANDES, C. A. & GARCEZ, F. R. 2013. Mutagenicity and recombinagenicity of *Ocotea acutifolia* (Lauraceae) aporphinoid alkaloids. *Mutation Research/Genetic Toxicology and Environmental Mutagenesis*, 757, 91-96.

DA SILVA DANNENBERG, G., FUNCK, G. D., MATTEI, F. J., DA SILVA, W. P. & FIORENTINI, Â. M. 2016. Antimicrobial and antioxidant activity of essential oil from pink pepper tree (*Schinus terebinthifolius* Raddi) *in vitro* and in cheese

experimentally contaminated with *Listeria monocytogenes*. *Innovative Food Science & Emerging Technologies*, 36, 120-127.

DAGAN, R. 2003. Antibiotic resistance and the potential impact of pneumococcal conjugate vaccines. *Communicable Diseases Intelligence Quarterly Report,* 27, S134.

DAVIDSON, P. M., TAYLOR, T. M. & SCHMIDT, S. E. 2013. Chemical preservatives and natural antimicrobial compounds. *Food Microbiology*. American Society of Microbiology.

DE LAS HERAS, A., CAIN, R. J., BIELECKA, M. K. & VAZQUEZ-BOLAND, J. A. 2011. Regulation of *Listeria* virulence: PrfA master and commander. *Current Opinion in Microbiology*, 14, 118-127.

DEL NOBILE, M. A., LUCERA, A., COSTA, C. & CONTE, A. 2012. Food applications of natural antimicrobial compounds. *Frontiers in Microbiology*, **3**, 287.

DELBÒ, R. M. 2013. Assessment report on *Glycyrrhiza glabra* L. and/or *Glycyrrhiza inflata* Bat. and/or *Glycyrrhiza uralensis* Fisch., radix.

DELCOUR, A. H. 2009. Outer membrane permeability and antibiotic resistance. *Biochimica et Biophysica Acta (BBA)-Proteins and Proteomics*, 1794, 808-816.

DESMOND, E. 2006. Reducing salt: A challenge for the meat industry. *Meat Science*, 74, 188-196.

DI BONAVENTURA, G., PICCOLOMINI, R., PALUDI, D., D'ORIO, V., VERGARA, A., CONTER, M. & IANIERI, A. 2008. Influence of temperature on biofilm formation by *Listeria monocytogenes* on various food-contact surfaces: relationship with motility and cell surface hydrophobicity. *Journal of Applied Microbiology*, 104, 1552-1561.

DIXON, R. A. & SUMNER, L. W. 2003. Legume natural products: understanding and manipulating complex pathways for human and animal health. *Plant Physiology*, 131, 878-885.

DOUGLAS, J., DOUGLAS, M., LAUREN, D., MARTIN, R., DEO, B., FOLLETT, J. & JENSEN, D. 2004. Effect of plant density and depth of harvest on the production and quality of licorice (*Glycyrrhiza glabra*) root harvested over 3 years. *New Zealand Journal of Crop and Horticultural Science*, 32, 363-373.

DOUMITH, M., BUCHRIESER, C., GLASER, P., JACQUET, C. & MARTIN, P. 2004. Differentiation of the major *Listeria monocytogenes* serovars by multiplex PCR. *Journal of Clinical Microbiology*, 42, 3819-3822.

DREVETS, D. A. & BRONZE, M. S. 2008. *Listeria monocytogenes*: epidemiology, human disease, and mechanisms of brain invasion. *FEMS Immunology & Medical Microbiology*, 53, 151-165.

EDEOGA, H., OKWU, D. & MBAEBIE, B. 2005. Phytochemical constituents of some Nigerian medicinal plants. *African Journal of Biotechnology*, 4, 685-688.

EITING, M., HAGELÜKEN, G., SCHUBERT, W. D. & HEINZ, D. W. 2005. The mutation G145S in PrfA, a key virulence regulator of *Listeria monocytogenes*, increases DNA-binding affinity by stabilizing the HTH motif. *Molecular Microbiology*, 56, 433-446.

EKOR, M. 2014. The growing use of herbal medicines: issues relating to adverse reactions and challenges in monitoring safety. *Frontiers in pharmacology*, 4, 177.

ESDAILE, D. J. 1995. Principles, benefits and limitations of the NOEL approach. *Toxicology in Transition.* Springer.

ESPITIA, P. J. P., PACHECO, J. J. R., MELO, N. R. D., SOARES, N. D. F. F. & DURANGO, A. M. 2013. Packaging properties and control of *Listeria monocytogenes* in bologna by cellulosic films incorporated with pediocin. *Brazilian Journal of Food Technology*, 16, 226-235.

ESSAWI, T. & SROUR, M. 2000. Screening of some Palestinian medicinal plants for antibacterial activity. *Journal of Ethnopharmacology*, 70, 343-349.

EVERIS, L. & BETTS, G. 2001. PH stress can cause cell elongation in *Bacillus* and *Clostridium* species: a research note. *Food Control*, 12, 53-56.

FAIR, R. J. & TOR, Y. 2014. Antibiotics and bacterial resistance in the 21st century. *Perspectives in Medicinal Chemistry*, 6, 25.

FAO, G. 2011. Global food losses and food waste-Extent, causes and prevention. *SAVE FOOD: An Initiative on Food Loss and Waste Reduction*.

FARHA, M. A., VERSCHOOR, C. P., BOWDISH, D. & BROWN, E. D. 2013. Collapsing the proton motive force to identify synergistic combinations against *Staphylococcus aureus*. *Chemistry & Biology*, 20, 1168-1178.

FATTOUCH, S., CABONI, P., CORONEO, V., TUBEROSO, C. I., ANGIONI, A., DESSI, S., MARZOUKI, N. & CABRAS, P. 2007. Antimicrobial activity of Tunisian quince (*Cydonia oblonga* Miller) pulp and peel polyphenolic extracts. *Journal of Agricultural and Food Chemistry*, 55, 963-969.

FERNÁNDEZ-NO, I., GUARDDON, M., BÖHME, K., CEPEDA, A., CALO-MATA, P. & BARROS-VELÁZQUEZ, J. 2011. Detection and quantification of spoilage and pathogenic *Bacillus cereus*, *Bacillus subtilis* and *Bacillus licheniformis* by real-time PCR. *Food Microbiology*, 28, 605-610.

FERREIRA, V., WIEDMANN, M., TEIXEIRA, P. & STASIEWICZ, M. 2014. *Listeria monocytogenes* persistence in food-associated environments: epidemiology, strain characteristics, and implications for public health. *Journal of Food Protection*, 77, 150-170.

FIAMEGOS, Y. C., KASTRITIS, P. L., EXARCHOU, V., HAN, H., BONVIN, A. M., VERVOORT, J., LEWIS, K., HAMBLIN, M. R. & TEGOS, G. P. 2011. Antimicrobial and efflux pump inhibitory activity of caffeoylquinic acids from *Artemisia absinthium* against Gram-positive pathogenic bacteria. *PLoS One*, 6, e18127.

FIORE, C., EISENHUT, M., KRAUSSE, R., RAGAZZI, E., PELLATI, D., ARMANINI, D. & BIELENBERG, J. 2008. Antiviral effects of *Glycyrrhiza* species. *Phytotherapy Research: An International Journal Devoted to Pharmacological and Toxicological Evaluation of Natural Product Derivatives*, 22, 141-148.

FIORE, C., EISENHUT, M., RAGAZZI, E., ZANCHIN, G. & ARMANINI, D. 2005. A history of the therapeutic use of liquorice in Europe. *Journal of Ethnopharmacology*, 99, 317-324.

FRANCINO, M. 2016. Antibiotics and the human gut microbiome: dysbioses and accumulation of resistances. *Frontiers in Microbiology*, 6, 1543.

FRESCO, P., BORGES, F., MARQUES, M. & DINIZ, C. 2010. The anticancer properties of dietary polyphenols and its relation with apoptosis. *Current Pharmaceutical Design*, 16, 114-134.

FRIIS-MØLLER, A., CHEN, M., FUURSTED, K., CHRISTENSEN, S. B. & KHARAZMI, A. 2002. *In vitro* antimycobacterial and antilegionella activity of licochalcone A from Chinese licorice roots. *Planta Medica*, 68, 416-419.

FRYE, D. M., ZWEIG, R., STURGEON, J., TORMEY, M., LECAVALIER, M., LEE, I., LAWANI, L. & MASCOLA, L. 2002. An outbreak of febrile gastroenteritis associated with delicatessen meat contaminated with *Listeria monocytogenes*. *Clinical Infectious Diseases*, 35, 943-949.

FU, Y., HSIEH, T.-C., GUO, J., KUNICKI, J., LEE, M. Y., DARZYNKIEWICZ, Z. & WU, J. M. 2004. Licochalcone-A, a novel flavonoid isolated from licorice root (*Glycyrrhiza glabra*), causes G2 and late-G1 arrests in androgen-independent PC-3 prostate cancer cells. *Biochemical and Biophysical Research Communications*, 322, 263-270.

FUKAI, T., MARUMO, A., KAITOU, K., KANDA, T., TERADA, S. & NOMURA, T. 2002a. Anti-*Helicobacter pylori* flavonoids from licorice extract. *Life Sciences*, 71, 1449-1463.

FUKAI, T., MARUMO, A., KAITOU, K., KANDA, T., TERADA, S. & NOMURA, T. 2002b. Antimicrobial activity of licorice flavonoids against methicillin-resistant *Staphylococcus aureus*. *Fitoterapia*, 73, 536-539.

FUKUCHI, K., OKUDAIRA, N., ADACHI, K., ODAI-IDE, R., WATANABE, S., OHNO, H., YAMAMOTO, M., KANAMOTO, T., TERAKUBO, S. & NAKASHIMA, H. 2016. Antiviral and antitumor activity of licorice root extracts. *In Vivo*, 30, 777-785.

GAGANDEEP, DHIMAN, M., MENDIZ, E., RAO, A. & KALE, R. 2005. Chemopreventive effects of mustard (*Brassica compestris*) on chemically induced tumorigenesis in murine forestomach and uterine cervix. *Human & Experimental Toxicology*, 24, 303-312.

GALLAGHER, D. L., EBEL, E. D. & KAUSE, J. R. 2003. FSIS risk assessment for *Listeria monocytogenes* in deli meats. *Food Safety and Inspection Service, US Department of Agriculture*.

GANDHI, M. & CHIKINDAS, M. L. 2007. *Listeria*: a foodborne pathogen that knows how to survive. *International Journal of Food Microbiology*, 113, 1-15.

GARDAN, R., COSSART, P. & LABADIE, J. 2003. Identification of *Listeria monocytogenes* genes involved in salt and alkaline-pH tolerance. *Applied and Environmental Microbiology*, 69, 3137-3143.

GAUL, L. K., FARAG, N. H., SHIM, T., KINGSLEY, M. A., SILK, B. J. & HYYTIA-TREES, E. 2012. Hospital-acquired listeriosis outbreak caused by contaminated diced celery—Texas, 2010. *Clinical Infectious Diseases*, 56, 20-26.

GAULIN, C., RAMSAY, D. & BEKAL, S. 2012. Widespread listeriosis outbreak attributable to pasteurized cheese, which led to extensive cross-contamination affecting cheese retailers, Quebec, Canada, 2008. *Journal of Food Protection*, 75, 71-78.

GEETHA, R. & ROY, A. 2012. *In vitro* evaluation of anti-bacterial activity of ethanolic root extract of *Glycyrrhiza glabra* on oral microbes. *International Journal* of Drug Development and Research, 4, 161-165.

GIALAMAS, H., ZINOVIADOU, K. G., BILIADERIS, C. G. & KOUTSOUMANIS, K. P. 2010. Development of a novel bioactive packaging based on the incorporation of *Lactobacillus sakei* into sodium-caseinate films for controlling *Listeria monocytogenes* in foods. *Food Research International*, 43, 2402-2408.

GILL, A. & HOLLEY, R. 2006a. Disruption of *Escherichia coli*, *Listeria monocytogenes* and *Lactobacillus sakei* cellular membranes by plant oil aromatics. *International Journal of Food Microbiology*, 108, 1-9.

GILL, A. & HOLLEY, R. 2006b. Inhibition of membrane bound ATPases of *Escherichia coli* and *Listeria monocytogenes* by plant oil aromatics. *International Journal of Food Microbiology*, 111, 170-174.

GILL, C., BADONI, M. & JONES, T. 2007. Behaviours of log phase cultures of eight strains of *Escherichia coli* incubated at temperatures of 2, 6, 8 and 10° C. *International Journal of Food Microbiology*, 119, 200-206.

GIOTIS, E. S., BLAIR, I. S. & MCDOWELL, D. A. 2007. Morphological changes in *Listeria monocytogenes* subjected to sublethal alkaline stress. *International Journal of Food Microbiology*, 120, 250-258.

GM GAHAN, C. 2012. The bacterial lux reporter system: applications in bacterial localisation studies. *Current Gene Therapy*, 12, 12-19.

GONI, P., LÓPEZ, P., SÁNCHEZ, C., GÓMEZ-LUS, R., BECERRIL, R. & NERÍN, C. 2009. Antimicrobial activity in the vapour phase of a combination of cinnamon and clove essential oils. *Food chemistry*, 116, 982-989.

GORMLEY, F., LITTLE, C., GRANT, K., DE PINNA, E. & MCLAUCHLIN, J. 2010. The microbiological safety of ready-to-eat specialty meats from markets and specialty food shops: a UK wide study with a focus on *Salmonella* and *Listeria monocytogenes*. *Food Microbiology*, 27, 243-249.

GRANDJEAN, P. 2016. Paracelsus revisited: the dose concept in a complex world. *Basic & Clinical Pharmacology & Toxicology*, 119, 126-132.

GRAVES, L. M., HELSEL, L. O., STEIGERWALT, A. G., MOREY, R. E., DANESHVAR, M. I., ROOF, S. E., ORSI, R. H., FORTES, E. D., MILILLO, S. R. & DEN BAKKER, H. C. 2010. *Listeria marthii* sp. nov., isolated from the natural environment, Finger Lakes National Forest. *International Journal of Systematic and Evolutionary Microbiology*, 60, 1280-1288.

GREGOR, C., GWOSCH, K. C., SAHL, S. J. & HELL, S. W. 2018. Strongly enhanced bacterial bioluminescence with the *ilux* operon for single-cell imaging. *Proceedings* of the National Academy of Sciences, 115(5), 962-967.

GUARRERA, P. M. 2005. Traditional phytotherapy in Central Italy (Marche, Abruzzo, and Latium). *Fitoterapia*, 76, 1-25.

GUPTA, A., MAHESHWARI, D. & KHANDELWAL, G. 2013. Antibacterial activity of *Glycyrrhiza glabra* roots against certain Gram-positive and Gram-negative bacterial strains. *Journal of Applied and Natural Science*, **5**, 459-464.

GUPTA, V. K., FATIMA, A., FARIDI, U., NEGI, A. S., SHANKER, K., KUMAR, J., RAHUJA, N., LUQMAN, S., SISODIA, B. S. & SAIKIA, D. 2008. Antimicrobial potential of *Glycyrrhiza glabra* roots. *Journal of Ethnopharmacology*, 116, 377-380.

GUTIERREZ, J., BARRY-RYAN, C. & BOURKE, P. 2009. Antimicrobial activity of plant essential oils using food model media: efficacy, synergistic potential and interactions with food components. *Food Microbiology*, 26, 142-150.

GUTIERREZ, J., RODRIGUEZ, G., BARRY-RYAN, C. & BOURKE, P. 2008. Efficacy of plant essential oils against foodborne pathogens and spoilage bacteria associated with ready-to-eat vegetables: antimicrobial and sensory screening. *Journal of Food Protection*, 71, 1846-1854.

HANSEN, M. B., NIELSEN, S. E. & BERG, K. 1989. Re-examination and further development of a precise and rapid dye method for measuring cell growth/cell kill. *Journal of Immunological Methods*, 119, 203-210.

HAO, H., HUI, W., LIU, P., LV, Q., ZENG, X., WANG, Y., ZHENG, X., ZHENG, Y., LI, J. & ZHOU, X. 2013. Effect of licochalcone A on growth and properties of *Streptococcus suis*. *PLoS One*, **8**, e67728.

HAROUN, M. F. & AL-KAYALI, R. S. 2016. Synergistic effect of *Thymbra spicata* L. extracts with antibiotics against multidrug-resistant *Staphylococcus aureus* and *Klebsiella pneumoniae* strains. *Iranian Journal of Basic Medical Sciences*, 19, 1193.

HARRIS, L., FARBER, J., BEUCHAT, L., PARISH, M., SUSLOW, T., GARRETT, E. & BUSTA, F. 2003. Outbreaks associated with fresh produce: incidence, growth, and survival of pathogens in fresh and fresh-cut produce. *Comprehensive Reviews in Food Science and Food Safety*, 2, 78-141.

HARRIS, L. K. & THERIOT, J. A. 2016. Relative rates of surface and volume synthesis set bacterial cell size. *Cell*, 165, 1479-1492.

HARTMANN, C., HIEKE, S., TAPER, C. & SIEGRIST, M. 2018. European consumer healthiness evaluation of 'Free-from'labelled food products. *Food Quality and Preference*, 68, 377-388.

HATANO, T., SHINTANI, Y., AGA, Y., SHIOTA, S., TSUCHIYA, T. & YOSHIDA, T. 2000. Phenolic constituents of licorice. VIII. Structures of glicophenone and glicoisoflavanone, and effects of licorice phenolics on methicillin-resistant *Staphylococcus aureus*. *Chemical and Pharmaceutical Bulletin*, 48, 1286-1292.

HAYRAPETYAN, H., HAZELEGER, W. C. & BEUMER, R. R. 2012. Inhibition of *Listeria monocytogenes* by pomegranate (*Punica granatum*) peel extract in meat paté at different temperatures. *Food Control*, 23, 66-72.

HAZELEGER, W. C., DALVOORDE, M. & BEUMER, R. R. 2006. Fluorescence microscopy of NaCl-stressed, elongated *Salmonella* and *Listeria* cells reveals the presence of septa in filaments. *International Journal of Food Microbiology*, 112, 288-290.

HE, J., CHEN, L., HEBER, D., SHI, W. & LU, Q.-Y. 2006. Antibacterial Compounds from *Glycyrrhiza uralensis*. *Journal of Natural Products*, 69, 121-124.

HENG, L. 2005. Flavour aspects of pea and its protein preparations in relation to novel protein foods. PhD thesis, Wageningen University, Wageningen, The Netherlands.

HEREU, A., BOVER-CID, S., GARRIGA, M. & AYMERICH, T. 2012. High hydrostatic pressure and biopreservation of dry-cured ham to meet the Food Safety Objectives for *Listeria monocytogenes*. *International Journal of Food Microbiology*, 154, 107-112.

HIBASAMI, H., IWASE, H., YOSHIOKA, K. & TAKAHASHI, H. 2006. Glycyrrhetic acid (a metabolic substance and aglycon of glycyrrhizin) induces apoptosis in human hepatoma, promyelotic leukemia and stomach cancer cells. *International Journal of Molecular Medicine*, 17, 215-219.

HILL, P. J. & STEWART, G. S. 1994. Use of *lux* genes in applied biochemistry. *Journal of Bioluminescence and Chemiluminescence*, 9, 211-215.

HINTZ, T., MATTHEWS, K. K. & DI, R. 2015. The use of plant antimicrobial compounds for food preservation. *BioMed research international*, 2015.

HOEFEL, D., GROOBY, W. L., MONIS, P. T., ANDREWS, S. & SAINT, C. P. 2003. Enumeration of water-borne bacteria using viability assays and flow cytometry: a comparison to culture-based techniques. *Journal of Microbiological Methods*, 55, 585-597.

HOF, H. 2003. History and epidemiology of listeriosis. *FEMS Immunology & Medical Microbiology*, 35, 199-202.

HOFFMANN, S., BATZ, M. B. & MORRIS JR, J. G. 2012. Annual cost of illness and quality-adjusted life year losses in the United States due to 14 foodborne pathogens. *Journal of Food Protection*, 75, 1292-1302.

HOLCK, A. L., AXELSSON, L., RODE, T. M., HØY, M., MÅGE, I., ALVSEIKE, O., TRINE, M., OMER, M. K., GRANUM, P. E. & HEIR, E. 2011. Reduction of verotoxigenic *Escherichia coli* in production of fermented sausages. *Meat Science*, 89, 286-295.

HSU, Y. L., WU, L. Y., HOU, M. F., TSAI, E. M., LEE, J. N., LIANG, H. L., JONG, Y. J., HUNG, C. H. & KUO, P. L. 2011. Glabridin, an isoflavan from licorice root, inhibits migration, invasion and angiogenesis of MDA-MB-231 human breast adenocarcinoma cells by inhibiting focal adhesion kinase/Rho signaling pathway. *Molecular Nutrition & Food Research*, 55, 318-327.

IBRAHIM, S. A., SALAMEH, M., PHETSOMPHOU, S., YANG, H. & SEO, C. 2006. Application of caffeine, 1, 3, 7-trimethylxanthine, to control *Escherichia coli* O157: H7. *Food Chemistry*, 99, 645-650.

International Organization for Standardization (ISO). 2012. Water quality - Determination of the genotoxicity of water and waste water-*Salmonella/microsome* fluctuation test (Ames fluctuation test) BS ISO 11350:2012. 1st. ed. BSI Standards Limited.

IRANI, M., SARMADI, M. & BERNARD, F. 2010. Leaves antimicrobial activity of *Glycyrrhiza glabra* L. *Iranian Journal of Pharmaceutical Research*, 9, 425.

ISBRUCKER, R. & BURDOCK, G. 2006. Risk and safety assessment on the consumption of Licorice root (*Glycyrrhiza* sp.), its extract and powder as a food ingredient, with emphasis on the pharmacology and toxicology of glycyrrhizin. *Regulatory Toxicology and Pharmacology*, 46, 167-192.

ISOM, L. L., KHAMBATTA, Z. S., MOLUF, J. L., AKERS, D. F. & MARTIN, S. E. 1995. Filament formation in *Listeria monocytogenes*. *Journal of Food Protection*, 58, 1031-1033.

IVANEK, R., GRÖHN, Y. T., TAUER, L. W. & WIEDMANN, M. 2005. The cost and benefit of *Listeria monocytogenes* food safety measures. *Critical Reviews in Food Science and Nutrition*, 44, 513-523.

JARVIS, N. A. 2016. *Listeria monocytogenes* strain variation: starvation, metabolism, and macrophage survival.

JI, S., LI, Z., SONG, W., WANG, Y., LIANG, W., LI, K., TANG, S., WANG, Q., QIAO, X. & ZHOU, D. 2016. Bioactive constituents of *Glycyrrhiza uralensis* (Licorice): discovery of the effective components of a traditional herbal medicine. *Journal of Natural Products*, 79, 281-292.

JOHNS, C. 2009. Glycyrrhizic acid toxicity caused by consumption of licorice candy cigars. *Canadian Journal of Emergency Medicine*, 11, 94-96.

JONES, G. S. & D'ORAZIO, S. E. 2013. *Listeria monocytogenes*: cultivation and laboratory maintenance. *Current Protocols in Microbiology*, 31, 9B. 2.1-9B. 2.7.

JONES, T. H., VAIL, K. M. & MCMULLEN, L. M. 2013. Filament formation by foodborne bacteria under sublethal stress. *International Journal of Food Microbiology*, 165, 97-110.

JØRGENSEN, F., STEPHENS, P. & KNØCHEL, S. 1995. The effect of osmotic shock and subsequent adaptation on the thermotolerance and cell morphology of *Listeria monocytogenes*. *Journal of Applied Microbiology*, 79, 274-281.

JOUX, F. & LEBARON, P. 1997. Ecological implications of an improved direct viable count method for aquatic bacteria. *Applied and Environmental Microbiology*, 63, 3643-3647.

KALATHENOS, P. & RUSSELL, N. 2003. Ethanol as a food preservative. *Food preservatives.* Springer.

KAMBOJ, V. P. 2000. Herbal medicine. *Current Science*, 78, 35-39.

KARAHAN, F., AVSAR, C., OZYIGIT, I. I. & BERBER, I. 2016. Antimicrobial and antioxidant activities of medicinal plant *Glycyrrhiza glabra* var. glandulifera from different habitats. *Biotechnology & Biotechnological Equipment*, 30, 797-804.

KARAMI, Z., MIRZAEI, H., EMAM-DJOMEH, Z., MAHOONAK, A. S. & KHOMEIRI, M. 2013. Effect of harvest time on antioxidant activity of *Glycyrrhiza glabra* root extract and evaluation of its antibacterial activity. *International Food Research Journal*, 20, 2951.

KASEMETS, K., IVASK, A., DUBOURGUIER, H.-C. & KAHRU, A. 2009. Toxicity of nanoparticles of ZnO, CuO and TiO2 to yeast *Saccharomyces cerevisiae*. *Toxicology in vitro*, 23, 1116-1122.

KELL, D. B., KAPRELYANTS, A. S., WEICHART, D. H., HARWOOD, C. R. & BARER, M. R. 1998. Viability and activity in readily culturable bacteria: a review and discussion of the practical issues. *Antonie Van Leeuwenhoek*, 73, 169-187.

KHAIRUNNISA, K. & KARTHIK, D. 2014. Evaluation of *in-vitro* apoptosis induction, cytotoxic activity of Hymenodictyon excelsum (Roxb) Wall in Dalton's lymphoma
ascites (DLA) and Lung fibroblast-Mouse L929 cell lines. *Journal of Applied Pharmaceutical Science*, 4, 11.

KHAN, M. M. T., PYLE, B. H. & CAMPER, A. K. 2010. Specific and rapid enumeration of viable but nonculturable and viable-culturable Gram-negative bacteria by using flow cytometry. *Applied and Environmental Microbiology*, 76, 5088-5096.

KHAZRAEI-MORADIAN, S., GANJALIKHANI-HAKEMI, M., ANDALIB, A., YAZDANI, R., ARASTEH, J. & KARDAR, G. A. 2017. The effect of licorice protein fractions on proliferation and apoptosis of gastrointestinal cancer cell lines. *Nutrition and Cancer*, 69, 330-339.

KHERALLAH, M. 2002. Bacterial structure and mechanisms of antimicrobial action. *Middle East Critical Care Assembly*.

KIEBOOM, J., KUSUMANINGRUM, H. D., TEMPELAARS, M. H., HAZELEGER, W. C., ABEE, T. & BEUMER, R. R. 2006. Survival, elongation, and elevated tolerance of *Salmonella Enterica* serovar Enteritidis at reduced water activity. *Journal of Food Protection*, 69, 2681-2686.

KIM, H. J., SEO, J.-Y., SUH, H.-J., LIM, S. S. & KIM, J.-S. 2012. Antioxidant activities of licorice-derived prenylflavonoids. *Nutrition Research and Practice*, 6, 491-498.

KIM, H. K., PARK, Y., KIM, H. N., CHOI, B. H., JEONG, H. G., LEE, D. G. & HAHM, K.-S. 2002. Antimicrobial mechanism of β -glycyrrhetinic acid isolated from licorice, *Glycyrrhiza glabra*. *Biotechnology Letters*, 24, 1899-1902.

KITAJIMA, Y. 1998. Introduction: electron microscopy for fungal cell ultrastructure. *Nippon Ishinkin Gakkai Zasshi*, 39, 121-122.

KOBAYASHI, I., MURAOKA, H., SAIKA, T., NISHIDA, M., FUJIOKA, T. & NASU, M. 2004. Micro-broth dilution method with air-dried microplate for determining MICs of clarithromycin and amoxycillin for *Helicobacter pylori* isolates. *Journal of Medical Microbiology*, 53, 403-406.

KOTZEKIDOU, P. 2013. Microbiological examination of ready-to-eat foods and ready-to-bake frozen pastries from university canteens. *Food Microbiology*, 34, 337-343.

KOZŁOWSKA, M., LAUDY, A. E., PRZYBYŁ, J., ZIARNO, M. & MAJEWSKA, E. 2015. Chemical composition and antibacterial activity of some medicinal plants from Lamiaceae family. *Acta Pol Pharm*, 72, 757-67.

KRAUSSE, R., BIELENBERG, J., BLASCHEK, W. & ULLMANN, U. 2004. *In vitro* anti-*Helicobacter pylori* activity of Extractum liquiritiae, glycyrrhizin and its metabolites. *Journal of Antimicrobial Chemotherapy*, 54, 243-246.

KRIS-ETHERTON, P. M., HECKER, K. D., BONANOME, A., COVAL, S. M., BINKOSKI, A. E., HILPERT, K. F., GRIEL, A. E. & ETHERTON, T. D. 2002. Bioactive compounds in foods: their role in the prevention of cardiovascular disease and cancer. *The American Journal of Medicine*, 113, 71-88.

KUMAR, A., KHAN, I. A., KOUL, S., KOUL, J. L., TANEJA, S. C., ALI, I., ALI, F., SHARMA, S., MIRZA, Z. M. & KUMAR, M. 2008. Novel structural analogues of piperine as inhibitors of the NorA efflux pump of *Staphylococcus aureus*. *Journal of Antimicrobial Chemotherapy*, 61, 1270-1276.

KUREK, A., GRUDNIAK, A. M., SZWED, M., KLICKA, A., SAMLUK, L., WOLSKA, K. I., JANISZOWSKA, W. & POPOWSKA, M. 2010. Oleanolic acid and ursolic acid affect peptidoglycan metabolism in *Listeria monocytogenes*. *Antonie Van Leeuwenhoek*, 97, 61.

KUSUMANINGRUM, H. D. 2003. Behaviour and cross-contamination of pathogenic bacteria in household kitchens-relevance to exposure assessment, PhD thesis, Wageningen University, Wageningen.

KWON, S. J., PARK, S. Y., KWON, G. T., LEE, K. W., KANG, Y.-H., CHOI, M.-S., YUN, J. W., JEON, J.-H., JUN, J. G. & PARK, J. H. Y. 2013. Licochalcone E present in licorice suppresses lung metastasis in the 4T1 mammary orthotopic cancer model. *Cancer Prevention Research*.

LANCIOTTI, R., GIANOTTI, A., PATRIGNANI, F., BELLETTI, N., GUERZONI, M. & GARDINI, F. 2004. Use of natural aroma compounds to improve shelf-life and safety of minimally processed fruits. *Trends in Food Science & Technology*, 15, 201-208.

LARA-LLEDÓ, M., OLAIMAT, A. & HOLLEY, R. A. 2012. Inhibition of *Listeria monocytogenes* on bologna sausages by an antimicrobial film containing mustard extract or sinigrin. *International Journal of Food Microbiology*, 156, 25-31.

LARKIN, T. 1983. Herbs are often more toxic than magical. *FDA Consumer. Food* and *Drug Administration (USA)*.

LAU, K., ZAININ, N., ABAS, F. & RUKAYADI, Y. 2014. Antibacterial and sporicidal activity of Eugenia polyantha Wight against *Bacillus cereus* and *Bacillus subtilis*. *International Journal of Current Microbiology and Applied Sciences*, 3, 499-510.

LE LOIR, Y., BARON, F. & GAUTIER, M. 2003. *Staphylococcus aureus* and food poisoning. *Genetics Molecular Research*, 2, 63-76.

LECLERCQ, A., CLERMONT, D., BIZET, C., GRIMONT, P. A., LE FLECHE-MATEOS, A., ROCHE, S. M., BUCHRIESER, C., CADET-DANIEL, V., LE MONNIER, A. & LECUIT, M. 2010. *Listeria rocourtiae* sp. nov. *International Journal of Systematic and Evolutionary Microbiology*, 60, 2210-2214.

LEE, S. K., PARK, K. K., PARK, J. H. Y., LIM, S. S. & CHUNG, W. Y. 2013. The inhibitory effect of roasted licorice extract on human metastatic breast cancer cell-induced bone destruction. *Phytotherapy Research*, 27, 1776-1783.

LEHTINEN, J., JÄRVINEN, S., VIRTA, M. & LILIUS, E.-M. 2006. Real-time monitoring of antimicrobial activity with the multiparameter microplate assay. *Journal of Microbiological Methods*, 66, 381-389.

LEUKO, S., LEGAT, A., FENDRIHAN, S. & STAN-LOTTER, H. 2004. Evaluation of the LIVE/DEAD BacLight kit for detection of extremophilic archaea and visualization of microorganisms in environmental hypersaline samples. *Applied and Environmental Microbiology*, 70, 6884-6886.

LEVIN, D. E., HOLLSTEIN, M., CHRISTMAN, M. F., SCHWIERS, E. A. & AMES, B. N. 1982. A new *Salmonella* tester strain (TA102) with AXT base pairs at the site of mutation detects oxidative mutagens. *Proceedings of the National Academy of Sciences*, 79, 7445-7449.

LEVY, S. 2002. Active efflux, a common mechanism for biocide and antibiotic resistance. *Journal of Applied Microbiology*, 92.

LEWIS, G. P. 2005. *Legumes of the World*, Royal Botanic Gardens Kew.

LI, B. & WEBSTER, T. J. 2018. Bacteria antibiotic resistance: new challenges and opportunities for implant-associated orthopedic infections. *Journal of Orthopaedic Research* ®, 36, 22-32.

LI, G., NIKOLIC, D. & VAN BREEMEN, R. B. 2016. Identification and chemical standardization of licorice raw materials and dietary supplements using UHPLC-MS/MS. *Journal of Agricultural and Food Chemistry*, 64, 8062-8070.

LIANOU, A. & SOFOS, J. N. 2007. A review of the incidence and transmission of *Listeria monocytogenes* in ready-to-eat products in retail and food service environments. *Journal of Food Protection*, 70, 2172-2198.

LIN, J., OPOKU, A., GEHEEB-KELLER, M., HUTCHINGS, A., TERBLANCHE, S., JÄGER, A. K. & VAN STADEN, J. 1999. Preliminary screening of some traditional Zulu medicinal plants for anti-inflammatory and anti-microbial activities. *Journal of Ethnopharmacology*, 68, 267-274.

LIU, X., MILLER, P., BASU, U. & MCMULLEN, L. M. 2014. Sodium chloride-induced filamentation and alternative gene expression of fts, murZ, and gnd in *Listeria monocytogenes* 08-5923 on vacuum-packaged ham. *FEMS Microbiology Letters*, 360, 152-156.

LONG, D. R., MEAD, J., HENDRICKS, J. M., HARDY, M. E. & VOYICH, J. M. 2013. 18β-Glycyrrhetinic acid inhibits methicillin-resistant *Staphylococcus aureus* survival and attenuates virulence gene expression. *Antimicrobial Agents and Chemotherapy*, 57, 241-247.

LORANG, J., TUORI, R., MARTINEZ, J., SAWYER, T., REDMAN, R., ROLLINS, J., WOLPERT, T., JOHNSON, K., RODRIGUEZ, R. & DICKMAN, M. 2001. Green fluorescent protein is lighting up fungal biology. *Applied and Environmental Microbiology*, 67, 1987-1994.

LOWDER, M., UNGE, A., MARAHA, N., JANSSON, J. K., SWIGGETT, J. & OLIVER, J. 2000. Effect of starvation and the viable-but-nonculturable state on green fluorescent protein (GFP) fluorescence in GFP-tagged *Pseudomonas fluorescens* A506. *Applied and Environmental Microbiology*, 66, 3160-3165.

LUBER, P., CRERAR, S., DUFOUR, C., FARBER, J., DATTA, A. & TODD, E. C. 2011. Controlling *Listeria monocytogenes* in ready-to-eat foods: working towards global scientific consensus and harmonization–recommendations for improved prevention and control. *Food Control*, 22, 1535-1549.

LUCERA, A., COSTA, C., CONTE, A. & DEL NOBILE, M. A. 2012. Food applications of natural antimicrobial compounds. *Frontiers in Microbiology*, 3.

LÜDE, S., VECCHIO, S., SINNO-TELLIER, S., DOPTER, A., MUSTONEN, H., VUCINIC, S., JONSSON, B., MÜLLER, D., VERAS GIMENEZ FRUCHTENGARTEN, L. & HRUBY, K. 2016. Adverse effects of plant food supplements and plants consumed as food: results from the poisons centres-based PlantLIBRA study. *Phytotherapy Research*, 30, 988-996.

LUNA, G., MANINI, E. & DANOVARO, R. 2002. Large fraction of dead and inactive bacteria in coastal marine sediments: comparison of protocols for determination and ecological significance. *Applied and Environmental Microbiology*, 68, 3509-3513.

LUNGU, B., RICKE, S. & JOHNSON, M. 2009. Growth, survival, proliferation and pathogenesis of *Listeria monocytogenes* under low oxygen or anaerobic conditions: a review. *Anaerobe*, 15, 7-17.

LUO, L., ZHANG, Z., WANG, H., WANG, P., LAN, R., DENG, J., MIAO, Y., WANG, Y., WANG, Y. & XU, J. 2017. A 12-month longitudinal study of *Listeria monocytogenes* contamination and persistence in pork retail markets in China. *Food Control,* 76, 66-73.

LV, F., LIANG, H., YUAN, Q. & LI, C. 2011. *In vitro* antimicrobial effects and mechanism of action of selected plant essential oil combinations against four food-related microorganisms. *Food Research International*, 44, 3057-3064.

MALEK JAFARIAN, M. & GHAZVINI, K. 2010. *In vitro* susceptibility of *Helicobacter pylori* to licorice extract. *Iranian Journal of Pharmaceutical Research*, 69-72.

MARGOLIN, W. 2000. Themes and variations in prokaryotic cell division. *FEMS Microbiology Reviews,* 24, 531-548.

MARON, D. M. & AMES, B. N. 1983. Revised methods for the *Salmonella* mutagenicity test. *Mutation Research/Environmental Mutagenesis and Related Subjects*, 113, 173-215.

MARTINS, N., BARROS, L., DUENAS, M., SANTOS-BUELGA, C. & FERREIRA, I. C. 2015. Characterization of phenolic compounds and antioxidant properties of *Glycyrrhiza glabra* L. rhizomes and roots. *Royal Society of Chemistry Advances*, 5, 26991-26997.

MASTROMATTEO, M., LUCERA, A., SINIGAGLIA, M. & CORBO, M. R. 2010. Synergic antimicrobial activity of lysozyme, nisin, and EDTA against *Listeria monocytogenes* in ostrich meat patties. *Journal of Food Science*, 75.

MC DERMOTT, P. F., WALKER, R. D. & WHITE, D. G. 2003. Antimicrobials: modes of action and mechanisms of resistance. *International Journal of Toxicology*, 22, 135-143.

MCCANN, J., SPINGARN, N. E., KOBORI, J. & AMES, B. N. 1975. Detection of carcinogens as mutagens: bacterial tester strains with R factor plasmids. *Proceedings of the National Academy of Sciences*, 72, 979-983.

MCLINDEN, T., SARGEANT, J. M., THOMAS, M. K., PAPADOPOULOS, A. & FAZIL, A. 2014. Component costs of foodborne illness: a scoping review. *BMC Public Health*, 14, 509.

MEGHASHRI, S. G. 2009. *In vitro* antifungal and antibacterial activities of root extract of *Glycyrrhiza glabra*. *Journal of Applied Sciences Research*, 1436-1439.

MEIGHEN, E. 1993. Bacterial bioluminescence: organization, regulation, and application of the *lux* genes. *The FASEB Journal*, 7, 1016-1022.

MIETTINEN, H. E., PIIPPO, K., HANNILA-HANDELBERG, T., PAUKKU, K., HILTUNEN, T. P., GAUTSCHI, I., SCHILD, L. & KONTULA, K. 2010. Licorice-induced

hypertension and common variants of genes regulating renal sodium reabsorption. *Annals of Medicine*, 42, 465-474.

MILES, A. A., MISRA, S. & IRWIN, J. 1938. The estimation of the bactericidal power of the blood. *Epidemiology & Infection*, 38, 732-749.

MININEL, F. J., JUNIOR, L., SÉRGIO, C., ESPANHA, L. G., RESENDE, F. A., VARANDA, E. A., LEITE, C. Q. F., VILEGAS, W. & DOS SANTOS, L. C. 2014. Characterization and quantification of compounds in the hydroalcoholic extract of the leaves from *Terminalia catappa* Linn. (Combretaceae) and their mutagenic activity. *Evidence-Based Complementary and Alternative Medicine*, 2014.

MINKOWSKI, P., STAEGE, H., GROSCURTH, P. & SCHAFFNER, A. 2001. Effects of trimethoprim and co-trimoxazole on the morphology of *Listeria monocytogenes* in culture medium and after phagocytosis. *Journal of Antimicrobial Chemotherapy*, 48, 185-193.

MIZUMOTO, A., OHASHI, S., HIROHASHI, K., AMANUMA, Y., MATSUDA, T. & MUTO, M. 2017. Molecular mechanisms of acetaldehyde-mediated carcinogenesis in squamous epithelium. *International Journal of Molecular Sciences*, 18, 1943.

MONTAÑEZ-IZQUIERDO, V. Y., SALAS-VÁZQUEZ, D. I. & RODRÍGUEZ-JEREZ, J. J. 2012. Use of epifluorescence microscopy to assess the effectiveness of phage P100 in controlling *Listeria monocytogenes* biofilms on stainless steel surfaces. *Food Control*, 23, 470-477.

MOOSAVI, M., JALALI, A., SIAHPOOSH, A., KIANIPUR, F. & FARAJZADEH-SHIKH, A. 2013. Assessing Mutagenicity of Methanolic Exteract of *Abrus precatorius* Seeds using Ames Bioassay. *Medical Sciences*, 13, 118-123.

MØRETRØ, T. & LANGSRUD, S. 2017. Residential bacteria on surfaces in the food industry and their implications for food safety and quality. *Comprehensive Reviews in Food Science and Food Safety*, 16, 1022-1041.

MORTELMANS, K. & ZEIGER, E. 2000. The Ames *Salmonella*/microsome mutagenicity assay. *Mutation Research/Fundamental and Molecular Mechanisms of Mutagenesis*, 455, 29-60.

MOTYL, M., DORSO, K., BARRETT, J. & GIACOBBE, R. 2006. Basic microbiological techniques used in antibacterial drug discovery. *Current Protocols in Pharmacology*, 13A. 3.1-13A. 3.22.

MURRAY, E. G. D., WEBB, R. A. & SWANN, M. B. R. 1926. A disease of rabbits characterised by a large mononuclear leucocytosis, caused by a hitherto undescribed bacillus *Bacterium monocytogenes* (n. sp.). *The Journal of Pathology and Bacteriology*, 29, 407-439.

MURRAY, M. & PIZZORNO JR, J. 2000. Botanical medicine-a modern perspective. *Text Book of Natural Medicine*, 1, 267-79.

MUSSARAT, S., ABDEL-SALAM, N. M., TARIQ, A., WAZIR, S. M., ULLAH, R. & ADNAN, M. 2014. Use of ethnomedicinal plants by the people living around Indus River. *Evidence-Based Complementary and Alternative Medicine*, 2014.

NAAS, H., MARTINEZ-DAWSON, R., HAN, I. & DAWSON, P. 2013. Effect of combining nisin with modified atmosphere packaging on inhibition of *Listeria monocytogenes* in ready-to-eat turkey bologna. *Poultry Science*, 92, 1930-1935.

NABAVI, S. M., MARCHESE, A., IZADI, M., CURTI, V., DAGLIA, M. & NABAVI, S. F. 2015. Plants belonging to the genus Thymus as antibacterial agents: from farm to pharmacy. *Food chemistry*, 173, 339-347.

NAIDU, A. 2000. Natural Food Antimicrobial Systems, CRC Press.

NAIR, S. S., PHARANDE, R. R., BANNALIKAR, A. S. & MUKNE, A. P. 2015. *In vitro* antimycobacterial activity of acetone extract of *Glycyrrhiza glabra*. *Journal of Pharmacy & Pharmacognosy Research*, **3**, 80-86.

NAZARI, S., RAMESHRAD, M. & HOSSEINZADEH, H. 2017. Toxicological Effects of *Glycyrrhiza glabra* (Licorice): a review. *Phytotherapy Research*.

NAZZARO, F., FRATIANNI, F., DE MARTINO, L., COPPOLA, R. & DE FEO, V. 2013. Effect of essential oils on pathogenic bacteria. *Pharmaceuticals*, 6, 1451-1474.

NEGI, P. S. 2012. Plant extracts for the control of bacterial growth: efficacy, stability and safety issues for food application. *International Journal of Food Microbiology*, 156, 7-17.

NESSLANY, F., SIMAR-MEINTIÈRES, S., FICHEUX, H. & MARZIN, D. 2009. Aloeemodin-induced DNA fragmentation in the mouse *in vivo* comet assay. *Mutation Research/Genetic Toxicology and Environmental Mutagenesis*, 678, 13-19.

NIMMO, G. R., BELL, J. M., MITCHELL, D., GOSBELL, I. B., PEARMAN, J. W. & TURNIDGE, J. D. 2003. Antimicrobial resistance in *Staphylococcus aureus* in Australian teaching hospitals, 1989-1999. *Microbial Drug Resistance*, 9, 155-160.

NIRMALA, P. & SELVARAJ, T. 2011. Anti-inflammatory and anti-bacterial activities of *Glycyrrhiza glabra* L. *Journal of Agricultural Technology*, 7, 815-823.

NOCKER, A., CASPERS, M., ESVELD-AMANATIDOU, A., VAN DER VOSSEN, J., SCHUREN, F., MONTIJN, R. & KORT, R. 2011. A multiparameter viability assay for stress profiling applied to the food pathogen *Listeria monocytogenes* F2365. *Applied and Environmental Microbiology*, AEM. 00142-11.

NØRRUNG, B., ANDERSEN, J. K. & BUNCIC, S. 2009. Main concerns of pathogenic microorganisms in meat. *Safety of Meat and Processed Meat.* Springer.

NOWAKOWSKA, Z. 2007. A review of anti-infective and anti-inflammatory chalcones. *European Journal of Medicinal Chemistry*, 42, 125-137.

NÚÑEZ-MONTERO, K., LECLERCQ, A., MOURA, A., VALES, G., PERAZA, J., PIZARRO-CERDÁ, J. & LECUIT, M. 2018. *Listeria costaricensis* sp. nov. *International Journal of Systematic and Evolutionary Microbiology*.

NYSTRÖM, T. 2001. Not quite dead enough: on bacterial life, culturability, senescence, and death. *Archives of Microbiology*, 176, 159-164.

OKEKE, I. N., LAXMINARAYAN, R., BHUTTA, Z. A., DUSE, A. G., JENKINS, P., O'BRIEN, T. F., PABLOS-MENDEZ, A. & KLUGMAN, K. P. 2005. Antimicrobial resistance in developing countries. Part I: recent trends and current status. *The Lancet Infectious Diseases*, *5*, 481-493.

OLSZEWSKA, M. A., PANFIL-KUNCEWICZ, H. & ŁANIEWSKA-TROKENHEIM, Ł. 2015. Detection of viable but nonculturable cells of *Listeria monocytogenes* with the use of direct epifluorescent filter technique. *Journal of Food Safety*, 35, 86-90.

ORGANIZATION, W. H. WHO estimates of the global burden of foodborne diseases: foodborne disease burden epidemiology reference group 2007-2015.

ORSI, R. H. & WIEDMANN, M. 2016. Characteristics and distribution of *Listeria* spp., including *Listeria* species newly described since 2009. *Applied Microbiology and Biotechnology*, 100, 5273-5287.

PAINTER, J. & SLUTSKER, L. 2007. Listeriosis in humans. *Food Science and Technology-New York-Marcel Dekker*, 161, 85.

PAPARELLA, A., SERIO, A. & LÓPEZ, C. C. 2012. Flow cytometry applications in food safety studies. *Flow Cytometry-Recent Perspectives.* InTech.

PAREKH, J. & CHANDA, S. 2007. Antibacterial and phytochemical studies on twelve species of Indian medicinal plants. *African Journal of Biomedical Research*, 10.

PAREKH, J., JADEJA, D. & CHANDA, S. 2006. Efficacy of aqueous and methanol extracts of some medicinal plants for potential antibacterial activity. *Turkish Journal of Biology*, 29, 203-210.

PATEL, S., NAG, M. K., DAHARWAL, S., SINGH, M. R. & SINGH, D. 2013. Plant toxins: an overview. *Research journal of Pharmacology and Pharmacodynamics*, **5**, II.

PEREHINEC, T. M., QAZI, S. N., GADDIPATI, S. R., SALISBURY, V., REES, C. E. & HILL, P. J. 2007. Construction and evaluation of multisite recombinatorial (Gateway) cloning vectors for Gram-positive bacteria. *BMC Molecular Biology*, 8, 80

PIDDOCK, L. J. 2006. Clinically relevant chromosomally encoded multidrug resistance efflux pumps in bacteria. *Clinical Microbiology Reviews*, 19, 382-402.

PIENAAR, J. A., SINGH, A. & BARNARD, T. G. 2016. The viable but non-culturable state in pathogenic *Escherichia coli*: a general review. *African Journal of Laboratory Medicine*, 5, 1-9.

PIKTEL, E., POGODA, K., ROMAN, M., NIEMIROWICZ, K., TOKAJUK, G., WRÓBLEWSKA, M., SZYNAKA, B., KWIATEK, W. M., SAVAGE, P. B. & BUCKI, R. 2017. Sporicidal activity of ceragenin CSA-13 against *Bacillus subtilis*. *Scientific Reports*, 7, 44452.

POOLE, K. 2004. Efflux-mediated multiresistance in Gram-negative bacteria. *Clinical Microbiology and Infection*, 10, 12-26.

PRASCH, S. & BUCAR, F. 2015. Plant derived inhibitors of bacterial efflux pumps: an update. *Phytochemistry Reviews*, 14, 961-974.

PRATAP CHANDRAN, R. 2015. Antimycobacterial activity of *Glycyrrhiza Glabra* Linn. Root extract against *Mycobacterium smegmatis*. *International Journal of PharmTech Research*, 7.

PROBES, M. 2004. LIVE/DEAD BacLight bacterial viability kits. *Product Information Sheet, Molecular Probes*, 7007, 05-15.

PROESTOS, C., BOZIARIS, I. S., KAPSOKEFALOU, M. & KOMAITIS, M. 2008. Natural antioxidant constituents from selected aromatic plants and their antimicrobial activity against selected pathogenic microorganisms. *Food Technology and Biotechnology*, 46, 151-156. Public Health England (PHE). 2018. Listeriosis in England and Wales Summary for 2017. Data from the national enhanced surveillance system for *Listeria monocytogenes*.

PUUPPONEN-PIMIÄ, R., NOHYNEK, L., MEIER, C., KÄHKÖNEN, M., HEINONEN, M., HOPIA, A. & OKSMAN-CALDENTEY, K. M. 2001. Antimicrobial properties of phenolic compounds from berries. *Journal of Applied Microbiology*, 90, 494-507.

PYLE, B. H., BROADAWAY, S. C. & MCFETERS, G. A. 1999. Sensitive detection of *Escherichia coli* O157: H7 in food and water by immunomagnetic separation and solid-phase laser cytometry. *Applied and Environmental Microbiology*, 65, 1966-1972.

QAZI, S., REES, C., MELLITS, K. & HILL, P. 2001. Development of gfp vectors for expression in *Listeria monocytogenes* and other low G+ C Gram-positive bacteria. *Microbial Ecology*, 41, 301-309.

RAGHU, R. 2013. *Listeria monocytogenes*: an interesting pathogen. *Microbiology Focus*, **5**, 1-2.

RAI, V. R. & BAI, J. A. 2014. *Microbial food safety and preservation techniques*, CRC Press.

RAO, S., TIMSINA, B. & NADUMANE, V. K. 2014. Antimicrobial effects of medicinal plants and their comparative cytotoxic effects on HEPG2 cell line. *International Journal of Pharmacy and Pharmaceutical Sciences*, 6, 101-105.

RASHID, M. M. U., KABIR, H., SAYEED, M. A., ALAM, R. & KABIR, M. F. 2013. Effect of flavonoid extract of the medicinal plant (*Glycyrrhiza glabra* L.) in the region of Djamaa (south of Algeria) on the growth of some human pathogenic bacteria. *Journal of Pharmacognosy and Phytochemistry*, 2.

RAYBAUDI-MASSILIA, R. M., MOSQUEDA-MELGAR, J., SOLIVA-FORTUNY, R. & MARTÍN-BELLOSO, O. 2009. Control of pathogenic and spoilage microorganisms in fresh-cut fruits and fruit juices by traditional and alternative natural antimicrobials. *Comprehensive Reviews in Food Science and Food Safety*, 8, 157-180.

REES, C., DOYLE, L. & TAYLOR, C. (2017). *Listeria monocytogenes*. In Foodborne Diseases. 253-276. (3rd). Elsevier.

REZK, A., AL-HASHIMI, A., JOHN, W., SCHEPKER, H., ULLRICH, M. S. & BRIX, K. 2015. Assessment of cytotoxicity exerted by leaf extracts from plants of the genus *Rhododendron* towards epidermal keratinocytes and intestine epithelial cells. *BMC Complementary and Alternative Medicine*, 15, 364.

RIETJENS, I. M., BOERSMA, M. G., VAN DER WOUDE, H., JEURISSEN, S. M., SCHUTTE, M. E. & ALINK, G. M. 2005. Flavonoids and alkenylbenzenes: mechanisms of mutagenic action and carcinogenic risk. *Mutation Research/Fundamental and Molecular Mechanisms of Mutagenesis*, 574, 124-138.

RISS, T. L., MORAVEC, R. A. & NILES, A. L. 2011. Cytotoxicity testing: measuring viable cells, dead cells, and detecting mechanism of cell death. *Mammalian Cell Viability*. Springer.

RISS, T. L., MORAVEC, R. A., NILES, A. L., DUELLMAN, S., BENINK, H. A., WORZELLA, T. J. & MINOR, L. 2016. Cell viability assays.

ROBINSON, G. M., TONKS, K. M., THORN, R. M. & REYNOLDS, D. M. 2011. Application of bacterial bioluminescence to assess the efficacy of fast-acting biocides. *Antimicrobial Agents and Chemotherapy*, 55, 5214-5219.

RODEIRO, I., CANCINO, L., GONZÁLEZ, J., MORFFI, J., GARRIDO, G., GONZÁLEZ, R., NUÑEZ, A. & DELGADO, R. 2006. Evaluation of the genotoxic potential of *Mangifera indica* L. extract (Vimang), a new natural product with antioxidant activity. *Food and Chemical Toxicology*, 44, 1707-1713.

RODINO, S., BUTU, A., BUTU, M. & CORNEA, P. 2015. Comparative studies on antibacterial activity of licorice, elderberry and dandelion. *Digest Journal of Nanomaterials and Biostructures*, 10, 947-955.

RODRIGUES, C. S., SÁ, C. V. G. C. D. & MELO, C. B. D. 2017. An overview of *Listeria monocytogenes* contamination in ready to eat meat, dairy and fishery foods. *Ciência Rural*, 47.

ROMAN, S., SANCHEZ-SILES, L. M. & SIEGRIST, M. 2017. The importance of food naturalness for consumers: results of a systematic review. *Trends in Food Science & Technology*, 67, 44-57.

ROSHAN, A., VERMA, N. K., KUMAR, C. S., CHANDRA, V., SINGH, D. P. & PANDAY, M. 2012. Phytochemical constituent, pharmacological activities and medicinal uses through the millenia of *Glycyrrhiza glabra* Linn: a review. *International Research Journal of Pharmacy*, **3**, 45-55.

ROSZAK, D. & COLWELL, R. 1987. Survival strategies of bacteria in the natural environment. *Microbiological Reviews*, 51, 365.

ROWAN, N. J. & ANDERSON, J. G. 1998. Effects of above-optimum growth temperature and cell morphology on thermotolerance of *Listeria monocytogenes* cells suspended in bovine milk. *Applied and Environmental Microbiology*, 64, 2065-2071.

RUUSUNEN, M. & PUOLANNE, E. 2005. Reducing sodium intake from meat products. *Meat Science*, 70, 531-541.

RYSER, E. T. & MARTH, E. H. 2007. Ecology of *Listeria* species and *L. monocytogenes* in the natural environment. *Listeria, Listeriosis, and Food Safety, Third Edition.* CRC Press.

SAXE, T. 1987. Toxicity of medicinal herbal preparations. *American Family Physician*, 35, 135-142.

SAXENA, S. 2005. *Glycyrrhiza glabra*: medicine over the millennium. *Natural Product Radiance*, 4(5):358–67

SCALLAN, E., HOEKSTRA, R. M., ANGULO, F. J., TAUXE, R. V., WIDDOWSON, M.-A., ROY, S. L., JONES, J. L. & GRIFFIN, P. M. 2011. Foodborne illness acquired in the United States—major pathogens. *Emerging Infectious Diseases*, 17, 7.

SEDIGHINIA, F. & AFSHAR, A. S. 2012. Antibacterial activity of *Glycyrrhiza glabra* against oral pathogens: an *in vitro* study. *Avicenna Journal of Phytomedicine*, 2, 118.

SHAKYA, A. K. 2016. Medicinal plants: future source of new drugs. *International Journal of Herbal Medicine*, 4, 59-64.

SHAN, B., CAI, Y.-Z., BROOKS, J. D. & CORKE, H. 2007. The *in vitro* antibacterial activity of dietary spice and medicinal herb extracts. *International Journal of Food Microbiology*, 117, 112-119.

SHEELA, M., RAMAKRISHNA, M. & SALIMATH, B. P. 2006. Angiogenic and proliferative effects of the cytokine VEGF in Ehrlich ascites tumor cells is inhibited by *Glycyrrhiza glabra*. *International Immunopharmacology*, 6, 494-498.

SHEN, H., ZENG, G., SUN, B., CAI, X., BI, L., TANG, G. & YANG, Y. 2015. A polysaccharide from *Glycyrrhiza inflata* Licorice inhibits proliferation of human oral cancer cells by inducing apoptosis via mitochondrial pathway. *Tumor Biology*, 36, 4825-4831.

SHIBATA, S. 2000. A drug over the millennia: pharmacognosy, chemistry, and pharmacology of licorice. *Yakugaku Zasshi*, 120, 849-862.

SHIRAZI, M., RANJBAR, R., ESHRAGHI, S., SADEGHI, G., JONAIDI, N., BAZZAZ, N., IZADI, M. & SADEGHIFARD, N. 2007. An evaluation of antibacterial activity of *Glycyrrhiza glabra* extract on the growth of *Salmonella*, *Shigella* and ETEC *E. coli*. *Journal of Biological Sciences*, 7.

SILVA-ANGULO, A. B., ZANINI, S. F., ROSENTHAL, A., RODRIGO, D., KLEIN, G. & MARTÍNEZ, A. 2015. Comparative study of the effects of citral on the growth and injury of *Listeria innocua* and *Listeria monocytogenes* cells. *PloS One*, 10, (2), e0114026.

SIMÕES, M., LEMOS, M. & SIMÕES, L. C. 2012. Phytochemicals against drugresistant microbes. *Dietary Phytochemicals and Microbes*. Springer.

SINGH, S. & SHALINI, R. 2016. Effect of hurdle technology in food preservation: a review. *Critical Reviews in Food Science and Nutrition*, 56, 641-649.

ŠKRINJAR, M. M. & NEMET, N. T. 2009. Antimicrobial effects of spices and herbs essential oils. *Acta Periodica Technologica*, 195-209.

SOFOS, J. N. & GEORNARAS, I. 2010. Overview of current meat hygiene and safety risks and summary of recent studies on biofilms, and control of *Escherichia coli* 0157: H7 in nonintact, and *Listeria monocytogenes* in ready-to-eat, meat products. *Meat Science*, 86, 2-14.

SOMOLINOS, M., GARCÍA, D., CONDÓN, S., MACKEY, B. & PAGÁN, R. 2010. Inactivation of *Escherichia coli* by citral. *Journal of Applied Microbiology*, 108, 1928-1939.

SOSA, A. D. J., AMÁBILE-CUEVAS, C. F., BYARUGABA, D. K., HSUEH, P.-R., KARIUKI, S. & OKEKE, I. N. 2010. *Antimicrobial Resistance in Developing Countries*, Springer.

SOULEF, K., ABDELOUAHAB, Y. & DALAL, B. 2014. Effect of glycosides extract of the medicinal plant *Glycyrrhiza glabra* L. from the region of Mlilli (southeast of Algeria) on the growth of some human pathogenic bacteria. *Journal of Science Innovations*, **3**, 28-34.

SRINIVASAN, D., NATHAN, S., SURESH, T. & PERUMALSAMY, P. L. 2001. Antimicrobial activity of certain Indian medicinal plants used in folkloric medicine. *Journal of Ethnopharmacology*, 74, 217-220. SRUJANA, T. S., KONDURI, R. B. & RAO, B. S. S. 2012. Phytochemical investigation and biological activity of leaves extract of plant *Boswellia serrata*. *The Pharma Innovation*, 1, 22.

STATTI, G. A., TUNDIS, R., SACCHETTI, G., MUZZOLI, M., BIANCHI, A. & MENICHINI, F. 2004. Variability in the content of active constituents and biological activity of *Glycyrrhiza glabra*. *Fitoterapia*, 75, 371-374.

STEVENSON, K., MCVEY, A. F., CLARK, I. B., SWAIN, P. S. & PILIZOTA, T. 2016. General calibration of microbial growth in microplate readers. *Scientific Reports*, 6, 38828.

STOCKS, S. 2004. Mechanism and use of the commercially available viability stain, BacLight. *Cytometry Part A: The Journal of the International Society for Analytical Cytology*, 61, 189-195.

SULTANA, S., HAQUE, A., HAMID, K., URMI, K. F. & ROY, S. 2010. Antimicrobial, cytotoxic and antioxidant activity of methanolic extract of *Glycyrrhiza glabra*. *Agriculture and Biology Journal of North America*, 1, 957-960.

SWAMINATHAN, B. & GERNER-SMIDT, P. 2007. The epidemiology of human listeriosis. *Microbes and Infection*, 9, 1236-1243.

SWAMY, M. K., AKHTAR, M. S. & SINNIAH, U. R. 2016. Antimicrobial properties of plant essential oils against human pathogens and their mode of action: an updated review. *Evidence-Based Complementary and Alternative Medicine*, 2016.

TAJKARIMI, M., IBRAHIM, S. A. & CLIVER, D. 2010. Antimicrobial herb and spice compounds in food. *Food Control*, 21, 1199-1218.

TANAKA, M. M., KENDAL, J. R. & LALAND, K. N. 2009. From traditional medicine to witchcraft: why medical treatments are not always efficacious. *PLoS One*, 4, e5192.

TANEMOTO, R., OKUYAMA, T., MATSUO, H., OKUMURA, T., IKEYA, Y. & NISHIZAWA, M. 2015. The constituents of licorice (*Glycyrrhiza uralensis*) differentially suppress nitric oxide production in interleukin-1β-treated hepatocytes. *Biochemistry and Biophysics Reports*, 2, 153-159.

TEGOS, G., STERMITZ, F. R., LOMOVSKAYA, O. & LEWIS, K. 2002. Multidrug pump inhibitors uncover remarkable activity of plant antimicrobials. *Antimicrobial Agents and Chemotherapy*, 46, 3133-3141.

TEJS, S. 2008. The Ames test: a methodological short review. *Environmental Biotechnology*, 4, 7-14.

THEINSATHID, P., VISESSANGUAN, W., KRUENATE, J., KINGCHA, Y. & KEERATIPIBUL, S. 2012. Antimicrobial activity of lauric arginate-coated polylactic acid films against *Listeria monocytogenes* and *Salmonella* Typhimurium on cooked sliced ham. *Journal of Food Science*, 77, M142-M149.

THOMAS, M. K., VRIEZEN, R., FARBER, J. M., CURRIE, A., SCHLECH, W. & FAZIL, A. 2015. Economic cost of a *Listeria monocytogenes* outbreak in Canada, 2008. *Foodborne Pathogens and Disease*, 12, 966-971.

TIWARI, B. K., VALDRAMIDIS, V. P., O'DONNELL, C. P., MUTHUKUMARAPPAN, K., BOURKE, P. & CULLEN, P. 2009. Application of natural antimicrobials for food preservation. *Journal of Agricultural and Food Chemistry*, 57, 5987-6000.

TOHMA, H. S. & GULÇIN, I. 2010. Antioxidant and radical scavenging activity of aerial parts and roots of Turkish liquorice (*Glycyrrhiza glabra* L.). *International Journal of Food Properties*, 13, 657-671.

TSUKIYAMA, R.-I., KATSURA, H., TOKURIKI, N. & KOBAYASHI, M. 2002. Antibacterial activity of licochalcone A against spore-forming bacteria. *Antimicrobial Agents and Chemotherapy*, 46, 1226-1230.

UNGE, A., TOMBOLINI, R., MØLBAK, L. & JANSSON, J. K. 1999. Simultaneous monitoring of cell number and metabolic activity of specific bacterial populations with a dualgfp-luxAB marker system. *Applied and Environmental Microbiology*, 65, 813-821.

UPADHYAY, A., UPADHYAYA, I., KOLLANOOR-JOHNY, A. & VENKITANARAYANAN, K. 2014. Combating pathogenic microorganisms using plant-derived antimicrobials: a mini review of the mechanistic basis. *BioMed Research International*, 2014.

VAN DE VENTER, T. 2000. Emerging food-borne diseases: a global responsibility. *Food Nutrition and Agriculture*, 4-13.

VATANYOOPAISARN, S., NAZLI, A., DODD, C. E., REES, C. E. & WAITES, W. M. 2000. Effect of flagella on initial attachment of *Listeria monocytogenes* to stainless steel. *Applied and Environmental Microbiology*, 66, 860-863.

VÁZQUEZ-BOLAND, J. A., KUHN, M., BERCHE, P., CHAKRABORTY, T., DOMINGUEZ-BERNAL, G., GOEBEL, W., GONZÁLEZ-ZORN, B., WEHLAND, J. & KREFT, J. 2001. *Listeria* pathogenesis and molecular virulence determinants. *Clinical Microbiology Reviews*, 14, 584-640.

VEDMAURTHY, R. B., PADMANABHAN, S., VIJAYAN, M., JAMAL, Z., KUNJUMMAN, J. & NARAYANAN, M. L. 2012. Compatibility of different solvents with *Salmonella* Typhimurium mutant strains in bacterial reverse mutation assay. *International Journal of Pharmacy and Pharmaceutical Sciences*, **4**, 283-284.

VERSCHAEVE, L. & VAN STADEN, J. 2008. Mutagenic and antimutagenic properties of extracts from South African traditional medicinal plants. *Journal of Ethnopharmacology*, 119, 575-587.

VIBHA, J., CHOUDHARY, K., SINGH, M., RATHORE, M. & SHEKHAWAT, N. 2009. A study on pharmacokinetics and therapeutic efficacy of *Glycyrrhiza glabra*: a miracle medicinal herb. *Botany Research International*, 2, 157-163.

VIUDA-MARTOS, M., RUIZ-NAVAJAS, Y., FERNÁNDEZ-LÓPEZ, J. & PÉREZ-ÁLVAREZ, J. 2010. Effect of added citrus fibre and spice essential oils on quality characteristics and shelf-life of mortadella. *Meat Science*, 85, 568-576.

WANG, Y.-J., ZHOU, S.-M., XU, G. & GAO, Y.-Q. 2015. Interference of Phenylethanoid Glycosides from Cistanche tubulosa with the MTT Assay. *Molecules*, 20, 8060-8071.

WANG, Z. Y. & NIXON, D. W. 2001. Licorice and cancer. *Nutrition and Cancer*, 39, 1-11.

WAY, S. S., THOMPSON, L. J., LOPES, J. E., HAJJAR, A. M., KOLLMANN, T. R., FREITAG, N. E. & WILSON, C. B. 2004. Characterization of flagellin expression and its role in *Listeria monocytogenes* infection and immunity. *Cellular Microbiology*, 6, 235-242.

WELLER, D., ANDRUS, A., WIEDMANN, M. & DEN BAKKER, H. C. 2015. *Listeria booriae* sp. nov. and *Listeria newyorkensis* sp. nov., from food processing environments in the USA. *International Journal of Systematic and Evolutionary Microbiology*, 65, 286-292.

WESSNER, D. R., MAIORANO, P. C., KENYON, J., PILLSBURY, R. & CAMPBELL, A. M. 2000. Spot-overlay Ames test of potential mutagens. *Association for Biology Laboratory Education*, 22, 1-18.

WHITE, P. C. 2001. 11β -hydroxysteroid dehydrogenase and its role in the syndrome of apparent mineralocorticoid excess. *The American Journal of the Medical Sciences*, 322, 308-315.

WORLD HEALTH ORGANIZATION (WHO).2007. Food safety & food-borne illness. Fact sheet no. 237.

WORLD HEALTH ORGANIZATION (WHO). 2018. *Listeriosis – Australia*. [online] Available at: https://www.who.int/csr/don/09-april-2018-listeriosis-australia/en/ [Accessed 28 Mar. 2019].

WORLD HEALTH ORGANIZATION (WHO). 2018. *Listeriosis – South Africa*. [online] Available at: https://www.who.int/csr/don/02-may-2018-listeriosis-south-africa/en/ [Accessed 26 Mar. 2019].

WORLDHEALTHORGANIZATION.Water,SanitationandHealthTeam. 2004. Water,sanitationandhygienelinkstohealth:factsandfigures. Geneva:[online]Availableat:http://www.who.int/iris/handle/10665/69489[Accessed 26 Mar. 2019].

WIDDEL, F. 2007. Theory and measurement of bacterial growth. *Di dalam Grundpraktikum Mikrobiologie*, 4, 1-11.

WINK, M. 2013. Evolution of secondary metabolites in legumes (Fabaceae). *South African Journal of Botany*, 89, 164-175.

WINK, M. 2018. Plant secondary metabolites modulate insect behavior-steps toward addiction? *Frontiers in physiology*, 9, 364.

WTO International trade statistics 2015. Geneva: World Trade Organization.

WU, Y., BAI, J., ZHONG, K., HUANG, Y., QI, H., JIANG, Y. & GAO, H. 2016. Antibacterial activity and membrane-disruptive mechanism of 3-p-transcoumaroyl-2-hydroxyquinic acid, a novel phenolic compound from pine needles of cedrus deodara, against *Staphylococcus aureus*. *Molecules*, 21, 1084.

XIONG, Y. Q., WILLARD, J., KADURUGAMUWA, J. L., YU, J., FRANCIS, K. P. & BAYER, A. S. 2005. Real-time *in vivo* bioluminescent imaging for evaluating the efficacy of antibiotics in a rat *Staphylococcus aureus* endocarditis model. *Antimicrobial agents and chemotherapy*, 49, 380-387.

XUESHENG, Y., JIANWEI, Z., BO, J. & FEI, W. 2009. Analysis on strychnine and brucine in Strychnos nux-vomical L. uncombined and combined with *Glycyrrhiza Uralensis* Fisch [J]. *Asia-Pacific Traditional Medicine*, 10, 009.

YAMAMURA, Y., KAWAKAMI, J., SANTA, T., KOTAKI, H., UCHINO, K., SAWADA, Y., TANAKA, N. & IGA, T. 1992. Pharmacokinetic profile of glycyrrhizin in healthy volunteers by a new high-performance liquid chromatographic method. *Journal of Pharmaceutical Sciences*, 81, 1042-1046.

YANG, L., WU, L., ZHU, S., LONG, Y., HANG, W. & YAN, X. 2009. Rapid, absolute, and simultaneous quantification of specific pathogenic strain and total bacterial cells using an ultrasensitive dual-color flow cytometer. *Analytical Chemistry*, 82, 1109-1116.

YANG, R., WANG, L.-Q. & LIU, Y. 2014. Antitumor activities of widely-used Chinese herb—licorice. *Chinese Herbal Medicines*, 6, 274-281.

YAP, P. S. X., YIAP, B. C., PING, H. C. & LIM, S. H. E. 2014. Essential oils, a new horizon in combating bacterial antibiotic resistance. *The Open Microbiology Journal*, 8, 6.

YOKOMAKU, D., YAMAGUCHI, N. & NASU, M. 2000. Improved direct viable count procedure for quantitative estimation of bacterial viability in freshwater environments. *Applied and Environmental Microbiology*, 66, 5544-5548.

YOON, G., DO JUNG, Y. & CHEON, S. H. 2005. Cytotoxic allyl retrochalcone from the roots of *Glycyrrhiza inflata*. *Chemical and Pharmaceutical Bulletin*, 53, 694-695.

ZAREI, M., KHEZIZADEH, M., KAZEMIPOUR, S., HESAMI, G. & BEMANI, E. 2012. Growth and cell morphology of *Listeria monocytogenes* as affected by various concentrations of NaCl and KCl. *Journal of Applied Biological Sciences*, 6, 55-58.

ZEIGER, E. 2001. Mutagens that are not carcinogens: faulty theory or faulty tests? *Mutation Research/Genetic Toxicology and Environmental Mutagenesis*, 492, 29-38.

ZHANG, G., MEREDITH, T. C. & KAHNE, D. 2013. On the essentiality of lipopolysaccharide to Gram-negative bacteria. Current opinion in microbiology, 16, 779-785.

ZHANG, H., KONG, B., XIONG, Y. L. & SUN, X. 2009. Antimicrobial activities of spice extracts against pathogenic and spoilage bacteria in modified atmosphere packaged fresh pork and vacuum packaged ham slices stored at 4 C. *Meat Science*, 81, 686-692.

ZHANG, Q. & YE, M. 2009. Chemical analysis of the Chinese herbal medicine Gan-Cao (licorice). *Journal of Chromatography A*, 1216, 1954-1969.

ZHANG, W. & KNABEL, S. J. 2005. Multiplex PCR assay simplifies serotyping and sequence typing of *Listeria monocytogenes* associated with human outbreaks. *Journal of Food Protection*, 68, 1907-1910.

ZHU, Q., GOONERATNE, R. & HUSSAIN, M. A. 2017. *Listeria monocytogenes* in fresh produce: outbreaks, prevalence and contamination levels. *Foods*, 6, 21.

ZIMMER, M. 2002. Green fluorescent protein (GFP): applications, structure, and related photophysical behavior. *Chemical Reviews*, 102, 759-782.

APPENDICES

APPENDIX 1: Certificate of Analysis of the liquorice extract

1.1 Analysis of the liquorice extract

According to the analysis provided by the company who produced the material (Phytoquest Ltd), the active constituent in the waste material was generated from *Glycyrrhiza uralensis* Root during the production of liquorice extracts for food production contains Licoricidin and Licorisoflavan A (see Figure 1.1 and 1.2). While this had been evaluated as a potential antioxidant material, it had not been assessed as an antimicrobial.

Figure 1.1 Certificate of Analysis of the liquorice extract



Certificate of Analysis

Licorice root CO2-to extract with MCT Oil, type no. 175.004

Variety Uralensis

batch no. 841106, lab no. 16646

production: May 2014 retest: May 2019

Glycyrrhiza uralensis - Root, dried raw material: country of origin of raw material: China country of origin of product: Germany

Certification:



Sensory Check

feature	reference	result
Appearance: Odour:	brown, viscous extract at room temperature light sweet flavour	meets

Analytical Check

feature	method	limits	value	unit		
Licoricidin	21.178.01, HPLC	n.s.	6,8	%		
Licorisoflavan A	21.178.01, HPLC	n.s.	2,2	%		
Sum of identified Isoflavans	21.178.01, HPLC	> = 8,5	9,0	%		
Content of alcohol (Ethanol)	21.025.02, GCFID	2,0 - 4,0	2,3	%		
		n.s. = not specified	n.d	. = not d	letec	ted

Digitally signed by Anja Cawelius Reason: Quality control Date: 2014.05.12 11:03:04 +02'00'

The product meets specifications no. 16.521.01/66.498.01; date of analysis: 2014.05.12 This computerized CoA has digital signature validated by FLAVEX QC.

The data in this report of analysis have been determined carefully and to the best of our knowledge. Depending on transport and storage conditions the indicated data can be subject to certain changes which are outside of our influence. Hence the report has not the meaning of a guaranty in the legal sense and does not dispense the customer from making his own quality control before using the product.



Page 1 of 1.







217

E-Mail: info@flavex.com http://www.flavex.com

Figure 1.2 Certificate of Analysis of the liquorice extract

	Licorice root CO2-to extract with MCT Oil Variety Uralensis, Type no. 175.004
Raw material:	Glycyrrhiza uralensis - Root, dried
Production:	By supercritical fluid extraction with natural earbon dioxide and a small amount of ethanol as entrainer, no inorganic salts, no heavy metals, no reproducible microorganisms [1]. The extract is standardised with MCT oil to a content of NLT 8,5 % isoflavanes.
Description:	Contains all CO2-soluble lipohilic and semipolar components. Brown, viscous extract at room temperature with light sweet flavour
Declaration:	NCI-Name (CTFA): Glycyrrhiza Uralensis (Licorice) Root Extract, CAS-No. 94349-91-4, EINECS-No. 305-209-3 and Caprylic/Capric Triglyceride, CAS-No. 73398-61-5, EINECS-No. 277-452-2
Halal - Status:	not certified
Kosher - Status:	Certified by London Beth Din Kashrut Division (KLBD). The certification is marked with the Kosher logo on the product.
Transport:	No dangerous good in the sense of the transport regulations.
Ingredients:	The extract contains NLT 8,5 % isoflavanes, e.g. licoricidin and licorisoflavan A and a small amount of essential oil, but no glycyrrhizic acid, 2 - 4 % ethanol.
Application:	Due to the ingredients the extract has antimicrobial, antioxidant and anti-inflammatory properties. It is used in cosmetics and supplements.
Naturalness:	The product is manufactured from the named raw material. It contains apart from the MCT oil no additives and no other technical adjuncts. The product is 100 % natural and corresponds to the EC Flavouring Regulation No. 1334/2008 for flavouring preparations.
Stability:	Unopened container under cool and dry storage conditions and exclusion of light at least 5 years.
References: [1] Mar	nninen P., Haivälä E., Sarimo S., Kallio H. : Z. Lebens Unters Forsch A (1997) 204: 202-205

Disclaimer: These data are given for customer's information only to the best of our knowledge but under exemption of liability especially regarding infringement of or prejudice to third party rights through the use of this product. This information is not a substitute for prior tests demonstrating the suitability of the product for the intended use. Users are responsible for ensuring the compliance with the applicable legislation. The concentrated FLAVEX extracts are raw materials for product formulation. Hence they are not intended for direct consumption in food and for undiluted topical application in cosmetics, perfumery and aromatherapy. Keep away from children.



APPENDIX 2: PREPARATION OF MEDIA

2.1 Media preparation

The suppliers for all chemicals substances and media used in this thesis are listed in Table 2.1. All media used in this thesis were prepared using reverse osmosis (RO) water followed by sterilization by autoclaving at 121 °C for 15 min or, if the components were heat labile, were filter sterilised using a 0.22 or 0.45 µm filter prior to use. The supplements and selective antibiotics were added to the molten agar after sterilizing, once it had cooled to approximately 50 °C. The media was poured into sterile Petri dishes (Sarstedt, Leicester, UK) in a laminar flow cabinet. Liquid media was generally stored at room temperature (unless specified) and prepared agar plates was stored at 4 °C.

2.1.1 PALCAM agar media

A vial of PALCAM Supplement (VWR, B-3001, Belgium) 5 ml was brought to room temperature prior to using and 5 ml of sterile RO water was added to it and mixed well. PALCAM agar Base (CM0877, Oxoid) was prepared by adding 34.5 g into 500 ml of RO water and sterilized by autoclaving then was cooled to 45 to 50 °C. Aseptically, the 5 ml of PALCAM Supplement was transferred into the medium then mixed well before dispensing into sterile Petri dishes.

2.1.2 Listeria Selective agar Base

27.75 g of Listeria Selective agar Base (CM0856, Oxoid) was suspended into 500 ml of RO water. This was sterilised by autoclaving then cooled to 50 °C before adding the contents of one vial of Listeria Selective Supplement (SR0140, Oxoid) dissolved in 5 ml of 70 % (v/v) ethanol then mixed well and poured into sterile Petri dishes.

2.1.3 Yeast Extract -Peptone Dextrose broth medium (YPD)

YPD powder (50 g) was suspend in 1 l of distilled water and autoclaved for 15 min at 121 °C. YPD broth was stored at room temperature for a several weeks.

2.1.4 Dulbecco's Modified Eagle medium (DMEM)

Dulbecco's Modified Eagle Medium (DMEM, 500 ml) high glucose was supplemented with 50 ml of foetal bovine serum (FBS), 5 ml of L-Glutamine solution and 10 ml of MEM Non-essential Amino Acid Solution (100x). Then 10 ml aliquots of the media were dispensed in tissue culture flasks and incubated at 37 °C in a humidified incubator gassed with CO₂ 5 % (v/v) for 48 h to ensure no contamination of the media had occurred during preparation. DMEM was stored at 4 °C.

2.1.5 Mueller Hinton agar

15.2 g Mueller Hinton agar (CM0337, Oxoid) was dissolved in 400 ml of RO water, sterilized by autoclaving at 121 °C for 15 min cooled the agar and poured into Petri dishes which were stored at 4 °C.

2.1.6 Nutrient broth No. 2

Nutrient Broth No.2 (CM0067, Oxoid) was prepared by adding 5 g of Nutrient broth No. 2 to 200 ml of RO water. After autoclaving the broth was stored at room temperature for several weeks.

2.1.7 Brain Heart Infusion (BHI) broth

BHI broth (CM1135, Oxoid) was prepared by dissolving 7.4 g of BHI broth in 200 ml of RO water and autoclaved for 15 min at 121 °C. After autoclaving, BHI broth was stored at room temperature for a several weeks.

2.1.8 Brain Heart Infusion (BHI) agar

BHI agar (Lp00II, Oxoid) was prepared by adding 14.8 g of BHI broth and 6 g of agar to 400 ml of RO water and autoclaved at 121 °C for 15 min. After cooling to 45 °C the media was poured into sterile Petri dishes. For use with the Miles & Misra technique (Section 2.12) the agar plates were left in a laminar flow cabinet at room temperature overnight to ensure the plate surface was completely dry before use.

2.1.9 Nutrient broth

Nutrient broth (CM0001, Oxoid) was prepared by dissolving 2.6 g of nutrient broth powder in 200 ml of RO water and autoclaved at 121 °C for 15 min and then was stored at room temperature.

2.1.10 Nutrient agar

Nutrient agar (CM0003, Oxoid) was prepared by dissolved 11.8 g of nutrient agar powder in 400 ml of RO water. The medium was autoclave-sterilized at 121 °C for 15 min and the cooled agar was poured into Petri dishes which were stored at 4 °C.

2.1.11 Top agar supplemented with limited histidine and biotin

Top agar was prepared by adding 3 g agar and 3 g sodium chloride to a flask containing 450 ml of RO water and dissolved by heating on a magnetic stirrer in order to melt the agar. Then, 50 ml of limited histidine and biotin solution (0.5 mM) were added to the molten agar and mixed. The mixture was dispensed into 50 ml in 100 ml Duran bottle then autoclaved at 121 °C for 20 min.

2.1.12 Minimum Glucose (MG) agar plates

MG agar media was prepared by adding of 7.5 g of agar to 465 ml of RO water in 500 ml Duran bottle. The mixture was autoclaved for 20 min at 121 °C. Once the agar had been cooled to approximately 55 °C, 10 ml of VB salt solution (50x) and 25 ml of 40 % (w/v) glucose solution were added aseptically then mixed thoroughly then ampicillin or ampicillin and tetracycline were added to the medium as required per each strain, the amount was added as indicated in (Table 2.4 and 2.5). Thorough mixing was applied after the addition of each of these ingredients to the sterilized agar. The agar medium was distributed into a petri dishes. The plates were stored at 4 °C after the agar was solidified for several weeks.

APPENDIX 3: PREPARATION OF CHEMICAL SOLUTIONS

3.1 Antibiotic solution preparation

Selected antibiotics were prepared in appropriate solvent or water according to manufacturer's instructions. The appropriate amount was added to the media to give final concentrations recommended for each strain to the molten agar media just before pouring of medium into Petri dish. Stock concentrations for antibiotics used in this thesis are listed in (Table 3.1) and antibiotic required for *Salmonella* and *Listeria* strains are listed in (Table 3.2).

3.1.1 Erythromycin 5 µg ml⁻¹

Stock solution of erythromycin 5 μ g ml⁻¹ was prepared by dissolving 50 mg of erythromycin in 10 ml of 70 % (v/v) ethanol and stored at -20 °C.

3.1.2 Ampicillin solution (0.8 %, W/V)

Ampicillin sodium salt solution was prepared by dissolving 200 mg of ampicillin in 25 ml of warm RO water. Once the ampicillin was dissolved the solution was sterilized using a 0.45 μ m membrane filter and stored at -20 °C.

3.1.3 Tetracycline solution (0.8 %, W/V)

Tetracycline solution was prepared by dissolving 200 mg of tetracycline powder in 25 ml of 70 % (v/v) ethanol, once the tetracycline was dissolved the solution was sterilized using a 0.45 μ m membrane filter then aliquots of 5 ml were dispensed and stored at 4 °C in the dark to protect against light (tetracycline is light sensitive).

Antibiotic	Solvent	Stock	Working
		concentration	Concentration
Erythromycin	70 % Ethanol	5 mg ml ⁻¹	5 µg ml⁻¹
Ampicillin sodium salt	Warm RO water	8 mg ml ⁻¹	24 µg ml ⁻¹
Tetracycline	70 % Ethanol	8 mg ml ⁻¹	2 µg ml ⁻¹

Table 3.1. Antibiotics used in this thesis

Strains	Strain numbers	Antibiotics Required
		for plasmid selection
<i>S.</i> Typhimurium	TA98 and TA100	Amp (24 µg ml ⁻¹)
S. Typhimurium	TA102	Amp (24 μ g ml ⁻¹) and
		Tet (2 μ g ml ⁻¹)
L. innocua	ATCC 11994 (Serotype 6a)	Ery (5 µg ml ⁻¹)
L. monocytogenes	10403S (wild type; 1/2a)	
	EGD (wild type; 1/2a)	
	ATCC 23074 (wild type; 4b)	Ery (5 µg ml ⁻¹)
	NCTC 10357 (Δ <i>prfA</i>)	
	NCTC 7973 (<i>prfA</i> *)	

Table 3.2. Antibiotics required for Salmonella and Listeria strains

*Amp (Ampicillin), *Tet (Tetracycline) and Ery (Erythromycin)

3.2 Mutagenic solutions (positive controls for Ames strains)

3.2.1 Sodium azide

A stock solution of sodium azide (S-2002- Sigma) (12.5 mg ml⁻¹) was made by dissolving 12.5 mg of sodium azide powder in 1 ml of RO water. From the stock solutions, 60 μ l was transferred into a 30-ml Universal tube (Sterilin, UK Ltd) containing 5 ml of RO water in order to prepare a working solution of 0.15 mg ml⁻¹ concentration. Both solutions were stored in the dark at 4 °C for several weeks.

3.2.2 Mitomycin C

A stock solution of mitomycin C (M-0503- Sigma) (2 mg ml⁻¹) was prepared by transferring 1 ml of RO water into a vial containing 2 mg of mitomycin powder. A working solution of 0.05 μ g ml⁻¹ concentration was made by adding 50 μ l of mitomycin C stock solution into a bijou bottle containing 2 ml of RO water then mixed well, then stored in the dark at 4 °C.

3.3 Buffers and solutions

3.3.1 Maximum recovery diluent (MRD)

MRD was prepared by suspending 1.9 g of maximum recovery diluent powder (CM0733, Oxoid) in 200 ml RO water and autoclaving for 15 min at 121 °C. After autoclaving, MRD was stored at room temperature for a several weeks.

3.3.2 Phosphate buffered saline (PBS)

PBS was prepared by dissolving one phosphate buffered saline tablet (Oxoid) in 200 ml of RO water. After autoclaving, PBS buffer was stored at room temperature.

3.3.3 Buffered peptone water (BPW)

BPW (F0213W, VWR, Prolabo Chemical, ISO) was prepared by suspending 0.5 g in 500 ml RO water and autoclaved for 15 min at 121 °C. After autoclaving, BPW was stored at room temperature for a several weeks.

3.4 Other chemical solutions

3.4.1 Sodium chloride 0.85 % (w/v)

0.85 g of NaCl (Fisher Scientific) was added into 100 ml of RO water. Nacl was stored at room temperature for a several weeks.

3.4.2 Vogel-Bonner medium E (50X)

The salts (10 g of magnesium sulphate, 100 g of citric acid monohydrate, 500 g of potassium phosphate, dibasic (anhydrous) and 175 g of sodium ammonium phosphate) were transferred in the order indicated into 670 ml of warm RO water in a 2 l Erlenmeyer flask with magnetic stirrer, every salt was completely dissolved before adding next salt. The volume was adjusted to 1 l, the solution was distributed into 200 ml and autoclaved for 20 min at 121 °C. The bottles caps were tightened after cooled and stored at room temperature in dark for several weeks.

3.4.3 Glucose solution 40 % (w/v)

Glucose solution (40 %) (Fisher Scientific) was prepared by adding 400 g of D-glucose to 700 ml of RO water in a 2-l Erlenmeyer flask. The glucose was completely dissolved by stirring. The volume was adjusted to 1 l using RO water. Divided into 200 ml then autoclave sterilized at 110 °C for 10 min. The bottles caps were tightened after cooled and stored at 4 °C.

3.4.4 Limited histidine/ biotin solution (0.5mM)

31 mg of D-biotin (Sigma-Aldrich) and 26.3 mg of L- Histidine. HCL (Acros Organics) compounds were added into 200 ml of RO (about boiling) in a 250-ml volumetric flask. Each salt was thoroughly dissolved by stirring on a magnetic stirrer when the mixture became very clear the next salt was added. The volume was adjusted to 250 ml using OR water. The solution was filter-sterilized by using a 0.45 µm membrane filter. The mixture was divided into 20 ml into glass 30 ml universal tubes and stored at 4 °C for several weeks.

2.23.3 Solubilization solution

Solubilization solution was prepared by adding 2 ml of Glacial acetic acid (Fisher Scientific) to 40 ml of RO water in a beaker then 40 ml of Dimethylformamide (DMF) was added. The mixture was mixed before adding 16 g of sodium dodecyl sulfate (SDS) (L4390, Sigma) thorough mixing was applied after the addition of each of these compounds then the solution was incubated at 37 °C to dissolve. The volume was adjusted to 100 ml using RO water.

2.23.4 MTT Assay reagent

A stock solution of MTT (*3*-(*4*,*5*-dimethylthiazol-2-yl)-*2*,*5*diphenyltetrazolium bromide, 5 mg ml⁻¹) (Acros Organics) was made by adding 10 ml of phosphate buffered saline solutions (BPS, Oxoid) into a vial containing 50 mg of MTT powder. MTT solution was stored at 4 °C in a bottle covered by aluminium foil to protect from light due to the MTT is light-sensitive and MTT stock was stored for up to 18 months.

PUBLICATIONS

Posters

SfAM Early Careers Autumn	Inhibition of Gram-positive bacteria by liquorice
Meeting; Oct 2014	extracts; identification of a potential new food
	grade antimicrobial.
Global Food Microbiology	Identification of the Antimicrobial Effect of
Conference; Aug 2016	Liquorice Extracts on Gram- Positive Bacteria:

Determination of Minimum Inhibitory Concentration and Mechanism of Action Using a *luxABCDE* Reporter Strain.

International Association for Characterisation of the antimicrobial effects of Food Protection; Mar 2017 liquorice extract: selective inhibition of a broad range of Gram-positive bacteria.

Paper

El AWAMIE, M. & REES, C. 2016. Identification of the Antimicrobial Effect of Liquorice Extracts on Gram-Positive Bacteria: Determination of Minimum Inhibitory Concentration and Mechanism of Action Using a *Luxabcde* Reporter Strain. *World Academy of Science, Engineering and Technology, International Journal of Medical, Health, Biomedical, Bioengineering and Pharmaceutical Engineering*, 10, 294-302.

228



Identification of the Antimicrobial Effect of Liquorice Extracts on Gram-Positive Bacteria: Determination of Minimum Inhibitory Concentration and Mechanism of Action Using a *luxABCDE* Reporter Strain.



UNTED KINCOOM - CHEM. - MALAYSIA.

The second second

Madiha El Awamie and Catherine Rees

-

Detaion of Food Sciences, School of Biosciences, The University of Nettingham, Satton Bonington Campus, 1812 510, United Kingdom Create consult of sector contractions

Introduction

Results

100

The public is demanding foods that are more natural and less processed; this has led to a search for new antimicrobial substances from other sources including plants. Liquorice plant is used as a traditional meticine. in different countries around the world to treat many diseases including bacterial infections and also commonly used as flavouring in food products. Studies of antimicrobial activity are normally carried out using challenge tests to determine both the activity range and MIC (minimum inhibitory concentration) of a substance. However it is also important to understand the mechanism of action of preservatives. Bacterial bioluminescence has been used in many studies as an in direct reporter of callular visbility. When transformed with the bacterial lux operon (Fig. 1) , bacteria produce light and the level of light produced is representative of the metabolic activity of the cells [1].



In this study we have used both traditional microbiological techniques and also bacterial bioluminescence to investigative the antimicrobial activity of an extract of Glycymhize glabre (ilquorice) that was provided as food industry waste material as a potential natural preservative [2].

Materials & Methods

- Gruwth was monitored using optical density (popular) or by viable count.
- All strains were cultured at 37 °C in BHt broth
- Microscopic observation were mode using a ±100 or immersion jens using light microscopic and images were captured using microscope with camera attached to a computer following Gram staining
- A Tecan micropiate reader was used to simultaneously monitory optical density (600nm) and to measure light output from cultures (relative light units; RLU). For experiments in the Tecan samples of the cultures (200 µl) transferred into a microplate plate and incubated at 37 °C in a Tecan microplate reader. Cell growth was monitored by using OO₀₀₀₀₀ with readings taken every 10 min for 24 h. To normalise light measurements against cell number the RLU values were divided by the OD values and the data plotted on a linear scale (Fig. 6).

To initially determine whether the extract had antimicrobial activity, bacteria were grown in the presence of liquorice extract at a range of concentrations and growth monitored over time by optical density. It was found that the extract only affected the growth of Gram positive bacteria (all the Gram-negative bacteria tested showed normal levels of growth; see Fig. 2). For Gram- positive bacteria, the MIC was found to be 50 µg ml⁻¹ with sub-lethal inhibition of growth still occurring at a concentration of 12.5 µg ml⁻² (Fig. 3).

To determine if the effects on Gram-positive bacteria were becteriocidal or bacteriostatic, viable count was also determined and it was found that for the 50 µg ml⁻¹ sample where growth with totally inhibited, the number of viable cells did not decline over time indicating that the effect of the extract was bacteriostatic rather than bacteriocidal (Fig. 4).

In addition changes in cell morphology were detected following Gram-staining of samples [Fig. 5]. It was found that samples treated with the liquorice estract were less able to take up the Crystal Violet stain and therefore appeared paler in colour and also the cell shapes were less regular than those seen in the control samples.

Monitoring the bioluminascence levels of the genetically angineered iss derivatives of these bacteria, it was clear that even when growth was not affected, a reduction is the light output could be seen. It was also noted that the metabolic level of the strains challenged with 50 µg mills were very similar to the background control, suggesting that in addition to effects on cell division, the cell membrane may be affected.



cell Cells were able to grow normally when liquorice removed indicating that the extract was bacteriostatic

- The extract was active at 50 µg m¹³ and this is a level that could be applied in a practical way.
- Microscopic analysis indicated that there were changes in cell shape that suggests an effect on cell well synthesis

Bioluminescence used to quickly evaluate the effect of an antimicrobial on bacterial metabolism. 24:11 AWAMUE, M. & REES C. 2016 Identification of the Assemicrobial Effect of Laparnice Extracts on Gram-Positive Bacteries Deterministics of Oran Mechanisms of Action Using a Lambode Reporter Steam World Academy of Science Engineering and Technology International Jaarnal of Medical, Health, Biomedical, Biotengineering and Pharmacentical Engineering 18, 294-302.



Characterisation of the Antimicrobial Effects of Liquorice Extract: Selective Inhibition of a Broad Range of Gram-Positive Bacteria



The University of Nottingham

Madiha El Awamie and Catherine Rees

Disision of Food Sciences, School of Biosciences, The University of Notlingham, Sutton Boolington Campus, LE12 SRD, United Kingdom Email: <u>strengtsformationtlineternation</u>

UNITED KINGDOM - CHINA - MALAYSIA

Introduction

Liquorice extract is commonly used as a food flavouring and we have previously shown that a waste material from the production of liquorice extract has antimicrobial activity against the Gram-positive bacterium Lister/o2. We also had evidence that the extract was able to inhibit the growth of other Grampositive bacteria. Since Bocillus, Brochthrix and Lactic Acid Bacteria (all Gram-positive) commonly contribute to the spoilage of cooked meats and L. monocytogenes contamination is a particular problem associated with sliced deli meats and therefore this extract could be of use to improve the safety of sliced deli meats. The purpose of this study was to further investigate the mechanism of action of the liquorice extract and investigate whether it has potential as a food preservative for sliced meats.

Results

Bioluminescent luxABCDE derivatives of Listeria were used to monitor cell viability after treatment with different concentrations of the liquorice extract (Fig. 1). It was clear that even at sub-inhibitory concentrations, when growth was not affected, there was a reduction is the light output indicated that the cells were experiencing sub-lethal injury. The metabolic levels of the strains challenged with 50 µg ml⁻¹ were very similar to the background control, suggesting that the integrity of the cell membrane may be affected.



Figure 1: Effect of liquorice extract on metabolic activity of Listeria Metabolic levels were monitored by measuring bioluminescence levels of strains carrying the besterial *lavABCDE* genes. Ught output (RLU) was divided by optical density (600nm) to normalize light levels produced from samples containing different numbers of cells

The growth of both Bacillus subtilis and S. aureus are also inhibited by this compound (Fig. 2). A LIVE/DEAD* stain was used to investigate the effect of the extract using both L. monocytogenes and Bacillus subtilis. Cells were exposed to either 12.5 or 50 µg ml1 of the extract and then imaged using fluorescent microscopy 1254 (Fig. 3). Cell injury was evident even immediately after treatment (appearance of red cells in 0 time point). An increasing proportion of red-stained cells were something detected over time confirming that the membrane was being damaged. For both Listeria and Bacillus samples treated with the extract, an increase in cell length and the appearance of chains of cells were seen during the 24 h of exposure, suggesting that cell division was also

being affected



Figure 2: Effect on growth of other Gram-positive bacteria

Evaluating activity on bologna-style sausage

To carry out these experiments L. monocytogenes cultures were centrifuged and cells re-suspended in buffered peptone water to a density of 1 x 106 CFU ml-1. Bologna-style sausages were sliced (~3 mm) and inoculated by dipping for 30 s in L. monocytogenes ± 50 µg ml⁻¹ liquorice extract and then allowed to dry for 30 min to remove excess liquid. The inoculated slices were stored at 6°C and samples (n = 3) removed for sampling at day 0 (slicing) and thereafter at day 5, 10 and 15. At sampling, slices were placed in a stomacher bag with 0-1% (w/v) 8PW and stomached for 60 s (300 RPM). The level of Listeria in the samples was determined by selectively enumeration by plating appropriate dilutions on Palcam Agar to selectively enumerate Listeria. The results showed that the extract was able to completely inhibit the growth of Listeria throughout the cold storage (Fig. 3) whereas an increase in 1 log₁₀ was seen in the control samples.



Figure 3: Experimental design and Effect of extract on growth of Listeria monocytogen tal numbers of bacteria on slices were detected. Control samples contained 1% (v/v) ethanol which was the solvent used for the liquorice extract.



Figure 3: Effect of extract on cell viability and cell shape

Overnight culture of L. monorytogenes and 8. subtiliz were prepared with liquorice extract at 12.5 µg mill & 50 µg mill and rol sample contained 1% (v/v) ethanol (solvent for the extract). Samples were taken at 0 i Circle Chen 8 with aeration. Samples were taken at 3, 6 and 24 h. Cells were harvested by centrifugation (15 min at 10,000 x g) and then ed into 0.85 % (w/v) NaCl before mixing with LIVE/DEAD* BacLight** Bacterial Viability sta n. Sa for 15 min in a dark, then 5 µl of the stained samples examined using a fluorescent microscope.

Conclusions/Significance

- · The extract was found to specifically inhibit only Gram-positive bacteria at low concentrations (50 µg ml-2) but effects were reversible and therefore it has a bacteriostatic action.
- All the different Gram-positive cell types tested are affected in the same way, indicating a common mode of action.
- · Microscopic analysis indicated that there were effects both on cell membrane integrity and cell division.
- This novel antimicrobial extract can inhibit the growth of L. monocytogenes and other Gram-positive spoilage organisms when applied to the surface of sliced deli meats.

Reference

 El Awenie, M., and Rees C (2016) Identification of the Antimicrobial Effect of Leasonce Extracts on Gram Positive Bacter Determination of Minimum Inhibitory Concentration and Mechanism of Action Using a IonABCDC Reporter Strain. Int. J. Cetermination of Minimum Inhibitory Concentration and Mechanism of Action Using a AuxAB Medical, Health, Bioreedical, Bioregineering and Pharmaceutical Engineering 10 (6) 294 302

Identification of the Antimicrobial Effect of Liquorice Extracts on Gram-Positive Bacteria: Determination of Minimum Inhibitory Concentration and Mechanism of Action Using a *luxABCDE* Reporter Strain

Madiha El Awamie, Catherine Rees

Abstract-Natural preservatives have been used as alternatives to traditional chemical preservatives; however, a limited number have been commercially developed and many remain to be investigated as sources of safer and effective antimicrobials. In this study, we have been investigating the antimicrobial activity of an extract of Glycyrrhiza glabra (liquorice) that was provided as a waste material from the production of liquorice flavourings for the food industry, and to investigate if this retained the expected antimicrobial activity so it could be used as a natural preservative. Antibacterial activity of liquorice extract was screened for evidence of growth inhibition against eight species of Gram-negative and Gram-positive bacteria. including Listeria monocytogenes, Listeria innocua, Staphylococcus aureus, Enterococcus faecalis and Bacillus subtilis. The Gramnegative bacteria tested include Pseudomonas aeruginosa, Escherichia coli and Salmonella typhimurium but none of these were affected by the extract. In contrast, for all of the Gram-positive bacteria tested, growth was inhibited as monitored using optical density. However parallel studies using viable count indicated that the cells were not killed meaning that the extract was bacteriostatic rather than bacteriocidal. The Minimum Inhibitory Concentration [MIC] and Minimum Bactericidal Concentration [MBC] of the extract was also determined and a concentration of 50 µg ml1 was found to have a strong bacteriostatic effect on Gram-positive bacteria. Microscopic analysis indicated that there were changes in cell shape suggesting the cell wall was affected. In addition, the use of a reporter strain of Listeria transformed with the bioluminescence genes *luxABCDE* indicated that cell energy levels were reduced when treated with either 12.5 or 50 µg ml⁻¹ of the extract, with the reduction in light output being proportional to the concentration of the extract used. Together these results suggest that the extract is inhibiting the growth of Gram-positive bacteria only by damaging the cell wall and/or membrane.

Keywords—Antibacterial activity, bioluminescence, Glycyrrhisa glabra, natural preservative.

I. INTRODUCTION

OVER the last decade, the foodborne pathogens became an increasing concern through food contamination of products, including meat, fresh fruits and vegetables, caused by bacteria such as *Listeria monocytogenes*, *Campylobacter*

Madiha El Awamie is with the Microbiology and Food Safety, The University of Nottingham, Nottingham, United Kingdom (e-mail: stxma46@nottingham.ac.uk)

Jejuni and Escherichia coli [1]. Health hazards posed by microbial pathogens in food are of major concern to all governments because of circulation of contaminated food between countries increases the chance of outbreaks of foodborne disease. According to International trade statistics (2015) by World Trade Organization (WTO), food represents 42% of total world exports, and levels have increased 6% per year since 2010, and so food is a substantial and increasing area of trade [2]. According to the definition of foodborne illnesses by The World Health Organization (WHO), these are diseases caused by agents that enter the body through the ingestion of food which either infectious or toxic in nature [3]. Approximately 1.8 million deaths were registered in 2005 caused by diarrhoeal disease, and a large proportion of these cases can be attributed to contamination of food and drinking water. However, levels of foodborne disease are generally under-reported and are difficult to measure accurately [4]. Although generally the safety of food has dramatically improved overall, foodborne outbreaks from microbial contamination are still prevalent in many countries, and failure to discover the presence of pathogenic microorganisms especially in the food industry may lead to serious effects on human health [5], [6].

Bacteria such as Campylobacter, Salmonella, L. monocytogenes and E. coli are commonly reported to be responsible for foodborne disease and outbreaks [7], [8]. Generally, all these bacterial pathogens are grouped together, but physiologically they are very different, in particular a major structural difference occurring in the different cell types. For instance, Campylobacter, Salmonella and E. coli are all Gram-negative bacteria and have a cells wall structure including two layers of membrane and a thin layer of the structural polysaccharide peptidoglycan (PG). In contrast Gram-positive bacteria, such as Listeria, Staphylococcus, Enterococcus and Bacillus, have only one cell membrane encased in a thick layer of PG. Hence when considering the action of new antimicrobial agents, it is important to test them against a range of bacterial cell types.

L. monocytogenes is a foodborne pathogen that is almost exclusively transmitted to human through consumption of contaminated food. Its unusual biology - being able to grow

World Academy of Science, Engineering and Technology International Journal of Medical, Health, Biomedical, Bioengineering and Pharmaceutical Engineering Vol 10, No.6, 2016

equally well in soil or in the GI tract of animals and within food production facilities - means that it is associated with a broad range of ready-to-eat (RTE) foods, including dairy products, fresh vegetables, processed meats and shellfish. Infection by this bacterium results in the disease listeriosis that can have a number of different serious consequences including meningitis, meningo-encephalitis and septicaemia and is distinguished by a high mortality rate of up to 30%. The severity of the diseases caused by this bacterium means that the economic costs associated with an infection is high. For instance, during an outbreak in Canada in 2008 related to consumption of contaminated delicatessen meat, 57 cases of listeriosis were reported, resulting in 24 deaths. The costs associated with the cases (including medical and nonmedical costs and productivity losses) were estimated to be nearly \$242 million Canadian dollars [9]. Infection is generally seen in people with weakened immune systems, including pregnant women, infants, and the elderly and although listeriosis occurs infrequently, it is a major concern for the food industry [10], [11].

There are many challenges exist for the control of L. monocytogenes in the food chain, including its ability to survive and grow at low temperatures. Although low temperature is normally used to prevent the growth of bacteria, L. monocytogenes has the unusual ability to adapt to low temperature conditions and grow in refrigerated foods. It can also tolerate and grow over a wide range of pH and moisture levels and can form biofilms making it resistant to disinfection. The bacterium can colonize in food production equipment and environments and, consequently, these bacteria have been shown to persist for months to years in foodprocessing plants, acting as a possible source of crosscontamination of product. Indeed, contamination of food products with Listeria has been identified at many different stages of food production including slicing, packaging and chill-storage [12], and therefore many different techniques are employed to inhibit, spread and survival of L. monocytogenes in food products. So, the control and elimination of this type of bacteria from food products is the best strategy to ensure food safety [13].

Generally, when testing antimicrobial agents, cells are challenged with the test chemical and the effect measured by the level of growth inhibition observed. Despite the fact that the definition of viability depends on the capacity of bacteria to multiply and divide, growth of bacteria in a liquid medium, or the formation of colonies on solid agar, is not considered sufficient evidence that bacteria are either alive or dead after treatment since antimicrobial agents may simply be bacteriostatic and prevent growth without causing permanent cell damage. To address this, other methods have been developed to report on cellular viability, including enzyme and respiratory activity and measurement of ATP levels. These parameters can also report on sub-lethal damage to cells, factors that affect membrane integrity or require energy for repair will reduce ATP levels but not result in cell death [14], [15].

Bioluminescence is the process by which visible light is emitted by an organism as a result of a chemical reaction catalysed by a luciferase enzyme and this phenomenon is seen in a variety of organisms, such as insects, fish, souid, shrimp, and jellyfish, but is also seen in bacteria. Bacterial luciferase (Lux) is encoded by the luxAB genes which catalyse the reduction of a long-chain fatty aldehyde in the presence of molecular oxygen and FMNH2 and results in the emission of light (Fig. 1 B). Cells encoding the full has operon (Fig. 1 A) generate a bioluminescence phenotype without the need to add any additional cofactors or exogenous substrates [15], [16]. Since FMNH₂ production is dependent on a functional electron transport chain, only metabolically active bacteria emit light and therefore it can be used to rapidly monitory antimicrobial activity [15]. Although many bacteria of interest are not naturally bioluminescent, the lux operon can be introduced into a number of different cells on plasmids as long as the genes are engineered to function in that particular cell type [17] and therefore bioluminescence has been widely used to study antimicrobial agents [15], [18].



Fig. 1 A) Organisation of the *lux* operon in *Vibrio* and *Photorhabdus* species including the luciferase genes *luxA* and *luxB* and the *luxCDE* genes which encode a fatty acid reductase complex that regenerates the long chain aldehyde substrate (luciferin) B) The chemical reaction giving rise to light production in bioluminescent bacteria in which the aldehyde is converted to the carboxylic acid in the presence of oxygen reduced flavin mononucleotide (FMNH₂); The by-products of this reaction are flavin mononucleotide (FMNH) and light in the visible spectrum (bioluminescence)

Another commonly used marker gene is the green fluorescent protein (GFP) produced by the jellyfish *Aequorea victoria* which has become a simple and flexible tool used in many applications in field of molecular biology, medicine and cell biology. The protein is small (27-kDa) and produces green fluorescence at 509 nm when excited with UV light at 398 nm [19], [20]. As GFP protein is extremely stable *in viva*, it has proved to be useful when studying cells exposed to some sort of stress condition, as the fluorescence signal does not decrease allowing cells to be easily visualised, even if nonviable [21]. Again, bacteria do not naturally produce *Gfp* but this can also be introduced into bacteria on plasmids [22] and can also be introduced onto plasmid in conjunction with the *lux* genes so that cells containing these plasmids are both bioluminescent and fluorescent [23]. World Academy of Science, Engineering and Technology International Journal of Medical, Health, Biomedical, Bioengineering and Pharmaceutical Engineering Vol:10, No:6, 2016

II. MATERIAL AND METHODS

A. Bacterial Strains, Plasmids, and Growth Conditions

Bacterial strains and plasmids used in this study are listed in Tables I and II, respectively. These strains were transformed with plasmids carrying the bacterial *luxABCDE* and *gfp* genes modified for expression in each of the different bacteria genera used. Strains were recovered MicrobankTM beads held at -80 °C by inoculating into Brain Heart Infusion (BHI) broth and then plating onto BHI agar with erythromycin (Erm; 5 µg ml^{-b}) to select for the presence of the plasmids. Plates were incubated for 24 h at 37 °C and stored at 4 °C. For each experiment, an isolated colony was inoculated in BHI broth supplements with Erm and incubated at 37 °C for 24 h.

TABLET

BACTE	RIAL STRAINS USED IN T	HIS STUDY	
Baclería	Strain numbers or source	Gra	im reaction group
Salmonella enterica serovar Typhimurium	Nottingham lab collect	tion	-ve
Excherichia coli	NCTC 86		-'YE
Staphylococcus aureus	Chicken isolate		+ve
Bacillus subtiliz	var. Niger 168		+ve
Listeria innocua	Nottingham lab collect	tion	+ve
Listeria monocytogenes	10403S (wild type, 1/ EGD (wild type, 1/2 ATCC 23074(wild type NCTC 10357 (Δρη/ NCTC 7973 (ρη/4*	/2a) 8a) 6, 4b) 74) 7	++c
BACTERI	TABLE II A WITH PLASMID USED IN	THIS STU	DY
Bacteria	Promoter & Insert	Vector	Antibiotic selection [µg ml ⁻¹]
L. innocua	BS10, hwABCDE, gfp	pUNK1	8 Erm [5]
L. monocytogenes	BS10 IucABCDE, gfp	pUNK1	8 Erm[5]
1 monocutopenes			

B. Liquorice Extract

NCTC 10357

Liquorice extract (3 mg) was weighed into a sterile bijoux bottle then dissolved in 1 ml of 70% (v/v) by incubating at 37 °C 10 min and was then stored at -20 °C prior to use.

pUNK18

Erm [51

xylA, IncABCDE, g/p

C. Determining Antibacterial Activity of Liquorice Extract

Overnight cultures were prepared and aliquots inoculated into sterile 250 ml Erlenmeyer flasks containing 100 ml of BHI broth. The inoculum of the broth culture was adjusted until the bacterial suspension reached an OD₆₀₀nm 0.05 which was equivalent to an approximately 1x10⁷ CFU ml⁻¹. Three flasks were prepared for each type of bacteria; the first flask (control) contained 1% (v/v) ethanol, the second flask contained 12.5 µg ml⁻¹ of liquorice extract and the third flask contained 50 µg ml⁻¹ of liquorice extract. Ethanol was used as the solvent to dissolve the liquorice extract and therefore ethanol alone (which is known to have antimicrobial activity at high concentration) was used as a control. The concentration of ethanol used was 1% (v/v) as this was the maximum level of ethanol present in either of the test samples following addition of the liquorice extract. Samples were incubated at 37 'C for 30 min with aeration (shaking at 150 rpm), and samples collected for OD measurement at 30 min intervals for approximately 3 h. In addition, the viable count of the culture was determined for each of the time points using standard dilution and plating methods.

D. Gram Stain

Cells were imaged using a standard light microscope after Gram-staining using a x100 oil immersion lens. Images were captured using microscope with camera attached to a computer, and also with using electronic microscope connected to a Cannon camera.

E. Growth in Tecan Microplate Reader

Overnight cultures of bacteria were used to inoculate fresh broth to an OD_{600mm} of 0.05. Samples of the cultures (200 µl) were transferred into the wells of a microtitre plate (Krystal Microplate, clear bottom or Porvair 96 well black, Porvair Sciences, England), and samples incubated at 37 °C in a Tecan Genesis Pro microplate reader. Cell growth was monitored by using OD_{600mm} with readings taken every 10 min for 24 h. Fluorescence (RFU; Excitation wavelength = 485 nm, Emission wavelength = 535 nm) and bioluminescence levels (RLU) readings were taken at regular periods over the course of the experiment as required.

III. RESULTS AND DISCUSSION

To determine whether the extract had antimicrobial activity, bacteria were grown in the presence of liquorice extract at range of concentrations and it was found that 50 µg ml-1 was sufficient to inhibit the growth of bacteria. For all further experiments two different concentrations were used (12.5 µg ml-1 and 50 µg ml-1) and growth was monitored over time using optical density. In addition to these two test conditions, a control was prepared containing 1% (v/v) ethanol to rule out effects of the solvent used to prepare the liquorice extract. It was found that the extract only affected the growth of Grampositive bacteria and at these concentrations all the Gramnegative bacteria tested showed normal levels of growth (see Fig. 2). For the Gram-positive bacteria, at 50 µg ml⁻¹ growth was completely inhibited, whereas at 12.5 µg ml⁻¹ growth rate was reduced but some growth was still detected (Fig. 2; panels C-E).

To determine if the effect of the liquorice extract on the Gram-positive organisms was bacteriocidal or bacteriostatic, *Listeria innocua* was chosen as a model organism being non-pathogenic and non-sporeforming. The experiment was repeated, this time monitoring both viable count and optical density. When 50 µg ml⁻¹ was used, it was found that although growth was totally inhibited, the number of viable cells did not decline over time indicating that the effect of the extract was bacteriostatic rather than bacteriocidal (Fig. 3).



World Academy of Science, Engineering and Technology International Journal of Medical, Health. Biomedical, Bioengineering and Pharmaceutical Engineering Vol.10, No.6, 2016

Fig. 2 The effect of liquorice extract on the growth of Gram-negative bacteria (a) & (b) and Gram-positive bacteria (c)-(e)



Fig. 3 The effect of liquorice extract on the growth (a) and the viable count (b) of Listeria innocua. Blue lines, 1% EtOH (control); Red lines, 12.5 µg ml⁻¹; Green lines, 50 µg ml⁻¹

As many of the bacteriostatic agents affect cell wall synthesis, experiments were carried out to monitor the cell morphology of *L. innocua* after the application of the liquorice was examined. Cultures were prepared as described above and then samples were taken at the begging and the end of experiment, assessment of cell morphology using Gram-

staining of samples. It was found that samples treated with the liquorice extract were less able to take up the Crystal Violet stain and therefore appeared paler in colour and also the cell shapes were shorter and rounder than those seen in the control samples (Fig. 4), typical of agents that result in inhibition of cell division. Therefore, the results suggested that the primary

international Science Index, Pharmacological and Pharmaceutical Sciences Vol:10, No:6, 2016 waset org/Publication/10004804

target of the liquorice extract is something common to the Gram-positive bacteria and may inhibit cell wall synthesis.



L. innocua Control (1% EtOH)



L innocua 12.5 µg ml⁻¹ Fig. 4 Effect of extract on cell morphology

To try and further investigate the action of the liquorice extract, genetically engineered derivatives were used that contain the bacterial hex genes. However, these genes are encoded on plasmids and therefore to ensure that the cells all retain the plasmid, antibiotic selection is needed. In addition, we wanted to investigate whether any differences would be seen if different strains of the same bacteria were used. Hence the effect of strain variation was investigated by using different strains of L. monocytogenes in addition to L. innocua. Three wild type strains of L. monocytogenes were used, including serovars 1/2a and 4b which represent the major causes of human listeriosis (Table I). In addition, two mutant strains which either overexpress the virulence genes (NCTC 7973) or cannot express the virulence genes (NCTC 10537) were used. To carry out these experiments, different strains of bacteria (Table I) were grown in BHI broth supplemented with Erm (5 µg ml⁻¹). These cultured were then diluted into fresh BHI + Erm to a standardized OD_{coo}nm of 0.05. To allow multiple samples to be monitored, and to allow simultaneous measurement of OD and light, cultures were transferred into a microtiter plate and incubated at 37 °C over a 24 h period. Due to the possibility of contamination using a microplate reader an uninoculated control sample was included (purple line, Fig 5) to show that any change in optical density or light was due to the organism inoculated into the broth.

The results gained using the microtitre plate showed the same pattern as before, in that a concentration of 50 µg ml⁻¹ totally inhibited growth, but at 12.5 µg ml⁻¹ very little inhibition of growth was seen (Fig. 5).

The pattern of results gained was exactly the same for all strains and species of *Listeria* tested, showing that this was a very reproducible effect. However, when the levels of light produced from each of the strains was examined (Fig. 6) it was clear that even when growth was not affected, a reduction is the light output could be seen, although this was more pronounced in some strains that others. This result indicated that the cells were experiencing some stress at the sub-lethal injury level as the level of light is representative of the overall metabolic state of the cells. It was also noted that the metabolic level of the strains challenged with 50 μ g ml⁻¹ were very similar to the background control, suggesting that in addition to effects on cell division, the cell membrane may be affected.

Since there were now both variations in growth rate and light output recorded it was important to normalize the bioluminescence data according to the number of cells in the sample (i.e. 10 cells producing 1 unit of light would give the same signal as 1 cell producing 10 units of light). To address this, the bioluminescence data was divided by the optical density data and the results are presented in Fig. 7. This treatment of the data clearly revealed that the cells treated with 12.5 µg ml⁻¹ were experiencing sub-lethal injury resulting in overall lower levels of metabolic activity. Since cell division was not affected in these cells as the growth rate was close to that of the control, this suggests that at sub-lethal levels the liquorice extract may be affecting either cell membrane integrity or other enzymes required for the production of FMNH₂ within the cells.

One question that remained was whether the presence of the antibiotic required for the selection of the plasmids was providing an additional stress on the cells that accounted for the results seen in the last experiment. To determine whether the presence of the antibiotic was required to maintain the hox plasmids inside the cells, the stability of the plasmid was determined by growing L. monocytogenes 10403S containing the plasmid in the presence and absence of the selective antibiotic (Erm 5µg ml-1), and rates of plasmid-loss were calculated (Tables III & IV). These experiments were also performed with and without the additional stress caused by the presence of the liquorice extract. The stability of the plasmid was determined by comparing the viable count of samples taken from the experiment and plating on agar with and without Erm and is expressed as the 100% of cells (viable count on BHI) retaining the plasmid (viable count on BHI Erm).

When grown in the absence of the antibiotic, the rates of plasmid loss were very low (92-80%; Table III) indicating that the plasmids themselves were very stable, but that plasmid stability varied with the concentration of the liquorice extract, suggesting that it was exerting a metabolic stress on the cells. In the presence of the antibiotic in the growth medium, the plasmid was more stable (as expected) but even then plasmid loss was seen when the cells were grown in the presence of the liquorice extract (93-97%; Table IV). This result validated the data shown in Fig. 7, since it is clear that the addition of the antibiotic alone is not causing a major stress on the bacterial cells. World Academy of Science, Engineering and Technology International Journal of Medical, Health, Biomedical, Bioengineering and Pharmaceutical Engineering Vol:10, No:6, 2016



	PLASMID STABILITY IN LM10403S WITHOU	03s without Selection		
LM 10403s (BS10, httABCDE, gfp) without Erm in the Growth Media	BHI Agar with 5 µg ml ⁻¹ Erm = Plasmid Present in Cell (cfu ml ⁻³)	BHI Agar without Erm = viable count (cfu ml ⁻¹)	% Bacteria with Plasmid	
EIOH Control	2.7×10 ³	3 2x10 ⁹	92%	
12.5 µg/ml liquorice extract	2.54x10 ⁹	3.06x10 ⁹	82%	
50 µg/ml liquorice extract	1x10 ⁶	Ix10 ⁶	80%4	

TABLE IV PLASMED STABILITY IN LM10403s WITH SELECTION (ERM)			
LM 10403s (BS10, IntABCDE, gfp) with 5 µg ml ⁻¹ Erm in the Growth Media	BHI Agar with 5 µg ml ⁻¹ Erm Plasmid Present in Cell (cfu ml ⁻¹)	BHI Agar without Erm - viable count (cfu ml-1)	% Bacteria with Plasmid
EtOH Control	1.43x10 ¹⁰	1.34x10 ¹⁰	100%
12.5 µg/ml liquorice extract	2.64x10 ¹⁰	2.83x10 ¹⁰	93%
50 µg/ ml liquorice extract	9.7x10 ⁷	9.9x10 ⁷	97%
*Erm enthromatin			

International Scholarly and Scientific Research & Innovation 10(6) 2016 299

scholar waset org/1999.9/10004804


World Academy of Science, Engineering and Technology International Journal of Medical, Health, Biomedical, Bioengineering and Pharmaceutical Engineering Vol 10, No 6, 2016

Fig. 6 Effect of liquorice extract on light emitted by Listeria strains

IV. CONCLUSION

In conclusion, the results of this work show that Grampositive bacteria are more susceptible to the liquorice extract than Gram-negative bacteria. A range of different cell types were tested (Gram-positive rods, Gram positive cocci and spore forming organisms) and all showed equal sensitivity suggesting there is a common target in all these different types of Gram-positive bacteria. Using *L. monocytogenes* strains and *L. innocua* as model organisms, the nature of the growth inhibition seen was further investigated and it was clear that cell growth, including cell division, was affected. At sub-lethal concentrations, even when growth rate was not particularly affected, there was evidence of sub-lethal injury and cell stress.

In this study, we have used bioluminescence to quickly evaluate the effect of an antimicrobial on bacterial metabolism. This approach has been used before by manyworkers but some caution must be taken to ensure that the results gained are not simply due to the presence of the plasmid in the strains. In this case we found that the stress caused by treating the bacteria with the liquorice extract did exert some additional metabolic pressure on the cells, leading

300

to plasmid loss unless antibiotic was added to the media but there was no evidence of a significant difference in results gained with and without antibiotic.

Another area of caution is that strains of bacteria such as Listeria that have been cultured for long periods of time in the laboratory may have lost some of their fitness. To address this, we tested a range of different isolates, including examples of the two serovars that are responsible for the majority of human infections. Again no different was seen between the sensitivity of different strains indicating that the results gained were very reproducible and not dependent on the test strain used. In addition, we tested two well characterised mutants of L. monocytogenes, one which over expressed the virulence genes (NCTC 7973 $prfA^*$) and another than is incapable of expressing the virulence genes due to a deletion of the virulence gene regulator (NCTC 10357; $\Delta pfrA$). The results gained with these strains were also identical the results gained with wildtype L. monocytogenes and L. innocua, which has a complete absence of any of the virulence genes found in Listeria. Again these results suggest that the target site is found in all cell types and is not associated with the virulence traits of these bacteria.



Fig. 7 Effect of liquorice extract on light emitted by Listeria strains, light emission was normalized by dividing RLU/OD

Treatment with 50 µg ml⁻¹ of liquorice extract completely inhibited the growth of all of the Gram-positive bacteria tested, but when plated on to media without the extract they were able to grow normally indicating that stress is bacteriostatic rather than bactericidal. It is interesting to note that without the use of the bioluminescent reporter genes, the

separate effect on cell metabolism would not have been evident and therefore this shows that using these tools is a powerful way to better identify stresses that have an effect on metabolic activity through damaging the membrane integrity or affecting enzymes associated with energy generation.

301

scholar waset org/1999.9/10004804

World Academy of Science, Engineering and Technology

Since this extract was produced as a waste product from the food industry, there is the potential that it could be applied as a novel food preservative. The extract was active at 50 µg ml⁻¹ and this is a level that could be used for such a practical application. However, more work would be required to establish whether there would be any organoleptic changes to the foods if used at this level. Another area of interest is if it could be used to inhibit the growth of Clastridium difficile in the human gut in patients suffering with hospital acquired infections. This difficult to treat Gram-positive bacterium causes a major problem, particularly in patients recovering from extensive antibiotic therapy. A formulation of this extract given as a drink could help target these bacteria and allow normal gut flora to re-establish without the need for further extensive antibiotic therapy.

REFERENCES

- Organization, W. H. WHO estimates of the global burden of foodborne diseases: foodborne disease burden epidemiology reference group 2007-2015 (cited 2016 Feb 8).
- [2] WTO. International trade statistics 2015. Geneva: World Trade Organization. [3]
- WHO. Food safety & food-borne illness. Fact sheet no. 237 (reviewed
- WHO. Food safety & food-borne illness. Fact sheet no. 237 (reviewed March 2007). Geneva: World Health Organization; 2007a.
 WHO. The world health report, 2007. Global public health security in the 21st century. Geneva: World Health Organization; 2007b.
 Farber, J., Coates, F. & Daley, E. 1992. Minimum water activity requirements for the growth of *Listeria monocytogenes*. *Letters in Applied Microbiology*, 15, 103-105.
 Mclinden, T., Sargeani, J. M., Thomas, M. K., Papadopoulos, A. & Fazil, A. 2014. Component costs of foodborne illness: a scoping review. *BMC Public Health*, 14, 1.
- BMC Public Health, 14, 1. Alocilja, E. C. & Radke, S. M. 2003. Market analysis of biosensors for 171
- [8]
- Audeija, E. C. & Ranke, S. M. 2003. Market analysis of biosensors for food safety. Biosensors and Bioelectronics, 18, 841-846. Chemburu, S., Wilkins, E. & Abdel-Hamid, 1. 2005. Detection of pathogenic bacteria in food samples using highly-dispersed carbon particles. Biosensors and Bioelectronics, 21, 491-499.
- [9] Theemas, M. K., Vriezen, R., Farber, J. M., Currie, A., Schlech, W. & Pazil, A. 2015. Economic cost of a Listeria monocytogener outbreak in Canada, 2008. Foodborne Pathogens and Disease, 12, 966-971.
 [10] Olszewska, M. A., Panfil-Kuncewicz, H. & Laniewska-Trokenheim, L. 2015. Detection of Viable but Non-culturable cells of Listeria monocytogener with the use of direct epifluorescent filter technique. Journal of Food Safety, 35, 86-90.
 [11] PAGHUL P. 2003. Context
- [11] RAGHU, R. 2013. Listeria monocytogenes: An interesting pathogen. Microbiology Focus 5, 1-2.
- [12] Montañez-Izquierdo, V. Y., Salas-Vázquez, D. I. & Rodriguez-Jerez, J. J. 2012. Use of epifluorescence microscopy to assess the effectiveness of J. 2012. phage P100 in controlling Listeria monocytogenes biofilms on stainless steel surfaces. Food Control, 23, 470-477.
- [13] Vazquez-Boland, J. A., Kuhn, M., Berche, P., Chakraborty, T., Dominguez-Bernal, G., Goebel, W., et al. (2001) *Linterio* pathogenesis and molecular virulence determinants. *Clinical Microbiology Reviews*, 1997 (2019) 1997 (201 14(3), 584e640
- [14] Nocker, A., Caspers, M., Esveld-Amanatidou, A., Van der Vossen, J., Schuren, F., Montijn, R. & Kort, R. 2011. A multiparameter viability assay for stress profiling applied to the food pathogen Listeria monocytogenes F2365. Applied and Environmental Microbiology, 77. 6433-40
- [15] Hill, P. J. & Stewart, G. S. 1994. Use of Jux genes in applied hioche mistry. Journal of Bioluminescence and Chemiluminescence, 9, 211-215
- 211-215.
 Meighen, E. 1993. Bacterial bioluminescence: organization, regulation, and application of the lux genes. *The FASEB Journal*, 7, 1016-1022.
 [17] Perehinee, T. M., Qazi, S. N., Gaddipati, S. R., Salisbury, V., Rees, C. E. & Hill, P. J. 2007. Construction and evaluation of multisite recombinatorial (Gneway) cloning vectors for Gram-positive bacteria. *BMC Information Biology*, 8, 1 BMC Molecular Biology, 8, 1.

International Scholarly and Scientific Research & Innovation 10(6) 2016 302

- [18] Robinson, G. M., Tonks, K. M., Thorn, R. M. & Reynolds, D. M. 2011. Application of bacterial bioluminescence to assess the efficacy of fast-acting biocides Antimicrobial Agents and Chemotherapy, 55, 5214-5219
- J. 19.
 Loreang, J., Tuori, R., Martinez, J., Sawyer, T., Redman, R., Rollins, J., Wolpert, T., Johnson, K., Rodriguez, R. & Dickman, M. 2001. Green fluorescent protein is lighting up fungal biology. *Applied and Environmental Microbiology*, 67, 1987-1994.
 Zimmer, M. 2002. Green fluorescent protein (GFP) applications, structure, and related photophysical behavior. *Chemical reviews*, 102, 759-782.
- 759-782
- [21] Lowder, M., Unge, A., Maraha, N., Jansson, J. K., Swiggett, J. & Oliver, J. D. 2000. Effect of Starvation and the Viable-but-Monculturable State on Green Fluorescent Protein (GFP) Fluorescence in GFP-Tagged Pseudomonas fluorescenu A506. Applied and Environmental
- Paradomonas Intorescenta A306. Applied and Environmental Microbiology, 66, 3160-3165.
 [22] Qazi, S., Rees, C., Mellits, K. & Hill, P. 2001. Development of gfp vectors for expression in *Listeria monocytogenes* and other low G+C Gram-positive bacteria. *Microbial Ecology*, 41, 301-309.
 [23] Qazi, S. N., Counil, E., Morrissey, J., Rees, C. E., Cockayne, A., Winzer, K., Chun, W. C., Williams, P. & Hill, P. J. 2001. agr expression proceeding service of intermediated Stendbedgeneers.
 - precedes escape of internalized Staphylococcus endosome. Infection and Immunity, 69, 7074-7082. wear from the host

scholar waset org/1999.9/10004804

10004804